

## UNITED KINGDOM · CHINA · MALAYSIA

## School of Veterinary Medicine and Science Faculty of Medicine and Health Sciences

# Toll-like receptors (TLRs) and T cell effector function

By: PhD student Majid Mohammed Mahmood

PhD Thesis submitted to the School of Veterinary Medicine and Science, Faculty of Medicine and Health Sciences, University of Nottingham, UK for the Degree of Doctor of Philosophy

#### Abstract

The aim of the project was to identify whether cattle T cell subsets express toll-like receptors (TLRs), and if so whether there was any functional consequence of this when they were stimulated with TLR-ligands. CD4+ T cells, CD8+ T cells, and  $\gamma\delta$  T cells were studied in the context of whole PBMC or as isolated cells. To develop rabbit Tregs as a source of cells to study expression of toll-like receptors (TLRs), putative rabbit regulatory T cells (Tregs) were developed as cultured cells from MLN as part of an ongoing project and to inform generation of cattle Treg development for this study. Culture of rabbit MLN cells resulted in detection of CD4+CD25hiFOXP3+ (putative nTregs) in about one fifth of the total cells. The cultured rabbit Tregs did not exhibit any immunosuppressive effect on autologous MLN (Mesenteric lymph node) cells in a suppression assy. This and the complexity of this method meant that the cattle Treg work was not initiated for this study.

The main part of the project involved screening of 10 bovine toll-like receptors (TLRs 1-10) on cattle blood samples (PBMC fraction and T cell subsets) plus investigation of their expression using RT-qPCR to detect transcripts in the T cell subsets. TLRs were variably expressed in the T cell subsets from the samples. With regard to function measured by proliferation in response to stimulation with TLR agonist (ligand), CFSE (Carboxy-fluorescein Succinimidyl Ester)-labelled PBMCs were stimulated with TLR1-9 agonists and each CFSE-labelled T cell subset identified by flow cytometry and proliferation based on the CFSE profile for each subset within the whole PBMC sample. No significant proliferation was recorded in activated PBMCs while T cell subsets showed variation in responses with

Ι

CD4 T cells generally most effective followed by CD8+ T cells and least proliferation was seen with  $\gamma\delta$  T cells.

The functional role of expressed TLRs stimulated with ligands and Con-A or antiCD3 were also assessed by expression or not of 7 cytokines (CXCL-8, IFNa, IFNy, TNFa, TGF $\beta$ , IL-4, and IL-10) by specific ELISA. However, the results showed variation between different cattle blood samples. T cells generally activated PBMCs producing CXCL-8 while the isolated T cell subsets showed a more selective secretion. No IFN a expression was found. IFNy was secreted in a limited amount in one PBMC animal and some  $\gamma\delta$  T cell samples rather than CD4 or CD8 T cells. TNF-a was produced by one PBMC sample stimulated with TLR2 agonist. The T cell subsets did not record any TNF-a secretion. There was no real functional link between the CFSE proliferation assay and the cytokine expression results.

In conclusion, TLRs are expressed on cattle T cell subsets but with variable activation outcomes after stimulation of the cells with TLR ligands.

#### Acknowledgment

The first thanks and invaluable gratitude to the God who alleviated the hardship I challenged during the whole my life. I would like to raise all my thanks and appreciations to my main supervisor, Professor David Haig, for his very generous continuing support as well as for his good humour especially when I was exhausted from experiments in the Lab. His guidance and following me up over my PhD made me so happy to be supervised by such a full of knowledge scientific man. In addition, I would like to show my extreme thank and gratitude to my co-supervisors Associate Professor Robin Flynn and Stephen Dunham for his further invaluable support.

All my thanks and deep appreciations to my Mother who did not stop asking about me throughout all the duration of my scholarship (since the day I travelled from Iraq to the UK up to now). My father and all brothers thank you all.

A very big thank would be deserved by my wife (Zeinab) who stood day by day beside me in all steps of my ordinary and scientific lives. I will carry this favour on my shoulders and will still show my love to her until death. Sure, I can not forget my little sons who made my life busy and funny (and terrible to some point!) at the same time, so double thank you both, you are my heart.

From my deep heart, I would like to thank my sponsor "The Higher Committee of Education Development in Iraq" abbreviated as HCED who funded my scholarship.

I would like to express my thanks to Belinda Wang, the hard worker technician, for her advice in the Lab.

 $\mathbf{III}$ 

A special thanks to David Onion and his assistant Nicola Croxall who helped me in the flow cytometry work in the Queen's Medical Centre (QMC) in the Medical School / University of Nottingham, UK. Huge thanks to Tracey Coffey who supplied me with published bovine TLRs primers and probes.

The final bundle of huge thanks would go to my beloved postgraduate friends Osama Kumati, Emma Drinkall, Azad Sulaiman, Adam Blanchard, Natalie Pickwell as well as the undergraduate friend Joseph Barton and others (please forgive me whoever I forgot to mention) who helped me in the Lab work phase. Thank you all.

## Dedication

To:

My parents, family members and all relatives ... With love

My real friends with a special respect ... A friend in need is a friend indeed

Humanity ... All human beings

Majid

### DECLARATION

I declare that the information in this thesis was carried out in accordance with the Regulations of University of Nottingham. The work is original, except where indicated by special reference in the text, and no part of the thesis has been submitted for any other academic award. Any views expressed in the thesis are those of the author.

Signature:

Date: 07-08-2020

## **Table of contents**

AbstractI
AcknowledgmentIII
DedicationV
DECLARATION
Table of contentsVII
List of tablesXII
List of figuresXIII
List of abbreviationsXVIII
1.1 The immune response1
1.2 The innate immune response 1
1.2.1 Innate immune cell pattern recognition receptors
1.3 The adaptive immune response 10
1.4 Regulatory T cells (Tregs) 22
1.5 Toll-like receptors (TLRs): A bridge between innate and adaptive immunity
1.5.1 TLR structure and function 25
1.5.2 T and B lymphocytes expressing TLRs
1.5.3 TLR signalling pathways
1.5.3.1 Extracellular TLR signalling pathway
1.5.3.2 Intracellular TLRs signalling pathways
1.6 Bovine TLRs – in depth highlights and specifications
1.6.1 Bovine TLR1
1.6.2 Bovine TLR2 39
1.6.3 Bovine TLR3 40
1.6.4 Bovine TLR4 40
1.6.5 Bovine TLR5 41
1.6.6 Bovine TLR6 42
1.6.7 Bovine TLR7 and TLR8 43
1.6.8 Bovine TLR9 44
1.6.9 Bovine TLR10 44
1.7 TLR agonists
1.8 TLR agonists /ligands as new generation adjuvants
1.9 Hypotheses, objectives, and questions 48
2.1 Animals 49
2.2.1 Rabbit MLN cells 50
2.2.2 Bovine blood 50

2.3 Isolation of PBMCs from bovine blood
2.4 Reagents
2.4.1 IMDM Complete Medium 53
2.4.2 Reagents and cytokines 53
2.4.3 Titration of reagents
2.4.3.1 Interleukin-2 (IL-2)53
2.4.3.2 Con A, PHA and PMA titration
2.4.4 TLR Ligands (Agonists) 54
2.5 Methods specific to the putative rabbit Treg study
2.5.1 Generation of rabbit regulatory T cells
2.5.2 Flow cytometry detection of putative rabbit Tregs
2.5.3 Sorting of putative rabbit Tregs
2.5.4 Putative rabbit Treg functional assay: target cell proliferation and labelling
2.5.5 Flow cytometry measurement of putative rabbit Treg functional assay T cells (inhibition/stimulation of proliferation by Treg dilutions).
2.6 Molecular techniques specific to the putative rabbit Treg study 64
2.6.1 Total RNA extraction for MLN and putative Treg cells
2.6.2 cDNA synthesis
2.6.3 SDHA (Housekeeping) gene expression
2.6.4 Real Time - Polymerase chain reaction (RT-gPCR)
2.7 Bovine lymphocyte study
2.7.1 Cytospin and Diff Kwick staining
2.7.2 PBMC fractionation
2.7.3 MACS protocol for fractionation of CD4, CD8, and vo T cells 70
2.7.4 Flow cytometry
2.7.5 Major T cell subsets multicolour fluorophores panel
2.8 Molecular techniques specific to Bovine lymphocyte study
2.8.1 Total RNA extraction
2.8.2 RNA guantification (Nanodrop®8000)
2.8.3 Complementary DNA synthesis
2.8.4 Polymerase chain reaction (PCR)
2.8.4.1 Conventional PCR
2.8.4.2 Gel electrophoresis
2.8.4.3 Quantitative real-time PCR (RT-qPCR)
2.8.4.4 RT-qPCR method for TLR quantification
2.8.4.5 Normalisation of bovine reference genes
2.9 Cytokine measurements (ELISA)

2.9.1 Generic materials and common steps
2.9.2 IL-8 (CXCL-8)
2.9.3 IFNa
2.9.4 TNF-a
2.9.5 IL-4
2.9.6 IL-10
2.9.7 TGFβ
2.9.8 IFNγ100
2.10 Statistical analysis102
3.1 Abstract
3.2 Introduction
3.3 Materials and methods105
Results
3.4 Preliminary results
3.4.1 TGF-β1105
3.4.2 IL-2
3.5 Con A, PHA and PMA titrations for stimulating T cells
3.6 Phenotype analysis of MLN and cultured MLN cells108
3.7 Target cells for putative Treg functional assay
3.8 Putative rabbit Treg functional assay112
3.9 RT-qPCR for FOXP3 in the MLN cells114
3.10 Discussion
4.1 Abstract
4.2 Introduction
4.3 Materials and methods119
Results
4.4 PBMC separation120
4.5 PBMCs stained with Diff Kwick123
4.6 Flow cytometry124
4.6.1 Major PBMC components124
4.7 PBMCs fractionated by MACS128
4.7.1 Flow cytometry analysis of T cell subsets fractionated by MACS (purity test)129
4.8 Bovine TLR gene expression130
4.8.1 Choice of reference genes for TLR expression analysis by RT-qPCR
4.8.2 Gel analysis of TLRs and reference gene amplicons after expression measurement by RT-qPCR136
4.8.3 RT-qPCR results of TLR1-10 gene expression in PBMCs138

4.8.3.1 RT-qPCR results of TLR1-10 gene expression on PBMCs138
4.8.3.2 RT-qPCR results of TLR1-10 gene expression in the fractionated T cell subsets
4.9 Discussion
5.1 Abstract
5.2 Introduction161
5.3.1 Stimulation of bovine CFSE-labelled PBMCs with TLR ligands 162
5.3.2 Stimulation of bovine CFSE-labelled T cell subsets with TLR ligands162
5.3.3 Stimulation of CFSE-labelled PBMCs with TLR ligands and analysis of T cell activation to TLR ligands within the whole PBMC population. 163
5.3.4 The effect of fractionated $\gamma\delta$ T cells on CFSE-labelled bovine PBMC proliferation164
5.3.4.2 Role of unstimulated purified $\gamma\delta$ T cells on activated bovine autologous PBMCs (stimulated by Con A)164
Results
5.4 PBMCs stimulated with TLR ligands and labelled with CFSE (proliferation assay)166
5.5 Proliferation of T cell subsets of the PBMC stimulated with TLR ligands
5.5.1 Results of CFSE-labelled CD4 T cells stimulated with TLR ligands170
5.5.2 Results of CFSE-labelled CD8 T cells stimulated with TLR ligands173
5.5.3 Results of CFSE-labelled $\gamma\delta$ T cells stimulated with TLR ligands $178$
5.6 Results of CFSE-labelled PBMCs stimulated with TLR ligands: T cell subset responses within the whole PBMC population
5.7 Functional role of bovine $\gamma\delta$ T cells without TLR-agonist stimulation
5.7.1 Unstimulated and/or activated purified $\gamma\delta$ T cells have a stimulatory effect on resting bovine autologous PBMCs (unstimulated)
5.7.2 Purified unstimulated $\gamma\delta$ T cells have stimulatory effect on activated bovine autologous PBMCs (stimulated)193
5.8 Discussion
6.1 Abstract
6.2 Introduction
6.3 Materials and methods204
Results
6.4 CXCL-8 production from TLR-ligand stimulated PBMCs and T cell
subsets

6.4.1 CXCL-8 results for PBMCs stimulated with TLR ligands205
6.4.2 CXCL-8 results of T cell subsets: CD4 T cells stimulated with TLR ligands212
6.4.3 CXCL8 results of CD8 T cells stimulated with TLR ligands214
6.4.4 CXCL8 results of $\gamma\delta$ T cells stimulated with TLR ligands217
6.5 IFN-a, TNF-a, and TGF-beta in TLRL-stimulated PBMCs and T cell subsets
6.6 IL-4
6.6.1 IL-4 results of PBMCs and T cell subsets stimulated with TLR ligands240
6.6.2 IL-4 results of CD4 T cells stimulated with TLR ligands247
6.6.3 IL-4 results of CD8 T cells stimulated with TLR ligands250
6.6.4 IL-4 results of $\gamma\delta$ T cells stimulated with TLR ligands253
6.7 IL-10 results of PBMCs and T cell subsets stimulated with TLR ligands
6.8 IFNy results of PBMCs and T cell subsets stimulated with TLR ligands
6.8.1 IFNy results of PBMCs stimulated with TLR ligands
6.8.2 IFNγ results of T cell subsets stimulated with TLR ligands274
6.9 Discussion
7.1 General overview
7.2 Putative Rabbit Tregs
7.3 Bovine T cell TLR gene expression
7.4 The activation status of bovine TLRs assessed by CFSE and cytokine assays and functional aspects of $\gamma\delta$ T cells
7.5 Cytokine responses
7.6 Conclusions and recommendations
8.1 Preparations of generic reagents, media, and compounds290
8.1.1 RPMI complete medium
8.1.2 Freeze-mix medium
8.1.3 MACS running buffer290
8.1.4 MACS rinsing buffer291
8.1.5 Tris Acetate EDTA buffer (50X TAE)
8.1.6 Tris Acetate EDTA buffer (1X TAE)
8.1.7 Erythrocyte lysis buffer (ELB)
8.2 Antibodies used in putative rabbit Treg project
8.3 Antibodies used in bovine TLR project
8.4 ELISA kits used for cytokines' measurements
8.5 Kits used for gene expression and molecular work
Bibliography

## List of tables

Table 1.1: Major cells involved in the innate immune response.         4
Table 1.2: Non TLRs pathogen recognition pattern classes.         9
Table 1.3: Major cells involved in the adaptive immune response
Table 1.4: Main functional cytokines released from immune cells.         18
Table 1.5: Ligands of different TLRs and their targets
Table 1.6: Gene accession numbers of up to date TLR1-10 genes in cattle
(species Bos taurus). Cited from National Centre for Biotechnology
Information (NCBI). Accessed 30-08-2017
Table 2.7: Animals used for blood collection. Age, gender, and breed with
the source are given
Table 2.8: Human TLR1-9 Agonist Kit components
Table 2.9: TLR ligand concentrations used for stimulation of PBMCs and/or
T cell subsets
Table 2.10: Antibodies used for labelling putative rabbit Tregs
Table 2.11: PCR program for rabbit MLN DNA amplification
Table 2.12: RT-qPCR universal program followed for running samples
(Roche)
Table 4.24: Volumes of bovine blood showing variable PBMC counts121
Table 4.25: Presence/absence of TLR target genes on PBMCs measured by
RT-qPCR from7 animals140
Table 4.26: Presence/absence of TLR target genes on CD4 T cells measured
by RT-qPCR from 7 animals143
Table 4.27: Presence/absence of TLR target genes on CD8 T cells measured
by RT-qPCR from 7 animals145
Table 4.28: Presence/absence of TLR target genes on $\gamma\delta$ T cells measured
by RT-qPCR from7 animals147

## List of figures

Figure 1.1: Schematic diagram showing a typical TLR structure26
Figure 1.2: TLR5 signalling pathway as an example of extracellular TLRs signalling pathway
Figure 2.3: Layering buffy coat cell suspension on Histopaque 107752
Figure 2.4: Study protocol for the generation of putative rabbit Tregs and the suppression assay
Figure 2.5: ELISA standard curve for Bovine IL-891
Figure 2.6: ELISA standard curve for Bovine IFNa93
Figure 2.7: ELISA standard curve for Bovine TNFa95
Figure 2.8: ELISA standard curve for Bovine IL-496
Figure 2.9: ELISA standard curve for Bovine IL-10
Figure 2.10: ELISA standard curve for Bovine TGFβ
Figure 2.11: ELISA standard curve for Bovine IFN gamma
Figure 3.12: IL-2 titration for MLN cell proliferation
Figure 3.13: Con A, PHA, and PMA titration for Treg assay target cell proliferation, using MLN cells
Figure 3.14: Triple stained generated Treg cells in Tr2 sample110
Figure 3.15: Validation FACscan for FOXP3 antibody-labelled cells in an MLN sample after 6 days incubation (Tr2 sample)110
Figure 3.16: Treg target cells proliferation labelled with CellTrace violet (CFSE, abbreviated as CTV)112
Figure 3.17: CD4+CD25hi cell sorting from putative rabbit Tregs generated in culture after 6 days incubation113
Figure 3.18: $5X10^5$ cells/ml of Treg target cells labelled with CFSE 113
Figure 3.19: Average +/- range of FOXP3 gene expression levels in duplicate samples of each of the MLN from 5 animals on day 6 of Treg culture
Figure 4.20: PBMC average cell count per 1 ml blood (PBMC index)122
Figure 4.21: PBMCs stained with Diff Kwick
Figure 4.22: FACS results for bovine PBMCs (Mean of 7 animal PBMC samples and range)
Figure 4.23: FACs dot plot analysis of PBMC samples from Bov A (A) and Bov B127

Figure 4.24: CD4, CD8 and $\gamma\delta$ T mean cell counts +/- sem (counted by haemocytometer) fractionated by MACS129
Figure 4.25: MACS enrichment of CD4+ T cells, CD8+ T cells and WC1+ $\gamma\delta$ T cells in PBMC samples 'Bov A' and 'Bov B'
Figure 4.26: RPLP2 amplification and standard curves133
Figure 4.27: RPLP0 amplification and standard curves134
Figure 4.28: GAPDH amplification and standard curves135
Figure 4.29: DNA products of reference and target genes expressed on PBMCs, amplified by RT-qPCR and ran on a 2% agarose gel137
Figure 4.30: TLR 1-10 gene expression by RT-qPCR from 7 animals of bovine blood presented as the Mean ± standard error138
Figure 4.31: TLR 1-10 gene expression by RT-qPCR from 6 animals of bovine blood PBMC shown as the Mean ± standard error142
Figure 4.32: TLR 1-10 gene expression by RT-qPCR from 7 animals of bovine blood shown as the Mean ± standard error144
Figure 4.33: TLR 1-10 gene expression by RT-qPCR from 7 animals of bovine blood shown as the Mean ± standard error146
Figure 4.34. TLR expression in the studied cells (PBMCs, CD4 T cells, CD8 cells and gamma delta T cells) presented in groups to show the comparison153
Figure 5.35. CFSE-labelled PBMCs stimulated with TLR ligands showing gated proliferative cells upon stimulation (blood animal Bov 5).
Figure 5.36: Proliferation (as percentages CFSE+) of 2 animal CFSE- labelled PBMCs stimulated with low (L), medium (M) and high (H) concentrations of TLR1-9 agonists as measured by flow cytometry 
Figure 5.37: Proliferation (percentages CFSE+) of 3 bovine CFSE-labelled PBMCs stimulated with high concentrations only of TLR1-9 agonists as measured by flow cytometry169
Figure 5.38: CFSE-labelled CD4 T cells stimulated with TLR ligands showing gated proliferative cells upon stimulation (animal Bov1)171
Figure 5.39: Proliferation (percentages) of bovine 1 and 2 CFSE-labelled CD4 T cells stimulated with 3 concentrations of TLR3, TLR8, and TLR9 agonists as measured by flow cytometry

Figure 5.40: Proliferation (percentages) of bovine 4,5 and 4098 CFSElabelled CD4 T cells stimulated with 2 concentrations of TLR3, TLR8, and TLR9 agonists as measured by flow cytometry. .... 173 Figure 5.42: Proliferation (percentages) of bovine CFSE-labelled CD8 T cells stimulated with high, medium and low concentrations of TLR3, TLR8, and TLR9 agonists as measured by flow cytometry.

- Figure 5.43: Proliferation (percentages) of bovine CFSE-labelled CD8 T cells stimulated with high, medium and low concentrations of TLR3, TLR8, and TLR9 agonists as measured by flow cytometry.
- Figure 5.45: CFSE-labelled  $\gamma\delta$  T cells stimulated with TLR ligands showing gated proliferative cells upon stimulation (animal Bov1)......179

- Figure 5.52: CFSE-labelled PBMCs from Bov 7 mixed with both unstimulated and activated  $\gamma\delta$  T cells separately......190

- Figure 5.55: Fixed number of CFSE-labelled PBMCs (5 ug/ml ConA activated for 3 days) at days 3 (after activation) and 5 (after two days of incubation in the assay) from Bov 7 mixed with different numbers of unstimulated purified  $\gamma\delta$  T cells separately.......194
- Figure 5.57: Proliferation (percentages) of CFSE-labelled PBMCs (5 ug/ml ConA activated for 3 days) mixed with unstimulated purified  $\gamma\delta$  T cells (for two days of the assay) separately in Bov 6 and Bov 7 blood animals showing the stimulatory role of  $\gamma\delta$  T cells......196
- Figure 6.58: CXCL-8 concentrations measured in the supernatant of PBMCs cultured in response to TLR1-9 agonists in 7 animals..........211
- Figure 6.60: CXCL-8 concentrations measured in the supernatant of CD8 T cells cultured in response to TLR3, TLR8, and TLR9 agonists in 4 animals. a) TLR3 agonist. b) TLR8 agonist. c) TLR9 agonist. .217
- Figure 6.61: CXCL-8 concentrations measured in the supernatant of  $\gamma\delta$  T cells cultured in response to TLR3, TLR8, and TLR9 agonists in 4 animals. a) TLR3 agonist. b) TLR8 agonist. c) TLR9 agonists. 220
- Figure 6.62(a-d): IFN-a concentrations measured in the supernatant of PBMCs cultured in response to TLR1-9 agonists in 7 animals . 223
- Figure 6.64 (a-i): TNF-a concentrations measured in the supernatant of PBMCs cultured in response to TLR1-9 agonists. CD4, CD8 and  $\gamma\delta$  T cells did not produce any TNF-a in the supernatant......230

- Figure 6.72 (a-i, TLR ligands 1-9 respectively): IFNγ concentrations measured in the supernatant of PBMCs cultured in response to TLR1-9 agonists in 7 animals......274

## List of abbreviations

Abs	Antibodies		
ALRs	Absent in Melanoma 2 (AIM2)-Like Receptors		
Amp	Ampere		
ANOVA	Analysis Of Variance		
APC	Allophycocyanin		
BCR	B Cell Receptor		
B-ME	Beta-Mercaptoethanol		
bp	Base Pair		
BSA	Bovine Serum Albumin		
CD	Cluster of Differentiation		
cDNA	Complementary DNA		
CFSE	Carboxyfluorescein Succinimidvl Ester		
CLRs	Transmembrane C-Type Lectin Recentors		
CMI	Cell-Mediated Immunity		
Con A	Concanavalin A		
Conc	Concentration		
Cq	Quantification Cycle		
CTLs	Cytotoxic T Cells		
CTV	CellTrace Violet		
DAMPs	Damage-Associated Molecular Patterns		
DCs	Dendritic Cells		
ddH <sub>2</sub> O or (DDI	Double Distilled Water / or Double Ionised Distilled		
$H_2O)$	Water		
DMSO	Dimethyl Sulfoxide		
DNA	Deoxyribonucleic Acid		
DPBS	Dulbecco's Phosphate Buffered Saline		
dsRNA	Double Stranded RNA		
DTH	Delayed Type Hypersensitivity		
EDTA	Ethylenediamintetraacetic Acid		
ELB	Erythrocyte Lysis Buffer		
ELISA	Enzyme-Linked Immunosorbent Assav		
FACS	Fluorescent Antibody Cell Sorting		
FAM	6 Carboxyfluorescein		
FCS / FBS	Foetal Calf Serum / Foetal Bovine Serum		
FITC	Fluorescein Isothiocvanate		
FOXP3	Forkhead Box P3		
FSC	Forward Scatter		
Fwd Primer	Forward Primer		
GAPDH	Glyceraldehyde 3-Phosphate Dehydrogenase		
gd or γδ T cells	Gamma-Delta T Cells		
gDNA	Genomic DNA		
HEK293	Human Embryonic Kidney 239		
HI	Humoral Immunity		
hr	Hour		
IFN-γ	Interferon Gamma		
IgA, IgG, IgM	Immunoglobulin A, G, M		
ΙΚΚ-β	I-Kappa B Kinase Complex		
IL-	Interleukin-		

IMDM	Iscove's Modified Dulbecco's Medium		
IRAK	Interleukin-1 Receptor-Associated Kinase		
IU	International Unit		
kb	Kilobase		
L	Litre		
L-glu	L-glutamin		
LN2	Liquid Nitrogen		
LRR	Leucine-Rich Repeat		
MACS	Magnetic Antibody Cell Sorting		
MAPK	Mitogen-Activated Protein Kinase		
MHC	Major Histocompatibility Complex		
min	Minute		
MKK6	Mitogen-Activated Protein Kinase (MAPK) Kinase-6		
ml	Millilitre		
MLN	Mesenteric Lymph Nodes		
mМ	Millimolar		
MØs	Macrophages		
MS	Microsoft Office		
MyD88	Myeloid Differentiation Primary Response Gene 88		
NCBI	National Centre for Biotechnology Information		
NEB / UK	New England Biolabs / United Kinadom		
NFκB	Nuclear Factor Kappa B		
ng	Nanogram		
NK cells	Natural Killer Cells		
NK T cells	Natural Killer T Cells		
NLRs	Nucleotide-Binding Oligomerization Domain (Nod)-		
	Like Receptors		
nm	Nanometre		
nM	Nanomolar		
NS	Non-Stained		
nsSNP	Non-Synonymous SNP		
NTC	No Template Control		
PAMPs	Pathogen-Associated Molecular Patterns		
PBMCs	Peripheral Blood Mononuclear Cells		
PBS	Phosphate Buffered Saline		
PE / or R-PE	Phycoerythrin / Or Red-Phycoerythrin		
Penstrep	Penicillin/Streptomycin		
PFA	Paraformaldehyde		
PHA	Phytohaemagglutinin		
PMA	Phorbol 12-Myristate 13-Acetate		
PMN cells	Polymorphnuclear Cells		
Pmol	Picomolar		
PRRs	Pattern Recognition Receptors		
RBCs	Red Blood Cells		
Rev Primer	Reverse Primer		
RLRs	Retinoic Acid-Inducible Gene (RIG)-I-Like Receptors		
RNA	Ribonucleic Acid		
RNS	Reactive Nitrogen Species		
RO water	Reverse Osmosis Water		
ROS	Reactive Oxygen Species		

RPLP0	Ribosomal Protein, Large P0		
RPLP2	Ribosomal Protein, Large P2		
RPMI	Roswell Park Memorial Institute		
RT-qPCR	Quantitative Real-Time PCR		
SARM	Sterile-Alpha and Armadillo-Motif-Containing Protein		
SDHA	Succinate Dehydrogenase		
sec	Seconds		
SNP	Single Nucleotide Polymorphism		
SRs	Scavenger Receptors		
SSC	Side Scatter		
ssRNA	Single Stranded RNA		
TAE buffer	Tris-Acetate-Edta Buffer		
TAMARA	N,N,N,N'-Tetramethyl-6-Carboxyrhodamine		
TCR	T Cell Receptor		
Th1, Th2, Th17	T Helper 1, T Helper 2, T Helper 17		
TIR	Toll/IL-1 Receptor		
TIRAP	Toll-Interleukin 1 Receptor (TIR) Domain Containing		
	Adapter Protein		
TLRs	Toll-Like Receptors		
TNF-a	Tumour Necrosis Factor alpha		
TRAF6	Tumour Necrosis Factor (TNF) Receptor Associated		
	Factor-6		
TRAM	TRIF-Related Adaptor Molecule		
Tregs	Regulatory T Cells		
TRIF	TIR-Domain-Containing-Adapter-Inducing Interferon-		
	beta		
u/ul	Unit/Microliter		
ug	Microgram		
ul or uL	Microliter		
uM	Micromolar		
UPL probes	Universal Probe Library Probe		
UV light	Ultraviolet Light		
V	Volt		
WEASEL	Walter & Eliza Analysis Software: Eclectic & Lucid		

**Chapter One** 

**Introduction and** 

**Literature Review** 

#### 1.1 The immune response

The immune response is essential for the protection of animals (including humans) against pathogenic microbial attacks and other disease processes such as cancer (Chaplin, 2010). It is grossly divided into two components; innate immunity which is non-antigen specific, and involves cells, such as macrophages (MØs), dendritic cells (DCs), Natural killer (NK) cells, mast cells, and epithelial cells at tissues exposed to the exterior (the gut, lung and skin) that respond rapidly to infection or changes in tissues. It is usually of short duration (Galli et al., 2011, Kelly and O'neill, 2015, Whitsett and Alenghat, 2015, Man and Kanneganti, 2016). The second main component of immunity is the adaptive immune response (also called acquired immunity) involving T and B cells as well as their subsets, which are highly specific for antigens, are generated in lymph nodes and show immunological memory. This means they can be recalled after several years or decades to respond rapidly and specifically when, for example, an infection occurs that the host has experienced before (Mantovani et al., 2011, Lanier, 2013).

#### 1.2 The innate immune response

Innate immunity is the starting point for foreign microbe recognition by the host. It is the front-line defence against invading pathogens or disease-related tissue changes. The innate immune cells at the site of microbial entry recognise relatively conserved molecules of the microbes not shared by the host (described in detail below). These pattern recognition receptors (PRR) on innate immune cells activate the cells (for example epithelial cells, monocytes / MØs, DCs) to produce a variety of mediators such as

pro-inflammatory cytokines, chemokines, and antimicrobial peptides (Medzhitov, 2001, Beutler, 2004). The cells involved in innate immune responses are listed in table 1.1.

Innate mechanisms of defence against pathogen infection or diseaseinduced change also include: apoptosis (controlled programmed cell death) (Roos *et al.*, 2004); opsonisation and phagocytosis (Greenberg and Grinstein, 2002); the complement system (Rus *et al.*, 2005) and clotting process (Engelmann and Massberg, 2012). Other components of innate immunity include the antimicrobial activity of normal gut flora (Boman, 2000); non-specific production of immunoglobulin A (IgA) from epithelial cell lining mucosal sites that access the exterior (Bidgood *et al.*, 2014); secretion of antimicrobial peptides such as defensins and lysozymes (Ouellette, 2011, Leysen *et al.*, 2013); cytolysis of target pathogeninfected cells by natural killer cells (NK cells) (Vivier *et al.*, 2011).

The proinflammatory cytokines released by innate immune cells include: interleukin-1 (IL-1); IL-6; the chemokine CXCL8 (also known as IL-8); tumour necrosis factor alpha (TNF- $\alpha$ ) and interferons including interferon gamma (IFN- $\gamma$ ), which is produced in abundance by NK cells. These are defined as low molecular weight proteins (usually glycoproteins) that attract, activate or differentiate leucocytes to stimulate inflammation and fever to create unsuitable environments for pathogen survival (Xu *et al.*, 2010, Abbas *et al.*, 2014).

Although multiple strategies for microbial elimination are involved in innate immunity, phagocytosis is an important element of innate immune responses. Neutrophils and mononuclear cells (for example MØs and some

DCs) have the ability to engulf pathogens that exceed 1  $\mu m$  in size (Medzhitov, 2001, Singh and Singh, 2013).

# **Ch1: Introduction**

#### Table 1.1: Major cells involved in the innate immune response.

Type of Cell	Description and function	Citation
Monocytes	Phagocytosis of circulating pathogens in blood, antigen presentation and	(Faist <i>et al</i> ., 1988, Gordon
	chemotaxis of other immune cells via releasing cytokines/chemokines.	and Taylor, 2005, Domínguez
	Differentiation into MØs in tissues.	and Ardavín, 2010).
Macrophages	Pathogen engulfment and presentation of antigens, releasing cytokines,	(Morrissette <i>et al</i> ., 1999,
(MØ)	healing process and scavenging after tissue damage, apoptosis.	Nadella <i>et al</i> ., 2012).
Kupffer cells	Liver MØs: see macrophage function.	(Movita <i>et al</i> ., 2012).
Microglial cells	Brain MØs: see macrophage function.	(Faustino <i>et al</i> ., 2011).
Dendritic	Efficient pinocytosis of pathogens (immature form) and presenting them	(Banchereau and Steinman,
cells (DC)	(mature form), to T cells as antigens with MHC molecules. Cytokine release.	1998, Geissmann <i>et al</i> .,
	Initiating and maintaining immune responses. Immune tolerance.	2010).

Mast cells	Allergic reactions (Type 1 hypersensitivity mediated by IgE in particular);	(Kalesnikoff and Galli, 2008,
	source of histamine, vasoactive amines, prostaglandins and leukotrienes.	Silver and Curley, 2013).
	Tissue modulation.	
Natural	Cytolysis of target, early sensors for viral and/or tumour attack.	(Shi <i>et al</i> ., 2011, Vivier <i>et al</i> .,
Killer (NK)	Inflammation, cytokine release (particularly IFN-gamma.	2011).
cells		
Neutrophils	Found in profuse amounts in blood, early response to some microbes, mainly	(Wright et al., 2010, Amulic
	bacteria. Sepsis formation (dead and alive); acute inflammation	<i>et al.</i> , 2012).
	phagocytosis; opsonisation; short acting, release of cytokines, secretion of	
	antimicrobial peptides (e.g. lysozymes that degrade cell walls of bacteria).	
Eosinophils	Increased in blood and tissues in parasitic infestation; allergic reactions;	(Akuthota <i>et al.</i> , 2010, Lloyd
	inflammation; antigen presentation; interaction with mast cells; tissue	and Saglani, 2013, Fajt <i>et al</i> .,
	modulation; involved in asthma.	2014),

7
۳.
=
D
-
2
•
7
0
<b>-</b>
3
ō
ŏ
ă

-
Ω
<b>_</b>
н
н
3
Ŧ
0
D
2
Q
Ξ.
0
3

Basophils	Less than 1% of total blood cells; allergic reactions associated with IgE;	(Ishida <i>et al</i> ., 2011, Siracusa		
	inflammation; production of cytokines; enhances T helper 2 progression.	<i>et al</i> ., 2011, Voehringer,		
		2013).		
Epithelial cells	Work as natural filters due to their existence on surfaces of lining layers of	(Peterson and Artis, 2014,		
	canals or tracts of many body systems exposed to exterior such as	Artis and Spits, 2015,		
	respiratory, digestive, urinary may allow expulsion and elimination of	Whitsett and Alenghat, 2015)		
	microbes by ciliary movements; non-specific Ab-Ag binding via IgA secretion.			

#### 1.2.1 Innate immune cell pattern recognition receptors

An important aspect of innate immune cells is their ability to recognise relatively-conserved components of microorganisms and initiate a rapid inflammatory response invasion or infection. Epithelial cells and dendritic cells amongst others express PRRs (pattern recognition receptors) that perform this function. They recognise pathogen-associated molecular patterns (PAMPs) on microbes and damage-associated molecular patterns (DAMPs) which are host related (Hayashi *et al.*, 2010, Takeuchi and Akira, 2010).

Six classes of pattern recognition receptors (PRRs) have been listed which recognise different PAMPs (see table 1.2). These are: cytoplasmic retinoic acid-inducible gene (RIG)-I-like receptors (RLRs); absent in melanoma 2 (AIM2)-like receptors (ALRs); transmembrane C-type lectin receptors (CLRs); scavenger receptors (SRs); nucleotide-binding oligomerization domain (NOD)-like receptors (NLRs); and Toll-like receptors (TLRs) (Areschoug and Gordon, 2009, Sattler *et al.*, 2012, Singh and Singh, 2013). Viruses, bacteria, parasites and other microbes express PAMPs that can be sensed and distinguished by PRRs in immune cells as non-self antigens and thus initiating immune response to infection or tissue damage/ change (Kumar and Bot, 2013).

These PAMPs can be detected by different recognition patterns initiating multiple signalling pathways, i.e. RIG-I-like receptors (RLRs) are responsible for sensing of RNA viruses in humans resulting in triggering innate immune response as well as inflammation throughout IFN gamma (Loo and Gale Jr, 2011). Another group of PRRs, AIM2-like receptors plays a fundamental role in sensing bacterial and viral DNA (Rathinam *et al.*,

2010). The next group of PRRS is the C-type lectin receptors (CLRs)– expressed by DCs and MØs – which are responsible for recognition of bacterial and fungal carbohydrates (Yabe *et al.*, 2014). In addition, scavenger receptors (SRs) expressed by mononuclear phagocytes (MØs and DCs) have the capability to detect or eliminate modified lipoproteins (the major constituent in both Gram negative and Gram-positive bacteria) thus enhancing innate immunity (Areschoug and Gordon, 2009, Canton *et al.*, 2013). Moreover, NOD-like receptors (NLRs) play a key role in detecting intracellular microbes via induction of inflammation and apoptosis (Kersse *et al.*, 2011). Finally, Toll-like receptors (TLRs), which will be the group of interest in this study, have been involved in a very wide range of sensing the variety of bacterial, viral, fungal, and protozoal PAMPs triggering inflammation and then innate immune responses (Kawai and Akira, 2010, Tartey and Takeuchi, 2014). The TLRs will be discussed in more detail below starting from section 1.5.

However, innate immunity is not always effective against pathogens; some fastidious microbes can cross over these barriers, and this is where the more specialised type of immune response named as adaptive immunity comes in (Beutler, 2004, Hoebe *et al.*, 2004). Innate immunity effector function can inform the type of adaptive immune response that is generated to infection or disease.

Table 1.2: Non TLRs pathoger	n recognition pattern classes.
------------------------------	--------------------------------

PRRs	Target(s)	Function(s)	References
RIG-I-like	RNA viruses	Stimulation of innate immunity and inflammation via	(Loo and Gale Jr,
receptors (RLRs)		IFN-γ secretion	2011)
AIM2-like	Bacterial and viral DNA	Activation of inflammasome and cell death by induction	(Rathinam <i>et al</i> .,
receptors (ALRs)		of proinflammatory cytokines (IL-1 $\beta$ and IL-18).	2010)
C-type lectin	Bacterial and fungal	Enhance APCs (mainly DCs & MØs) functions as well as	(Figdor <i>et al</i> .,
receptors (CLRs)	carbohydrates	regulation of DCs migration and their interaction with T	2002)
		cells	
Scavenger	Varied, but mainly	Phagocytosis of varieties of foreign bodies especially	(Areschoug and
receptors (SRs)	bacterial lipoproteins	bacterial.	Gordon, 2009)
NOD-like	peptidoglycan of	Higher NLRs levels could indicate stress. Stimulation of	(Kersse <i>et al</i> .,
receptors (NLRs)	intracellular bacteria	inflammation and apoptosis.	2011)

#### 1.3 The adaptive immune response

Adaptive (acquired) immunity develops after an initial innate immune response, where T cells (involved in cell-mediated immunity) and B cells (involved in humoral immunity) are generated in lymph nodes (Bonizzi and Karin, 2004, Ferrand and Ferrero, 2013). The adaptive immune response confers specificity and memory on immune responses. To generate adaptive immune responses to infection, the following sequence of events typically takes place. The pathogen infects and is recognised by the innate immune system that generates inflammation, activation and recruitment of cells, some of which are involved in generating the adaptive immune response to infection. Pathogens are processed by innate immune antigenpresenting cells such as MØs and immature DCs. DCs are very important for generating and maintaining immune responses. As they mature after taking up pathogen or pathogen fragments, they migrate to local lymph nodes and become very good at presenting antigen fragments coupled to major histocompatibility complex molecules (MHC class I and MHC class II, see below) to T and B cells (Palm and Medzhitov, 2009).

The MHC molecules are highly variable (polymorphic) and differ between individuals in outbred animal populations. MHC class I molecules are present on all nucleated cells and act as restriction elements for antigen presentation for intracellular pathogens (for example all viruses, some bacteria) to cytotoxic T cells in particular (CTLs, see below). The CTLs will then destroy the infected cell. The MHC class II molecules are mainly on antigen presenting cells such as monocyte / MØs, DCs and B cells, and are restriction elements for antigen from extracellular invading pathogens to helper T cells in particular (see below). The MHC molecules are involved in

the thymus in making sure that self-reactive T cells do not develop. Their role in antigen presentation is to ensure that only pathogen-specific responses are generated (Comber and Philip, 2014, Fooksman, 2014).

As stated above, acquired immunity consists of two major compartments; humoral immunity and cell-mediated immunity (CMI). Humoral immunity (HI) is mainly dependent on stimulated / activated B cells (the name was derived from Bone marrow in mammals and Bursa of fabricious in avian as they are synthesised there) which have a unique receptor called the B cell receptor (BCR) expressed on the outer membrane (Michael, 1995).

B cells have detailed antigen specificity and develop into either effector plasma cells that secrete antibody, or into memory cells that circulate, ready to respond rapidly to infection with the same pathogen antigens that stimulated their development. Antibodies in mammals belong to five immunoglobulin classes: IgM, IgG, IgA, IgD and IgE. IgG is further divided into several subclasses, depending on the host species (Schroeder Jr and Cavacini, 2010).

Immature B cells express IgM, while IgM and IgD are displayed on more mature B cells to enable their infiltration and access to lymphoid tissues (Yuan and Witte, 1988, Brezski and Monroe, 2008). Mature B cells undergo differentiation into plasma cells or memory cells under stimulation by expressed antigen and regulatory cytokines (IL-4, IL-5, IL-10 and IL-13, IFN-gamma) secreted by Ag presenting and T helper cells (tables 1.3 and 1.4).

Plasma cells are responsible for maintaining secretion of immunoglobulins at about 10,000 Ig/second (Abbas *et al.*, 2014). On the other hand, memory B cells are responsible for re-stimulation of the immune system

when a recurrent infection with the same antigen is encountered (Weill *et al.*, 2013).

The second compartment of adaptive immunity is cell-mediated immunity (CMI). CMI depends on T lymphocytes (named so because they develop and mature in the Thymus). The T cell receptor (TCR) recognises antigen fragments (epitopes) and is made up of constant and variable gene segments – the alpha and beta gene products. Cluster of differentiation 3 (CD3) is the common marker for T cells which binds with TCR forming the TCR complex (Abbas et al., 2014). The two originally-described principal types of T cells are helper T cells (that express the CD4 molecule and use the MHC-II pathway) that are involved in helping B cells and other effector T cells, like Cytotoxic T cells (CTLs), to develop and function in immune responses, and CTLs (that express the CD8 molecule), which kill infected cells by the MHC-I restriction pathway. A third type of T cell uses a different configuration of the T cell receptor that uses gamma and delta TCR chains rather than alpha and beta ones. More recently the CD4+ T helper cell subset has been divided into several subtypes - Th1, Th2, TH17 and regulatory T cells (that can also express CD8) (Delves *et al.*, 2011).

T helper 1 (Th1) cells are involved in cell-mediated immune responses to many pathogens, particularly intracellular ones. They are responsible for delayed type hypersensitivity (DTH) reactions to these. Th1 develop in the presence of the cytokine s IL-12 and IFN-gamma. Cytokines released from activated Th1 include IL-2, IFN- $\gamma$  and TNF- $\beta$  (table 1.4) amongst others that activate MØs for pathogen clearance and generate antibody responses with certain IgG subclasses (that differ between the different species of animal). Th2 cells are involved in allergic reactions and immune responses

to nematode parasites. They also help B cells make antibodies of the IgE class and some IgG subsets (depending on species, e.g. IgG1). Th2 release IL-2 in common with Th1 cells but are the principal T cell source of IL4, IL-5, IL-10 and IL13 (table 1.4) that stimulate the recruitment and activation of mast cells, eosinophils amongst others at the site of infection (Beckmann *et al.*, 1992, Viallard *et al.*, 1999).

T helper 17 (Th17) cells are a further subset of T cells developed from CD4 T cell differentiation that secretes IL-17 and IL-22 involved in neutrophil accumulation amongst other activities. This pathway is important in response to inflammation caused by intracellular bacteria (Delves *et al.*, 2011) (Table 1.4).

CTLs are involved in the killing (by apoptosis induction mainly) of cells infected by intracellular pathogens. They require antigen presentation restricted by MHC class I molecules, but their activity can be enhanced by the activation of Th1 cells as helper cells. There is a principal mechanism where CTLS kill target cells. The perforin-granzyme pathway involves poreforming molecules called perforin to puncturing the infected cell allowing the access of granzyme enzymes that induce apoptosis in the cell (Podack and Dennert, 1983, Tschopp and Nabholz, 1990).

Although naïve T cells develop and differentiate into either CD4+ or CD8+ effector T cells which, in turn, proliferate to increase their numbers to fight specific pathogens, they subsequently undergo death and reduction of their numbers to establish homeostasis inside the body (Abbas *et al.*, 2014). Nonetheless, a small number of these cells persist and differentiate as memory T cells, which are responsible for rapid recall responses to

encountering the same pathogen antigens again (Sallusto *et al.*, 1999, Masopust *et al.*, 2001).

A further subset of T cells is the gamma-delta T cells that use the gamma and delta chains of the T cell receptor rather than the alpha and beta ones (Holtmeier and Kabelitz, 2005). This subset of T cells represents the least common T cells (up to 5% in blood), whereas they are more abundant in epithelial surfaces (25-60% in the gut). Although these cells had been described as of unknown function, the distinct  $\gamma\delta$  TCR can be used as a pathogen recognition receptor, thus it links innate and adaptive immunity. These cells are part of the CD4-CD8- T lymphocytes in peripheral blood; they recognise pathogens via non-classical MHC class I and MHC class II initiating cell cytotoxicity. Furthermore, these cells can neutralise intestinal, cutaneous and genital epithelium pathogens (Li *et al.*, 1998, Carding and Egan, 2002, Dar *et al.*, 2014). Termination of the immune response process is regulated by specific subsets of T cells called regulatory T cells (Tregs) formerly called T suppressor cells (Beissert *et al.*, 2006).
Table 1.3: Major	cells involved i	n the adaptive	immune response.
------------------	------------------	----------------	------------------

Type of Cell	Function	Surface	Cytokine	Citation
		marker	secretion	
B lymphocytes	Responsible for adaptive humoral immunity involving	CD19, CD20,	IL2, IL4,	(Lund and Randall,
(B cells)	antibody production (IgM, IgG, IgA, IgE, IgD) through	CD34, CD38,	TNFa, IL6,	2010, Yang <i>et al</i> .,
	differentiation to plasma cells; antigen presentation;	and CD45R	IFNy, IL12,	2010, Mauri and
	production of memory cells.		and TNFa	Bosma, 2012)
Plasma	Generated from B cells. They are the antibody-secreting	CD138, CD78	IL21, IgG3	(Calame, 2001)
cells	cells.			
T lymphocytes	Responsible for adaptive cellular immunity; three major	CD3, CD4,	Many	(Abbas <i>et al</i> ., 2014)
(T cells)	classes – helper T cells (CD4+, MHC class –II restricted),	CD8	Interleukins,	
	cytotoxic T cells (CD8+, MHC class I restricted), and $\gamma\delta$ T		e.g. IL-2,	
	cells (non-classical MHC class I and class II restricted).		IFN-γ, IL-4,	
	Regulatory T cells (CD4, many are FOXP3+), and memory		TGF -beta,	
	T cell subsets (see below)		IL-10.	
T helper 1	Mainly direct adaptive immunity to drive cell-mediated	CD4+CCR5	E.g. IL1, IL2,	(Viallard <i>et al</i> ., 1999)
(Th1)	immunity and provide B cell help for antibody production,		and IFNy	

	release cytokines (IL-2, IFN- $\gamma$ and TNF- $\beta$ ), induction of			
	phagocytosis via activation of MØs, DCs and help for CTL			
	cells.			
Th2	Chiefly guide adaptive immunity against extracellular	CD4+CCR4	IL4, IL5,	(Delves <i>et al</i> ., 2011)
	pathogens, and involved in anti-nematode parasite immune		IL10, IL13	
	responses; release of IL-4, IL-5, IL-10 and IL13 driving			
	humoral immunity pathway by stimulation of B cells, IgG1			
	and IgE production, class switching, stimulation of			
	hematopoiesis			
Th17	Induction of inflammation via IL-17 secretion; recruitment	CD4+CCR6	IL17	(Ouyang <i>et al</i> ., 2008,
	and activation of neutrophils; antimicrobial activity via IL-			Lee <i>et al</i> ., 2009)
	22 released at mucosal surfaces			
Regulatory T	Mainly CD4+ CD25+ FOXP3+ T cells, but other subsets	CD4, CD25,	TGF, IL-10	(Sakaguchi <i>et al</i> ., 2008)
cells (Tregs)	exist. Control (supress) immune responses and	CD127, and		
	autoimmunity, tissue grafts and some cancers by	FOXP3		
	termination of immune responses and establishing			
	equilibrium and homeostasis. Secretion of inhibitory			
	cytokines such as IL-10 and TGF- $\beta$			

Memory T cells	Re-stimulation to generate T effector cells and B cell help	CD4, CD8	IL7, IFNγ,	(Farber <i>et al</i> ., 2014)
	on exposure to previously encountered pathogen antigens.		TNF-a	
Cytotoxic T	Antiviral and antitumor activities. Recognition of	CD8	IL1, TNF-a	(Barry and Bleackley,
lymphocytes	intracellular pathogen antigens in association with MHC I.		and IFN-γ	2002)
(CTLs)				
γδ T cells	Non-classical MHC class I and class II restricted. Act directly	γδ TCR	IFN-γ, IL4,	(Chien <i>et al</i> ., 2014,
	when exposed to antigen by secretion of either IL-17		IL10, and	Guzman <i>et al</i> ., 2014,
	causing inflammation or IFN gamma release when exposed		IL17	Hovav, 2017)
	to viruses or cancer (amongst other responses).			
	Immunosuppression (thought to be a regulatory T cell			
	subset in cattle).			

## Table 1.4: Main functional cytokines released from immune cells.

Cytokine	Description	References
Interleukin-1	Released by monocytes, MØs, DCs, T cells, B cells, neutrophils, glial cells,	(Dinarello, 2009,
alpha and beta:	endothelium, myocytes, fibroblasts, NK cells and keratinocytes.	Dinarello, 2011).
(IL-1a) and	Acts as T helper 2 (Th2) stimulation; B and T cell maturation; NK cell activator;	
(IL-1β)	induces pyrexia (endogenous pyrogen); takes part in acute inflammation,	
	endothelial cell activation.	
IL-2	Released by activated Th cells.	(Gaffen and Liu,
	Acts as activator and proliferator for T, B and NK cells.	2004).
IL-3	Produced by T helper cells and promotes stem cells progression and	(Llop-Guevara <i>et al</i> .,
	diversification along with SCF; induction of mucosal mast cell development.	2011).
IL-4	Secreted by Th2 cells, MØs, mast cells, basophils.	(Beckmann <i>et al</i> .,
	Stimulates B cells for IgG1 and IgE production, propagation and differentiation	1992, Van Dyken and
	of stimulated B cell (class switching); stimulate Th2 cell development.	Locksley, 2013).

IL-5	Produced by Th2 cells, eosinophils and mast cells.	(Mainou-Fowler et al.,
	Proliferation and differentiation of eosinophils and activated B cells; IgG1 and	1994, Molfino <i>et al</i> .,
	IgE production;	2012).
IL-6	Produced by monocytes, MØs, fibroblasts, endothelial cells, Th2 cells, mast cells	(Scheller <i>et al</i> ., 2011,
	and hepatocytes. Also acts as a pro-inflammatory cytokine; maturation of B	Garbers <i>et al</i> ., 2012).
	cells into plasma cells; stimulates plasma cells to produce antibodies;	
	differentiation of stem cells; stimulates platelet synthesis (clotting formation);	
	triggers liver to produce acute phase proteins;	
IL-7	Released by thymocytes and bone marrow cells.	(Fry and Mackall,
	Acts on stem cell differentiation into B and T progenitor cells; enhances	2002, Huang and
	monocyte and MØ killing mechanisms; activates T cytotoxic cells (CTLs).	Luther, 2012).
CXCL-8	Chemokine; Produced by monocytes, MØs, endothelial and epithelial cells, T	(Meade <i>et al</i> ., 2012).
(formerly IL-8)	cells, neutrophils, hepatocytes, keratinocytes, chondrocytes, fibroblasts.	
	Acts as a neutrophil chemotactic factor, which activates them to produce	
	lysozymes; supports phagocytic activity of mononuclear cells; angiogenesis.	

IL-10	Produced by mononuclear cells, regulatory T cells, T helper 2 cells, B cells, (and	(Sabat <i>et al</i> ., 2010,
	$\gamma\delta$ T cells in cattle).	Guzman <i>et al</i> ., 2014).
	Immunosuppression (termination of immune response), known to work	
	together with TGFβ	
IL-17	Synthesised by Th17 CD4+ T cells and $\gamma\delta$ T cells in humans and cattle.	(Peckham <i>et al</i> .,
	Multifunctional cytokine such as induction of inflammation, allergy-associated	2014, Gaffen, 2016,
	cytokine, stimulation of other immune cells to produce a wide range of	Steinbach <i>et al</i> .,
	cytokines during TB infection in combination with IL-22.	2016)
Tumour necrosis	Produced by MØs, NK and mast cells, T and B cells.	(Aggarwal <i>et al</i> .,
factor alpha &	Acts as antitumor cytokine; proinflammatory; apoptosis; hematopoietic cell	2012).
beta	proliferation; angiogenesis; B cell maturation and Ab synthesis; neutrophil	
(TNF-a) & (TNF-β)	activation; phagocytosis.	
Interferon gamma	Secreted by majority of lymphocytes, particularly Th1 cells and NK cells.	(Young and Hardy,
(IFN-γ)	Antiviral and antitumor activities; activates class switching in B cells (IgG);	1995).
	enhances MHC I and MHC II expression and antigen presentation enhances cell	
	cytotoxicity.	

IFN-a &	Type I interferons. Various leukocytes release IFN-a as an antiviral and	(Taniguchi and
IFN-β	antitumor mediator, while IFN- $\boldsymbol{\beta}$ is released by fibroblasts with the same	Takaoka, 2001,
	function as IFN-a.	Belardelli <i>et al</i> .,
		2002).
Transforming	Approximately shares similar secretion cites and functions with those	(Calon <i>et al</i> ., 2014,
growth factor beta	mentioned in IL-10 plus a few extra functional roles in reaction to cancer cells	Meng <i>et al</i> ., 2016)
(TGFβ)	and initiation of fibrosis in chronic diseases.	

## 1.4 Regulatory T cells (Tregs)

Peripheral immune tolerance (unresponsiveness status of the immune system represented by lymph nodes to a certain antigen or immunogen) is controlled by regulatory T cells (Tregs), which are an important category of T cells involved in down-regulating immune responses at the end of their effector phase in order to prevent tissue damage. As unregulated immune responses are seen in autoimmune reactions, Tregs are important in controlling these diseases (Sakaguchi *et al.*, 1995, Takahashi *et al.*, 2000). They were originally identified as CD4+CD25+FOXP3+ cells, where CD25 is the a-chain of the IL-2 receptor which is expressed on most activated T cells. FOXP3 is a transcription factor involved in Treg development and function (Sakaguchi *et al.*, 2009, Kassiotis and Liston, 2011).

Two types of Tregs have been described; natural CD4+CD25+ Tregs (nTregs) and induced (iTregs) Tregs (or sometimes called adaptive Tregs) both of which express the transcriptional factor forkhead box P3 (FOXP3) expressed on the inner nuclear membrane (Hori *et al.*, 2003, Rudra *et al.*, 2012). nTregs originate in the thymus and iTregs develop out of it (Curotto de Lafaille and Lafaille, 2009). nTregs form 5-10% of a typical overall T cell population in humans and mice as they develop from naïve CD4+ T lymphocytes (Shevach, 2002). They are characterised by their anergic properties *in vitro* (lack of response to TCR of T cells), while they are able to suppress CD4+ and CD8+ T cell subset growth and cytokine release (Takahashi *et al.*, 1998, Piccirillo and Shevach, 2001).

iTregs (CD4+CD25+FOXP3+) are induced during the later stages of immune responses and are the principal Tregs which take part in different immuno-suppressive mechanisms on effector immune cells. Their activities

include inhibition of target cell proliferation and cytokine production. They are active against a range of immune cells including CD4+ effector T cells and CD8+ effector T lymphocytes, NK cells as well as phagocytic cells and APCs, such as monocyte / MØ lineage cells (Sakaguchi *et al*., 2008). These suppressive activities are mediated mainly by two suppressive cytokines, TGF-beta and IL-10.

Induced Tregs require IL-2 for their growth, maintenance and function (Chen *et al.*, 2011, Visan, 2014). In addition, TGF- $\beta$  is required to generate FOXP3 gene expression which changes CD4+CD25– naïve Tregs into active CD4+CD25+ Tregs and full function (Chen *et al.*, 2003).

Vignali *et al.* (2008) suggested the division of Tregs into four functional groups depending on the cytokines they produce; the first group releases IL-10, IL-35 (a potent immunosuppressive cytokine secreted by Tregs that inhibits T cell proliferation) and TGF- $\beta$ , as immuno-suppressive cytokines; the second group uses cytolysis by releasing granzyme A or B as well as pore forming protein (perforin) to induce apoptosis in the target cells; the third group cause metabolic disruption which influences mainly effector T cells by interference with cyclic AMP (cAMP) pathway or deprivation of cytokines leading to apoptosis; and the final group modifies DC maturation and/or function by CD223-MHCII engagement which inhibits DC maturation.

## **1.5 Toll-like receptors (TLRs): A bridge between**

## innate and adaptive immunity

Pathogen recognition receptors (PRRs) consist of several families of receptors of which the TLR group is the interesting group for this (my) study. The Toll receptor was discovered firstly in *Drosophila* insects more than 25 years ago as an immune defence factor (Hashimoto *et al.*, 1988). Later, it was found that a homologue toll-like receptor in humans existed that also had similarity in terms of structure with the human IL-1 receptor which provided a link to a role in innate immunity in vertebrates (Rock *et al.*, 1998, Aderem and Ulevitch, 2000). In mammals, it has been shown that TLRs play an important role in innate immunity and also informing adaptive immune responses (Akira, 2003). 10 types of TLR have been recognised in humans and 13 in mice (Pasare and Medzhitov, 2005), while there are up to 23 TLRs in fish (Palti, 2011).

Many immune cells and 'barrier' cells express TLRs such as DCs, MØs, T and B cells, polymorphonuclear cells, microglial cells, endothelium, and epithelium (Bsibsi *et al.*, 2002, Armstrong *et al.*, 2004, Miller *et al.*, 2005). These cells recognise microbial PAMPs via TLRs and bind them, leading to the release of reactive oxygen species (ROS) metabolites, reactive nitrogen species (RNS) metabolites, pro-inflammatory cytokines like IL-1, IL-6, IL-8, and TNF-a, as well as co-stimulatory molecules, which include CD28, CD40, CD80 (B7.1) and CD86 (B7.2), which can activate T cells, B cells and DCs, thus triggering adaptive immunity (Werling and Jungi, 2003).

#### 1.5.1 TLR structure and function

Figure 1.1 shows the structure typical of TLRs. They are type I transmembrane receptors which are glycoproteins consisting of 3 portions; first, a unique conserved intracytoplasmic Toll/IL-1 receptor (TIR) as a Cterminal signalling domain to all TLRs; secondly, common а transmembrane portion as a single helical connecting part; and third, an extracellular N-terminal leucine-rich repeat (LRR) domain that differs morphologically amongst TLRs (Werling et al., 2009, Reuven et al., 2014). The intracellular (TIR) domain is responsible for cascades of events leading toward triggering the signalling system which is regulated by one of 5 important adapter proteins: myeloid differentiation primary response gene 88 (MyD88), MyD88-adapter like (MAL) named as Toll-interleukin 1 receptor (TIR) domain containing adapter protein (TIRAP) as well; TIRdomain-containing-adapter-inducing interferon- $\beta$  (TRIF); TRIF-related adaptor molecule (TRAM) and Sterile-alpha and Armadillo-motif-containing protein (SARM) (O'Neill and Bowie, 2007). TLRs can work individually or as heterodimers as in TLR1/2, TLR2/6 interaction by binding to each other to exert potential immunological effect where these proteins traffic to induce activation of transcription factors (Triantafilou *et al.*, 2006).

The transmembrane domains of TLRs play an important role in recognition of RNA and/or DNA of PAMPs (mostly viruses) by binding to them in the endoplasmic reticulum via UNC93B (a transmembrane protein important for signalling of human and murine TLR3, 7 and 9) (Brinkmann *et al.*, 2007, Kim *et al.*, 2008).



Figure 1.1: Schematic diagram showing a typical TLR structure. ECD= extracellular domain, LRR= leucine rich repeat, TIR= Toll/IL-1 receptor.

The glycoprotein part consists of LRRs of 20–30 amino acids which is the immunologically active component of TLRs responsible for capturing PAMPs. The LRR proteins give TLRs a horseshoe shape as a curved solenoid structure; therefore, its outline resembles two surface coils (convex and concave faces) forming an "m-shape" ending with two loops (Bella *et al.*, 2008, Botos *et al.*, 2011). These LRR proteins can be further divided into 7 subfamilies in accordance to their sequence and structural basis, six of them have been identified in vertebrates and one in plants (Matsushima *et al.*, 2007). The differences in these LRR proteins result in capturing varieties of pathogens by their PAMPs as listed in table (1.5).

Functionally, TLRs have multiple signalling functions, i.e. the major specific TLRs for sensing viral antigens are TLR3, TLR7, TLR8 and TLR9 which are responsible for recognition of dsRNA viruses, ssRNA viruses and DNA

viruses respectively (Alexopoulou *et al.*, 2001, Takeshita *et al.*, 2001, Heil *et al.*, 2004). Recently, expression of TLR10 has been reported in human monocytes *in vitro* after infecting them with Influenza virus resulting in proinflammatory cytokines secretion (Lee *et al.*, 2014). TLR2/6 (heterodimer) on host cells can recognise PAMPs of *Saccharomyces cerevisiae*; and TLR2 and TLR4 recognise PAMPs in *Cryptococcus neoformans* and *Candida albicans* (Akira *et al.*, 2006). Moreover, some large-size parasites (helminths) have PAMPs that could also be detected by TLRs (Venugopal *et al.*, 2009) e.g. *Filaria* and *Schistosoma* species could be recognised by TLR2 and TLR4. Added to that, small size parasites (protozoa) can be recognised by TLR2, TLR4 and TLR9 for example in the response to malaria (*Plasmodium falciparum*); as well as specific findings illustrated in TLR11 and TLR12 which play an important role in detection of profilin antigen (actin-binding protein) expressed by *Toxoplasma gondii* in mice (Shi *et al.*, 2012, Koblansky *et al.*, 2013).

On the other hand, a wide spectrum of TLRs have been described in terms of bacterial PAMPs recognition, which could be subdivided into groups of TLRs, e.g. TLR1/2 heterodimer, TLR2 and TLR2/6 on mammalian cells can sense for bacterial lipoproteins; TLR2 peptidoglycan and lipoteichoic acid; TLR4 lipopolysaccharide and TLR5 flagellated bacteria (Philpott and Girardin, 2004, Akira *et al.*, 2006). Moreover, protozoa can be sensed by certain types of TLRs such as *Toxoplasma gondii* which is recognised by TLR11 and TLR12 (Shi *et al.*, 2012, Koblansky *et al.*, 2013); *Trypanosoma cruzi*, *T. brucei*, *Leishmania major* and *Plasmodium falciparum* (Malaria) by TLR2 and TLR4 (Gazzinelli and Denkers, 2006).

Table 1.5: Ligands	of different TL	Rs and their targets.
--------------------	-----------------	-----------------------

TLRs	Dimerization	Cells expressing	Adaptor	Target recognition and ligands	References
		TLRs	molecule		
TLR1	Heterodimer	MØs and PMNs	MyD88/	Triacyl lipopeptides and peptidoglycans in	(Wyllie <i>et al</i> ., 2000,
	with TLR2		TIRAP	majority of bacteria and Mycobacteria;	Takeuchi <i>et al</i> .,
				other bacterial soluble components.	2002)
TLR2	Heterodimer	Microglia, PMNs,	MyD88/	Lipoprotein in different bacteria and other	(Aliprantis <i>et al</i> .,
	with TLR1 or	MØs, monocytes, B	TIRAP	pathogens; Peptidoglycan (PGN) and/or	1999, Means <i>et al</i> .,
	with TLR6	cells, CD4+ T cells		Lipoteichoic acid (LPA) of many Gram +ve	1999, Asea <i>et al</i> .,
		including Tregs, DCs		bacteria; Lipoarabinomannan in	2002, Bieback <i>et al</i> .,
		and respiratory		Mycobacteria; Zymosan in fungi; Heat	2002, Compton <i>et al</i> .,
		epithelia		shock protein 70 (Hsp70) in host cells;	2003, Bellocchio <i>et</i>
				Hemagglutinin in Paramyxovirus and	<i>al</i> ., 2004)
				Cytomegalovirus (CMV).	

	1				
TLR3	Homodimer	Intracellular in T	TRIF	Double stranded RNA (dsRNA) viruses	(Alexopoulou et al.,
		cells, DCs and			2001)
		placental epithelia			
TLR4	Homodimer	Placental epithelia,	MyD88/	Lipopolysaccharide in many Gram -ve	(Chow <i>et al</i> ., 1999,
		monocytes	TIRAP,	bacteria; Fusion protein in Respiratory	Ohashi <i>et al</i> ., 2000,
			SARM	syncytial virus (RSV); Hsp60, Hsp70 and	Smiley <i>et al</i> ., 2001,
				Fibrinogen in host cells.	Asea <i>et al</i> ., 2002)
TLR5	Homodimer	Monocytes, NKT	MyD88	Flagellin in motile bacteria	(Hayashi <i>et al</i> ., 2001)
		cells, intestinal			
		epithelia and MØs			
TLR6	Heterodimer	Monocytes,	MyD88	Diacyl lipopeptides in Mycoplasma	(Takeuchi <i>et al</i> .,
	with TLR2	spleenocytes and			2001)
		thymocytes			
			1		

TLR7 &	Homodimer	Intracellular in DCs,	MyD88	Single stranded RNA (ssRNA) and Single	(Heil <i>et al</i> ., 2004)
TLR8		T cells MØs and		stranded DNA viruses	
		spleenocytes			
TLR9	Homodimer	Intracellular in DCs,	MyD88	DNA viruses; Unmethylated CpG DNA in	(Takeshita <i>et al</i> .,
		T and B cells, NK		bacteria, viruses, fungi; hemozoin in	2001, Coban <i>et al</i> .,
		cells and monocytes		Malaria	2005)
TLR10	Not	Spleenocytes, lymph	MyD88	Influenza viruses (Orthomyxoviruses) /	(Lee <i>et al</i> ., 2014)
	confirmed	nodes, thymocytes,		recently discovered	
		and respiratory			
		epithelia			
TLR11	No data	DCs, MØs, epithelial	MyD88	Blockage of Salmonella adhesion in	(Shi <i>et al</i> ., 2012,
		cells of liver, gut		murine gut; profilin in Toxoplasma gondii	Koblansky <i>et al</i> .,
		kidney, and bladder			2013)
TLR12	No data	DCs	MyD88	Profilin in Toxoplasma gondii	(Koblansky <i>et al</i> .,
					2013)

				_	
TLR13	No data	DCs and MØs	MvD88	Receptor for RNA Bacteria	(Hidmark <i>et al.,</i>
					(
					2012)
					2012)
	No data	Unknown	Mainly	Fish TI Rs	(Palti 2011 Pietretti
	NO Gata	OTIKITOWIT	Hanny		(1 ald, 2011, 1 lettett
тірээ			MUDOO		and Wiggortian
TLRZJ			MyDoo		and wiegerijes,
					2014)
					2014)
				1	

#### 1.5.2 T and B lymphocytes expressing TLRs

Although previous studies showed the prominent role of TLRs in initiation of innate immune responses, recent studies have described that TLRs are expressed on adaptive immune cells (T and B cells) (Reynolds and Dong, 2013, Buchta and Bishop, 2014a and b). Both B cells (naïve and activated) and memory B cells express TLR1, 4, 6, 7, 9 and 10 in humans (Hornung *et al.*, 2002, Mita *et al.*, 2002, Bernasconi *et al.*, 2003). In addition, TLR2 can be expressed on human B cells in response to protein A (a virulence factor in *Staphylococcus aureus*) resulting in B cell receptor (BCR) activation (Bekeredjian-Ding *et al.*, 2007). The BCR consists of immunoglobulin molecules located on the outer surface of B cells. Further investigations showed that human plasma cells express TLR1-9 which enhance antibody production leading to a more powerful humoral immune response (Dorner *et al.*, 2009).

T cells also have TLR expression in mice and humans. CD4+ T cells express TLRs 2,3,4 and 5 and TLR7, 8, 9 (Gelman *et al.*, 2004, Komai-Koma *et al.*, 2004, Caron *et al.*, 2005, Zanin-Zhorov *et al.*, 2007); while Th1, Th2 and Th17 express TLR2 and TLR4 (Matsuguchi *et al.*, 2000, Imanishi *et al.*, 2007, Reynolds *et al.*, 2010, Reynolds *et al.*, 2012). CD8+ T cells express TLR2, 3 and 9 (Komai-Koma *et al.*, 2004, Babu *et al.*, 2006, Tabiasco *et al.*, 2006). Moreover, Gamma delta T cells have the capability to express TLR1, 2 and 3 and TLR6, 7 and 8 (Pietschmann *et al.*, 2009), whereas natural killer T cells (NKT cells) could express TLR2, 3, 4 and 5, TLR7 and TLR9 (Shimizu *et al.*, 2002, Saikh *et al.*, 2003, Kulkarni *et al.*, 2012) and finally Tregs are through to express TLRS 1-8 (Sutmuller *et al.*, 2006).

#### 1.5.3 TLR signalling pathways

#### **1.5.3.1 Extracellular TLR signalling pathway**

Extracellular TLR homodimers include TLR1, TLR2, TLR4, TLR5 and TLR6, plus heterodimers (TLR1/2, TLR2/6). They share approximately the same signalling pathway. As an example of extracellular TLRs, the TLR5 signalling pathway (see figure 1.2) shows the classical signalling route. The recognition of flagellin by TLR5 leads to the recruitment and activation of MyD88, which leads to activation of transcription factor NF $\kappa$ B resulting in secretion of pro-inflammatory cytokines (Barton and Medzhitov, 2003). MyD88 is an adaptor protein composed of a TIR domain that binds to the N-terminal protein in interleukin-1 receptor-associated kinase (IRAK) and tumour necrosis factor (TNF) receptor associated factor-6 (TRAF6) (Medzhitov et al., 1998). This involves the kinases IRAK1 then IRAK4. responsible for indirect activation of TRAF6 via a phosphorylation process (Kawai and Akira, 2007). The activated TRAF6 attracts TGF-β-activated kinase-1 (TAK1), which is an important enzyme for upregulation of I-kappa B kinase complex (IKK- $\beta$ ) and mitogen-activated protein kinase (MAPK) kinase-6 (MKK6) This series of events activates NFkB transcription factor (Wang et al., 2001, Adhikari et al., 2007). This in turn induces several events, most notably pro-inflammatory cytokine production that includes: IL-1β, IL-6, IL-8, and TNFa (figure 1.2) (Brasier, 2010, Tornatore et al., 2012). Also, Meade et al. (2012) added that TLR5 stimulated cells lead to CXCL-8 production by monocytes, MØs, endothelial and epithelial cells, T cells, and neutrophils, which enhance attraction of phagocytic cells and maintain the activity of mononuclear cells and angiogenesis.



Figure 1.2: TLR5 signalling pathway as an example of extracellular TLR signalling pathway.

#### **1.5.3.2 Intracellular TLRs signalling pathways**

Intracellular TLRs include TLR3, 7, 8 and 9 (mostly antiviral sensors). These intracellular TLRs recognise the nucleic acids of viruses in a mechanism that terminates with type 1 Interferon (IFNα and β) production. This happens through interferon-regulatory factors (IRF3, IRF5, and IRF7 known as effective transcription factors), these are important for T lymphocyte activation (mainly cytotoxic T cells) to trigger antiviral immune responses (Moynagh, 2005, Kawai and Akira, 2007).

However, Gay *et al.* (2014) proposed that these intracellular TLRs possibly could share the same pathway with the extracellular TLRs through activation of the MyD88-dependent pathway (see above section 1.5.3.1) to end up with induction of pro-inflammatory cytokines. This is not necessarily only NFκB transcription factor activation, but some other transcription factors might be involved such as cAMP-responsive element-binding protein (CREB) and activator protein-1 (AP-1) in case of intracellular TLR7, 8, and 9 signalling pathways (Hemmi *et al.*, 2002, Latz *et al.*, 2007, Tanji *et al.*, 2013).

In contrast, TLR3 uniquely reveals a different signalling route from all other TLRs after engagement with dsRNA viruses. Experimental infection with orthomyxoviruses (that contain the causative agents of influenza, also called influenza viruses) *ex vivo* resulted in activation of TIR domain-containing adaptor protein inducing IFN $\beta$  (TRIF), which (TRIF-driven route) leads to production of the TRAF3 effector proteins necessary for binding of IKK epsilon (IKK $\epsilon$ ) protein with either IRAK1 or TBK1 to produce IRF3 or IRF7 transcription factors respectively that penetrate the nuclear

membrane to stimulate transcription for type I IFN secretion (Liu *et al.*, 2008, Luo *et al.*, 2012, Teijaro *et al.*, 2014).

# 1.6 Bovine TLRs - in depth highlights and

# specifications

After discovery of the PRR molecules (including TLRs) in humans and some laboratory animals (mice, rats, and guinea pigs), attempts commenced to study the presence and function of these sensory molecules in domestic animals and ruminants (Turin and Riva, 2008, Smith *et al.*, 2012). Initial attempts (Menzies and Ingham, 2006) to investigate expression of TLR1-10 by real time PCR (RT-qPCR) in ruminants (a focus on cattle and sheep), revealed the presence of 10 TLRs (TLR1 to TLR10) that shared 83-90% amino acid similarity to human TLRs 1-10 [confirmed when McGuire *et al.* (2006) mapped them]. Menzies and Ingham also observed that all bovine TLR1-10 genes except TLR6 were expressed in the skin, Peyer's patches, and mesenteric lymph nodes (gut associated lymphoid organs) of cattle but at low levels, except for bovine TLR2, and 7 which were strongly expressed in the skin.

The genome of *Bos taurus* has also been mapped revealing 10 TLRs (TLR1-10) genes (McGuire *et al.*, 2006, Seabury *et al.*, 2010). These bovine TLR sequences are currently available on the website of the National Centre for Biotechnology Information (NCBI) (see table 1.6).

# Table 1.6: Gene accession numbers of up to date TLR1-10 genes in cattle (species *Bos taurus*). Cited from National Centre for Biotechnology Information (NCBI). Accessed 30-08-2017.

Gene	mRNA size /	Location	Accession No. /
name	base pair (bp)	(chromosome No.)	GenBank (NCBI)
TLR1	2319 bp	6	NM_001046504
TLR2	3513 bp	17	NM_174197
TLR3	3025 bp	27	NM_001008664
TLR4	3739 bp	8	NM_174198
TLR5	2800 bp	16	NM_001040501
TLR6	3095 bp	6	NM_001001159
TLR7	3177 bp	X	NM_001033761
TLR8	3075 bp	X	NM_001033937
TLR9 3265 bp		22	NM_183081
TLR10	3163 bp	6	NM_001076918

## 1.6.1 Bovine TLR1

This is an extracellular protein expressed in response to exposure of host cells to the lipid compartments of bacteria (such as lipoteichoic acid, a lipoprotein in most Gram-positive bacteria). Farhat *et al.* (2010) sequenced the full length bovine TLR1 gene and showed the protein bound synthetic diacyl and triacyl lipoproteins. Later on, Ruiz-Larranaga *et al.* (2011) described the genetic diversity of this gene by describing 24 SNPS in the TLR1 of Holstein-Friesian cattle infected by *Mycobacterium avium paratuberculosis.* Further studies by Russell *et al.* (2012) pointed out the importance of bovine TLR1 and its variants, synonymous and non-

synonymous SNPs (sSNPs and nsSNPs), in detection of bacterial lipoproteins during clinical mastitis.

Upon stimulation with bacterial lipoproteins, studies in cattle showed expression of TLR1, 2, and 6 RNA in the epithelium of endometrial and stromal cells but functionally, only TLR1 and TLR2 resulted in induction of an inflammatory reaction represented by detection of high levels of IL-6 and IL-8 secreted cytokines (Turner *et al.*, 2014).

#### 1.6.2 Bovine TLR2

The first report to identify bovine TLR2 was in 2003 when White *et al*. (2003) studied the sequence similarity in this gene between human and bovine TLR2. In cattle, studies showed that bovine TLR2 and TLR4 were detected in mammary epithelial cells in *Staphylococcal* mastitis cases (Goldammer *et al.*, 2004).

In the intestine, TLR2 and TLR4 were detected in the epithelial layer infected with *Cryptosporidium parvum* (a protozoan parasite that causes diarrheal enteritis in calves) through secretion of CXCL-8 (Yang *et al.*, 2015). One of the most important respiratory diseases in livestock is *Mycoplasma*. Bovine TLR2 is thought to play an important role in detection of *Mycoplasma bovis* lipid associated particles through the MyD88-driven pathway resulting in pro-inflammatory IL-1 $\beta$  secretion (Wang *et al.*, 2016). TLR2 has been detected in the cattle reproductive organs (mainly corpus luteum in the ovaries of Chinese Holstein cattle) which may play a role (along with TLR4) in early sensing of virulent Gram-negative bacteria by triggering inflammation (Lüttgenau *et al.*, 2016). More recent observations (Zhao *et al.*, 2017) revealed that some TLR2 SNPs might be associated

with recognition of *Mycobacterium* infected cattle (Chinese Holstein species).

#### 1.6.3 Bovine TLR3

Bovine intracellular TLR3 was first confirmed by McGuire *et al.* (2006). TLR3 in cattle contains many SNPs and presents with numerous variants (Cargill and Womack, 2007, Zimin *et al.*, 2009). A new cell line model for studying the immune functions of TLR3 (known as dsRNA sensor) was developed by Chiba *et al.*, (2012) who used cells derived from bovine intestinal epithelial (BIE cells) on which TLR3 was highly expressed, which upon stimulation with polyI:C (TLR3 agonist) resulted in detectable levels of pro-inflammatory cytokines.

Rensetti *et al.* (2016) stated that TLR3 is involved in recognition of BoHV-1 and BoHV-5 in the infected neurons and olfactory tract. This notion has been recently further supported by Oliveira *et al.*, (2017) who found a strong correlation between TLR3 signalling in the CNS with cows infected by BoHV-5 viruses.

#### 1.6.4 Bovine TLR4

This particular receptor (TLR4) is a sensory receptor for LPS in most Gramnegative bacteria. In cattle, TLR4 structure and function does not differ much from those in other mammals. Bovine TLR4 often functions along with TLR2 to recognise infection, for example *C. parvum* in the intestine (see section 1.6.2) and trigger inflammation (Yang *et al.*, 2015). Both TLR4 and TLR2 undergo biological changes for up to 7 days after birth. Their expression dropped in the endometrium from cows diagnosed with dystocia due to retained placenta compared to healthy cows (Martins *et al.*, 2016). Shimizu *et al.*, (2017) identified numerous numbers of TLR4 SNPs in exon 3 of the gene in Holstein cows which might have functional consequences. TLR4 may be involved in susceptibility / resistance of Holstein cows to digital dermatitis infection (El-Shafaey *et al.*, 2017). Moreover, properties of milk production in Vrindavani cattle (Indian breed) were thought to be linked to TLR4 gene expression and SNP variants (Mishra *et al.*, 2016). In contrast, *Sahiwal* and *Hariana* cattle (Indian breeds) demonstrated marked down regulation of TLR4 gene expression in PMNs during winter and summer seasons (Swain *et al.*, 2017).

Interestingly, bovine TLR4 can be negatively affected (downregulated) by exposure to certain antibiotics such as sodium houttuyfonate (SH). This compound is known to have anti-inflammatory properties in treatment of bovine mastitis, which also suppresses TLR4 expression in the mammary gland epithelium in response to LPS stimulation by Gram-negative bacteria (mainly *E. coli*). The suppression could be attributed to blockage of transcription factor NF $\kappa$ B activation (Wang *et al.*, 2017).

#### 1.6.5 Bovine TLR5

Flagellated motile bacteria can be sensed by TLR5 in mammals. This was investigated by Tahoun *et al.*, (2015) who identified structural and functional similarity between human TLR5 (hTLR5) and bovine TLR5 (boTLR5). They transfected human embryonic kidney (HEK293) and embryonic bovine lung cell lines with boTLR5 and stimulated them with a flagellin derived from *E. coli* strain O157:H7. Both cell lines responded positively by production of CXCL-8 in the supernatant.

However, previous studies showed that bovine TLR5 was less responsive to flagellins than its human counterpart (Metcalfe *et al.*, 2014; Lankester et al., 2016). In a flagellin TLR5 adjuvant study in Zebu cattle, the effect of flagellin was not effective but interestingly was associated with antibody immunosuppression (Lankester *et al.*, 2016). Based on the above information, boTLR5 may play an additional role or even alternative to that known in humans.

#### **1.6.6 Bovine TLR6**

Bovine TLR6 was discovered in the same region of about 50 kB in chromosome 6 that contains TLR6, TLR1, and TLR10 (Opsal *et al.*, 2006). Peptidoglycan binding and SNP variants have described (Seabury and Womack, 2008). In addition, bovine TLR6, 4, and 2 were screened for polymorphisms in European cattle breeds and 16 SNPs were identified (Mariotti *et al.*, 2009)

As described in section 1.6.1, bovine TLR6 was expressed (along with either TLR1 or TLR2) in the endometrial epithelium but reported as non functional (Turner *et al.*, 2014). However, Song *et al.* (2014) identified 4 SNPs of bovine TLR6 that played an important role against TB in Chinese Holstein cows and lack of them might expand the risk factor of TB. Finally, heterodimerisation of TLR6 with TLR2 was formerly described in humans and some animals (Triantafilou *et al.*, 2006, Ren *et al.*, 2016), but this is still untested in cattle.

#### 1.6.7 Bovine TLR7 and TLR8

Both TLR7 and TLR8 are known in mammals as antiviral intracellular PRRs which recognise ssRNA viruses and to a lesser extent some DNA viruses and this leads to secretion of type I interferons (IFNa and  $\beta$ ) through activation of IRF transcription factor (Hart et al., 2005, Stary et al., 2007, Wei et al., 2009, van Haren et al., 2016, Gidon et al., 2017). Bovine TLR7 and TLR8 (TLR7/8 hereafter) were mapped and confirmed by RT-qPCR (McGuire et al., 2006, Menzies and Ingham, 2006). More than 100 SNPs have been documented in TLR3, 7, and 8 by (Cargill and Womack, 2007) who attributed this to possible multiple functions of these TLRs. Also, Indian crossbred cows expressed TLR7 on PBMCs at variable rates according to geographical distribution of the livestock (Singh *et al.*, 2014). Bovine TLR7/8 have been studied by Rensetti et al. (2016) who confirmed the expression of TLR7/8 in the central nervous system of calves infected with bovine alpha herpes virus (BoHV-1 and BoHV-5 strains). These TLRs participated (although TLR7 was much more prominently expressed) in initiation of inflammation as detected in the histopathological sections as part of their sensory role upon recognition of herpes viruses by comparison with calves infected with mock viral suspension.

Interestingly, Schaut *et al.* (2016) stated that monocyte-derived MΦs infected with bovine viral diarrheal virus-2 (BVDV-2) expressed higher levels of TLR7 determined by RT-qPCR and prompted large quantities of secreted IL-6. With regards to bovine TLR8 (boTLR8) protein, Zhu *et al.* (2009) noticed that it is located inside cells infected with BHV-1.

#### 1.6.8 Bovine TLR9

TLR9 recognises non-methylated CpG motifs existing in the DNA of various viruses and bacteria. Bovine TLR9 (boTLR9) shares many of the features of other mammalian TLR9s (Griebel *et al.*, 2005).

Schneberger *et al.* (2011) developed a mouse anti-bovine TLR9 antibody which detected TLR-9 by western blot, immunohistochemistry and electron microscopy for its expression in the bovine respiratory system (airway epithelium, endothelium, lung tissue including cellular compartments such as alveolar MΦs as well as intravascular monocytes and MΦs).

Molecular studies in cattle highlighted SNP variants of TLR-9, some of which were thought to participate in host susceptibility to TB infection in Chinese Holstein cows (Sun *et al.*, 2012). In addition, the structure of three forms of *Bos taurus* TLR9 was further functionally identified by Ohto *et al.*, (2015) as a sensory molecule for unmethylated CpG DNA viruses. All three forms of boTLR9 bound to CpG unmethylated DNA and elicited inflammation.

Parameswaran *et al.*, (2014) used unmethylated CpG oligonucleotides (CpGODN) as an adjuvant to target TLR9 in immunisation studies against malignant catarrhal fever. They found that the inclusion of CpGODN did not have an enhancing effect on protective immunity over Emulsigen adjuvant when included together and little effect on its own.

#### 1.6.9 Bovine TLR10

Less is known about TLR10 and its ligands in mammals. boTLR10 was first identified in the same region of the genome as TLR1 and TLR6 and is about 50 kB in chromosome 6 (Opsal *et al.*, 2006). The biological topography of

boTLR10 was drawn by Werling *et al.* (2006) who noticed diversified presence of boTLR10 mRNA transcripts in some APC cells detected by RTqPCR in monocytes, monocyte-derived MΦs, monocyte- derived DCs, and afferent lymph DCs (both CD172a<sup>+</sup> and CD172a<sup>-</sup>), whereas alveolar MΦs, DCs derived from bone marrow, and B lymphocytes consistently expressed boTLR10.

This bovine PRR molecule, along with the rest of the bovine TLRs, has been sequenced (McGuire *et al*., 2006, Seabury *et al*., 2007, Zimin *et al*., 2009, Seabury *et al*., 2010).

# 1.7 TLR agonists

In mammals, conserved regions of microbial organisms (PAMPs) induce expression of cognate TLRs on immune cells. This stimulation results in an inflammatory immune response that results in pro-inflammatory cytokine secretion as described in previous sections. These microbial molecules (PAMPs) have been synthesised and are available from companies for research (e.g. InvivoGen and AdipoGen), where 10 ligands for human TLR1-9 are made by InvivoGen and 16 ligands by AdipoGen (see links):

- 1. (http://www.invivogen.com/human-tlr1-9-agonist-kit)
- (<u>http://www.adipogen.com/media/Catalogs/PDFs/Innate\_Immunit</u> y\_Flyer\_2015\_Adipogen.pdf).

These companies manufacture human TLR1-9 and TLR11/12 ligands for academic and commercial purposes. To date there are no available TLR10 agonists with the exception that TLR2/6 agonist might also be considered as a putative TLR10 agonist (Tocris-Bioscience, 2017).

## 1.8 TLR agonists /ligands as new generation

## adjuvants

The expression of TLRs on immune cells such as DCs, MØs, T cells and PMNs can control the direction and magnitude of immune responses affecting innate and adaptive immunity. Thus, TLR ligands (PAMPS etc) are likely to be useful new generation adjuvants that can affect the magnitude and type of immune response to vaccination (Hayashi *et al.*, 2001, Kaisho and Akira, 2002, Hedayat *et al.*, 2011).

Traditional vaccines, using well tested adjuvants such as alum and Quil-A amongst others still provide efficacies against existing pathogens and some newly emerging ones. However, the ability to direct the type of immune response to a protective one involving new adjuvants is particularly timely and important (Tomljenovic and Shaw, 2011, Tomljenovic and Shaw, 2012). In addition, the overuse of antibiotics is leading to antibiotic resistance in bacteria and new vaccines are very important to develop (Davies and Davies, 2010). For all those purposes, the use of TLR agonists as a new promising generation of adjuvants will be useful to deliver vaccines against pathogens in both human and veterinary medicine. In the case of TLR5, the starting point of this interest was in 2002 when Liaudet et al. (2002) showed that the flagellin-TLR5 pathway induced inflammation and that this could be utilised to help cure endotoxic shock, purulent infection attributed by pyogenic bacteria, pulmonary affections and airway epithelial disturbances. This idea was exploited by Blohmke et al. (2008) who stated that TLR5 agonists could be used effectively against harmful bacteria such as Burkholderia cepacia and Pseudomonas aeruginosa as a model for cystic fibrosis patients who developed severe chronic pulmonary

signs. They recorded increased pro-inflammatory mediators as well as higher levels of antibody in those patients compared with controls suggesting that TLR5 agonists could be used as part of a treatment to protect against cystic fibrosis.

A patent was lodged by Rhee *et al.* (2012) who created TLR5 agonist mutants by genetic manipulation of flagellins to produce a powerful TLR5flagellin signalling cascade. Furthermore, Minton (2014) described the importance of adding TLR5 agonists to influenza vaccine in mice. They suggested that gut microbes stimulate the murine host to produce an antibody response to influenza vaccine in the presence of flagellin which leads to higher TLR5 expression on B cells resulting in induction of cytokine release. Finally, a patent was successfully obtained for a novel discovery of an immunomodulatory flagellin peptide that possesses TLR5 binding properties (Aderem *et al.*, 2014).

## 1.9 Hypotheses, objectives, and questions

- 1. I hypothesise that cattle T cells express TLRs, and they are differentially expressed between the T cell subsets CD4+, CD8+,  $\gamma\delta$  TCR T cells, and in addition that they are functional.
- Function of T cell TLRs will be measured by stimulation with TLR ligands and by comparison with Con-A (T cell mitogen) or anti-CD3 (a known stimulus for T cells) as controls.

It is possible that engagement of TLRs on DC/ monocyte antigenpresenting cells will have a different function than those on effector T cells (CD4 T cells,  $\gamma\delta$  T cells or CD8 T cells) or Tregs and understanding these interactions will inform the choice of vaccine adjuvant for protective immune responses to pathogens of veterinary importance. On this basis, we formed and developed our research objectives /questions:

1) To what extent do bovine PBMCs, T cells and/or their subsets express TLRs?

We proposed in the first instance to undertake a broad screen of bovine PBMCs and purified major T cell subsets (CD4, CD8 and  $\gamma\delta$  T cells) for reactivity to TLR ligands.

**Chapter Two** 

**General Materials & Methods** 

# 2.1 Animals

Cattle 18-24 months of age (table 2.7) were used. Two sources were used for the blood of these animals: Sera Laboratories, West Sussex, UK and Elliot Abattoir, Chesterfield, UK. Animals were kept off food for 24 hours and off water for 12 hours before slaughter. All animals were overtly clinically free of disease. Ethical approval was obtained from the SVMS Clinical Ethical Review Committee. All material (blood) was taken immediately post-mortem.

Table 2.7: Animals used for blood collection. Age, gender, and breed wi	th
the source are given.	

Lot number (Seralab)/ blood animal	Animals description	Source
BOV4098	18 months female Limousine breed calf	Sera Laboratories, West Sussex, UK
Bov A (EA)*	22 months female Limousine breed calf	R B Elliott & Sons Abattoir, Chesterfield, UK
Bov B (EA)	18 months female Limousine breed calf	R B Elliott & Sons Abattoir, Chesterfield, UK
Bov 1 (EA)	24 months male Limousine breed calf	R B Elliott & Sons Abattoir, Chesterfield, UK
Bov 2 (EA)	18 months male Friesian calf	R B Elliott & Sons Abattoir, Chesterfield, UK
Bov 3 (EA)	20 months male Limousine calf	R B Elliott & Sons Abattoir, Chesterfield, UK
Bov 4 (EA)	20 months male Limousine calf	R B Elliott & Sons Abattoir, Chesterfield, UK
Bov 5 (EA)	20 months female Friesian calf	R B Elliott & Sons Abattoir, Chesterfield, UK
Bov 6 (EA)	20 months male Limousine calf	R B Elliott & Sons Abattoir, Chesterfield, UK
Bov 7 (EA)	20 months male Limousine calf	R B Elliott & Sons Abattoir, Chesterfield, UK
Bov 8 (EA)	20 months male Limousine calf	R B Elliott & Sons Abattoir, Chesterfield, UK

\* EA = Elliot Abattoir, Chesterfield, UK
#### 2.2 Blood and tissue collection

#### 2.2.1 Rabbit MLN cells

These cells were already prepared by a previous Postdoctoral scientist (Nevi Parameswaran) from 6 New Zealand white healthy rabbits' (of either sex), 2-4 months of age (Charles River, UK). Mesenteric lymph nodes (MLN) were harvested and chopped into small slices, prior to single cell preparation in sterile bags by using a stomacher machine (10 seconds). Cells and detritus were filtered with PBS through two layers of Whatman sterile lens tissue then adjusted to 1X10<sup>7</sup> cells/ml of cryopreservation medium (!0% v/v DMSO in DMEM containing 50% FCS) and stored in cryotubes in liquid nitrogen. These were harvested by thawing in a water bath prior to use.

#### **2.2.2 Bovine blood**

Sera Laboratories West Sussex, UK: whole bovine blood (for PBMC isolation), taken from healthy animals (vessel held under the throat during exsanguination) by vacutainer (containing lithium heparin, H0878, Sigma-Aldrich, Missouri, USA) and shipped immediately after collection at ambient temperature.

R B Elliott & Sons Abattoir (Chesterfield, Derbyshire, UK) was the other source of blood, post mortem from slaughtered cattle. In this instance, 20 mg Lithium heparin as an anticoagulant was dissolved in sterile 50 ml PBS and transferred into a sterile 1 L screw-capped glass bottle inside a class

II microbiology safety cabinet. The bottles were transported in a special tightly-sealed clinical transfer box.

#### 2.3 Isolation of PBMCs from bovine blood

Blood was processed inside the class II microbiology safety cabinet under aseptic conditions. The blood was diluted 1:1 with sterile Dulbecco's Phosphate Buffered Saline (D8537 - Sigma-Aldrich, Haverhill, UK), which is abbreviated as DPBS hereafter, then distributed into 50 ml falcon tubes. Buffy coat was collected by Pasteur pipettes and diluted again with PBS (1:1), then centrifuged at 300g for 30 minutes at 22°C, brakes off. After that, the cell suspension was layered on Histopaque-1077 (10771 Sigma-Aldrich, Haverhill, UK). This was achieved by adding 30 ml cell suspension to 15 ml Histopaque solution then centrifuged at 300g for 30 min at 22  $^{\circ}$ C, brakes off. The mononuclear cell layer (PBMCs) was aspirated and collected by Pasteur pipette into 50 ml falcon tubes (see figure 2.3), diluted with DPBS and centrifuged at 300g for 30 minutes at 22°C, brakes off. Cells were washed with DPBS twice and centrifuged at 300g for 10 minutes at 22 °C, brakes on. Cell pellets were covered with 5ml Erythrocyte Lysis Buffer (ELB\*), incubated at 37°C for 5 minutes to lyse all the remaining RBCs and centrifuged at 300g for 10 minutes at 22 °C, brakes on (repeated twice until seeing a clear translucent solution/ disappearance of the red colour). To remove the ELB solution, the cell suspension was washed again twice with RPMI 1640 complete medium. RPMI 1640 complete medium (see appendix 8.1.1) consists of the following: 500 ml RPMI 1640 medium (11544526 – Fisher Scientific, Loughborough, UK) plus 10% FCS (10073772 – Fisher Scientific, Loughborough, UK) plus 1% (10,000 U/ml)

penicillin and 1 mg/mL streptomycin (15140122 – Fisher Scientific, Loughborough, UK) plus 1% L-glutamine (G7513 - SIGMA-ALDRICH, Dorset, UK) and finally 1 ml Gibco Amphotericin B (previously named as Fungizone®) as an antimycotic (15290-018 – Fisher Scientific, Loughborough, UK) at 0.25 µg/ml as a final working concentration.

Cells were counted by haemocytometer then mixed with the freeze mix (see appendix 8.1.2) solution: 90% FCS + 10% DMSO (Dimethyl sulfoxide). The final cell count was adjusted to be  $1 \times 10^7$  cells/ml and 1ml poured into cryotubes, which were stored in a -80 °C freezer overnight, and finally transferred to liquid nitrogen next day to store until use.



Figure 2.3: Layering buffy coat cell suspension on Histopaque 1077.

#### 2.4 Reagents

#### 2.4.1 IMDM Complete Medium

It was prepared as follows; Iscove's Modified Dulbecco's Medium (IMDM) (10474202 - Thermofisher Scientific, Rugby, UK). 1% 100U/mL penicillin and 50 (µg/mL) streptomycin (penstrep) made up from Pen/Strep stock (10,000 IU/ml penicillin and 10 mg/mL streptomycin at final working concentration 100 IU/mL penicillin and 100 (µg/mL) streptomycin (15140122 - Thermofisher Scientific, Loughborough, UK) plus 10% fetal calf serum (FCS) (10082139 - Thermofisher Scientific, Loughborough, UK) were added.

#### 2.4.2 Reagents and cytokines

Concanavalin A (Con A) (C5275), phytohaemagglutinin (PHA) (L1668) and phorbol 12-myristate 13-acetate (PMA) (P1585) were all purchased from Sigma-Aldrich, Haverhill, UK. Transforming Growth Factor β1 (TGF-β1) (100-21 – Peprotech, USA), Interleukin 2 (IL-2) (202-IL-010 - R&D Systems, Abingdon, UK) and CellTrace<sup>™</sup> Violet Cell Proliferation Kit (C34557 - Thermofisher Scientific, Rugby, UK).

#### 2.4.3 Titration of reagents

Several titrations were done to choose the best cytokine concentration for putative rabbit Tregs and Treg target cell development. These include:

#### 2.4.3.1 Interleukin-2 (IL-2)

Four concentrations of IL-2 were added to 1 X 10<sup>6</sup> MLN cells/ml in IMDM complete medium (0.1ng/ml, 1ng/ml, 10ng/ml, and 50 ng/ml) and

incubated for 4 days, when viable cells (trypan blue exclusion) were counted by haemocytometer.

#### 2.4.3.2 Con A, PHA and PMA titration

These were titrated for optimal stimulation of MLN cells to develop putative rabbit Tregs or Treg assay T cells. For this purpose, 1 X 10<sup>6</sup> MLN cells/ml in IMDM medium were divided into 13 groups and treated as follows:

Group	Contents
G1	Con A 1 µg/ml
G2	Con A 5 µg/ml
G3	Con A 10 µg/ml
G4	Con A 1 µg/ml + 10 ng/ml PMA
G5	Con A 5 µg/ml + 10 ng/ml PMA
G6	Con A 10 µg/ml + 10 ng/ml PMA
G7	PHA 2 µg/ml
G8	PHA 5 µg/ml
G9	PHA 10 µg/ml
G10	PHA 2 µg/ml + 10 ng/ml PMA
G11	PHA 5 µg/ml + 10 ng/ml PMA
G12	PHA 10 µg/ml + 10 ng/ml PMA
Control	Only MLN cells without any additives

#### 2.4.4 TLR Ligands (Agonists)

Human TLR1-9 agonist Kit (tlrl-kit1hw, InvivoGen, San Diego, USA), a commercial kit, was used to stimulate bovine PBMCs and fractionated T cell

subsets (CD4, CD8 and  $\gamma\delta$  T cells) seeded in RPMI complete medium. The kit components contained a wide range of TLR agonists synthesised from microbial components (mainly bacterial, viral and *Mycoplasma spp*) tested to stimulate human PBMCs and approved as specific stimulants (table 2.8). Instructions from the supplier were followed to inform dose response experiments to select optimally active concentrations for this study.

The concentrations of these TLR ligands used to stimulate bovine PBMCs and/or fractionated T cell subsets along with positive controls are illustrated in table 2.9.

Agonists	Product	Working conc	Stock solution
			conc
TLR1/2	Pam3CSK4	0.1-1 µg/ml	100 µg/ml
TLR2	HKLM	10 <sup>8</sup> cells/ml	10 <sup>10</sup> cells/ml
TLR3	Poly(I:C)	10 ng-10 µg/ml	1 mg/ml
TLR3	Poly(I:C) LMW	30ng-10 µg/ml	1 mg/ml
TLR4	LPS	10 ng-10 µg/ml	100 µg/ml
TLR5	Flagellin	10 ng-10 µg/ml	100 µg/ml
TLR6/2	FSL-1	1 ng-1 µg/ml	100 µg/ml
TLR7	Imiquimod	0.25-10 µg/ml	100 µg/ml
TLR8	ssRNA40	0.25-10 µg/ml	100 µg/ml
TLR9	ODN2006	5 μΜ	500 µM

#### Table 2.8: Human TLR1-9 Agonist Kit components

Pam3CSK4 = synthetic tripalmitoylated lipopeptide, HKLM = heat-killed preparation of *Listeria monocytogenes*, Poly(I:C) = synthetic analogue of double-stranded RNA, Poly(I:C) LMW = the same but with low molecular weight, LPS = Lipopolysaccharide, FSL-1 = synthetic lipoprotein, Imiquimod = imidazoquinoline amine, ssRNA40 = 20-mer phosphothioate protected single-stranded RNA oligonucleotide, ODN2006 = synthetic oligonucleotides containing unmethylated CpG dinucleotides.

Working Conc 3

5 ug/ml

1 ug/ml

Product	Agonist	(Low)	(Moderate)	(High)
		L	м	н
Pam3CSK4	TLR1/2 Agonist	0.2 µg/ml	0.5 µg/ml	1 µg/ml
HKLM	TLR2 Agonist	1 X 10 <sup>7</sup> cells/ml	5 X 107 cells/ml	1 X 10 <sup>8</sup> cells/ml
Poly(I:C)	TLR3 Agonist	2.5 µg/ml	5 µg/ml	10 µg/ml
Poly(I:C) LMW	TLR3 Agonist	1 μg/ml	5 µg/ml	10 µg/ml
LPS	TLR4 Agonist	0.1 µg/ml	0.5 µg/ml	1 µg/ml
Flagellin	TLR5 Agonist	0.1 µg/ml	0.5 µg/ml	1 µg/ml
FSL-1	TLR6/2 Agonist	0.1 µg/ml	0.5 µg/ml	1 µg/ml
Imiquimod	TLR7 Agonist	0.5 µg/ml	1.5 µg/ml	2.5 μg/ml
ssRNA40	TLR8 Agonist	0.5 µg/ml	1 µg/ml	3 µg/ml
ODN2006	TLR9 Agonist	0.3 µM	1 µM	3 µM

Working Conc 1

**Working Conc 2** 

Table 2.9: TLR ligand concentrations used for stimulation of PBMCs and/or T cell subsets.

\* CD3 = anti-bovine CD3 monoclonal antibody (WS0561B-100, Kingfisher Biotech, St. Paul, USA)

Concanavalin A

CD3 \*

**Controls used** 

anti-bovine CD3

monoclonal antibody

Con A

Control +ve (For PBMCs

Control +ve (for CD4 &

& γδ T cells).

CD8 T cells).

# 2.5 Methods specific to the putative rabbit Treg study

#### 2.5.1 Generation of rabbit regulatory T cells

Putative rabbit Tregs were generated *in vitro*. Cryotubes of rabbit MLN cells were thawed gently in the water bath at 37°C and transferred dropwise into a 15ml falcon tubes loaded with pre-warmed IMDM complete medium (containing glutamax, 1% penstrep and 10% FCS) then spun at 300 g for 5 minutes to wash out any remaining DMSO (repeated twice). Cells were cultured in 24 well plates by seeding 2 X 10<sup>5</sup> MLN cells/ml in 1 ml IMDM complete culture medium for 6 days incubation at 37°C in a humidified atmosphere of 5% CO<sub>2</sub> in air. IL-2 (50 ng/ml), TGF- $\beta$ 1 (2 ng/ml) and Con A (1 µg/ml) were added to the culture as required according to the protocol. After 6 days incubation flow cytometry was done.

#### 2.5.2 Flow cytometry detection of putative rabbit Tregs

After 6 days incubation, RPMI 1640 (61870010 - Thermofisher Scientific, Rugby, UK) was used with 2% FCS supplement to suspend cells for labelling with fluorescent-labelled antibodies: anti-CD4, anti-CD25 and anti-FOXP3 conjugated with three types of fluorochromes, Fluorescein isothiocyanate (FITC, for anti-CD4), Phycoerythrin (PE, for anti-CD25) and Allophycocyanin (APC, for anti-FOXP3) respectively (see tables 2.10 and 2.11 for antibodies used and their dilutions / working concentrations), also see appendix 8.2.

For this purpose, cells were spun at  $250 \times g$  (benchtop centrifuge) for 7 minutes and pelleted, then RPMI culture medium was added. Cells were

counted to 5 X 10<sup>5</sup> cells/45ul and placed in 1.5 ml eppendorf tubes. Antibodies (5ul, 1:20 diluted stock) mouse IgG2b anti-rabbit CD25 (unconjugated), 5ul (1:20 diluted) mouse IgG2b purified isotype control for CD25, 5ul mouse IgG2a anti-rabbit CD4 FITC, and 5ul mouse IgG2a CD4 isotype control-FITC were added to labelled eppendorf tubes (table 2.11) and incubated in the dark in the fridge (4 °C) for 30 minutes. After that, cells were centrifuged at 2000 rpm for 2 minutes and washed twice with RPMI then 5ul of (1:20 diluted) goat anti-mouse IgG2b (R-PE-labelled) was added to the labelled wells and again incubated in the dark in the fridge for 30 minutes. Cells were washed twice by spinning at 2000 rpm for 2 minutes leaving 100ul supernatant.

FOXP3/Transcription Factor Staining Buffer kit (00-5523 – Thermofisher Scientific, Rugby, UK) containing cell permeabilization Buffer (00-8333 -Thermofisher Scientific, Rugby, UK) was used to make pores in putative rabbit Tregs to allow anti FOXP3 antibody penetration. The active components are formaldehyde (for fixation) and saponin (for permeabilisation). To all wells, 200 µl of eBioscience fixation/ permeabilisation working solution (supplied in the kit), was added. It was prepared by dilution of the fixation/ permeabilization concentrate in fixation/ permeabilization diluent (1 in 4), which was then added to wells and incubated for 30 minutes in the fridge, centrifuged at 2000 rpm for 2 min and supernatants discarded. 45 µl of eBioscience permeabilization buffer prepared by dilution of 10X eBioscience permeabilization buffer (supplied in kit) 1 in 10 distilled water) was added to all tubes. 5µl of rat IgG2a isotype control (FOXP3 isotype control) was added to control cell

wells and 5  $\mu$ l of anti-rat/mouse FOXP3 to experimental cell wells were added then plates were incubated overnight in a fridge.

Next day, all cells were washed twice with eBioscience permeabilization buffer by centrifugation at 2000 rpm for 2 min. Cells were fixed with 2% PFA and finally suspended in 0.5 ml of PBS and shipped to the Queens Medical Centre (QMC, University of Nottingham) to be analysed in a Beckman Coulter MoFlo XDPflow cytometer for cell counting with the help of David Onion. Data were analysed by Walter & Eliza Analysis Software: Eclectic & Lucid abbreviated as WEASEL software.

# **Ch2: Materials & Methods**

#### Table 2.10: Antibodies used for labelling putative rabbit Tregs.

Ab	Isotype/ Clone	Primary/	Stock	Working	Cat #, supplier
		secondary	conc	conc	
Mouse IgG2a isotype control:FITC	IgG2a	Primary	0.1 mg/ml	10 µg/ml	MCA929F- BIO-RAD, Watford,
					UK
Mouse anti rabbit CD4:FITC	IgG2a / KEN-4	Primary	0.1 mg/ml	10 µg/ml	MCA799F - BIO-RAD, Watford,
					UK
Mouse IgG2b purified CD25 isotype	IgG2a	Primary	0.1 mg/ml	2 µg/ml	MG2b00 - Thermofisher
control					Scientific, Rugby, UK
Mouse anti rabbit CD25	IgG2b / KEI-alpha1	Primary	1 mg/ml	20 µg/ml	MCA1119GA - BIO-RAD,
(unconjugated)					Watford, UK
Goat anti-Mouse IgG2b Human	IgG2b	Secondary	0.1 mg/ml	2 µg/ml	M32404 – Life technologies,
Adsorbed R-PE					Carlsbad, USA
Anti-Mouse/Rat FOXP3 APC	IgG2a, kappa / FJK-	Primary	0.2 mg/ml	20 µg/ml	17-5773-82 - Thermofisher
	16s				Scientific, Rugby, UK
Rat IgG2a K Isotype Control APC	IgG2a, kappa /	Primary	0.2 mg/ml	20 µg/ml	17-4321-41 - Thermofisher
	eBR2a				Scientific, Rugby, UK

#### 2.5.3 Sorting of putative rabbit Tregs

Putative rabbit Tregs were generated as described above (2.5.1) and identified as being CD4, CD25, FOXP3+ cells. For cell sorting, CD4+ and CD25 high+ T cells were selected as these contained the FOXP3+ cells (as determined by flow cytometry analysis – see results). A fraction of labelled cells were not fixed with 2% PFA but kept alive and suspended in 1 ml complete RPMI medium then labelled with anti-CD4FITC and anti-CD25PE. Samples were subjected to cell sorting using the Beckman Coulter MoFlo XDP Cell Sorter equipped with 488nm and 405nm lasers to obtain forward and side scatter. Emitted fluorescence light was collected using 450/465nm band pass filter. Sorted putative rabbit Tregs (CD4+CD25hi) were gated on FITC and PE quadrants then collected into fresh RPMI complete medium to be used later in the assay.

#### 2.5.4 Putative rabbit Treg functional assay: target cell

#### proliferation and labelling

Cell trace violet is a DNA dye that binds to DNA as cells divide the daughter cells will have half the signal of the parent cells and so on through subsequent divisions, leading to a progressive diminution of CFSE signal.

Target cells for the putative rabbit Treg assay were autologous MLN cells that were thawed from liquid nitrogen to room temperature inside a class II microbiology safety cabinet under aseptic conditions, then cultured in IMDM complete medium (containing glutamax, 1% penstrep and 10% FCS) for 4 days in a CO<sub>2</sub> incubator with Con A (1  $\mu$ g/ml) + PMA (10 ng/ml). The culture of these cells was co-ordinated such that the cells were ready for use as Treg assay target cells at the same time as the end of the culture period of the generation of putative rabbit Tregs (2.5.1). For labelling with CFSE:  $5X10^5$  target MLN cells/ml were labelled with 1µL of 5 mM CellTrace Violet using CellTrace<sup>TM</sup> Violet Cell Proliferation Kit (C34557 - Thermofisher Scientific, Rugby, UK) for each ml of cell suspension. Cells were incubated for 4 days at 37°C and wrapped with foil (to be protected from light) in the CO<sub>2</sub> incubator. A labelled sample was taken at day 0 to be a control for CFSE labelling efficiency of the cells.

### 2.5.5 Flow cytometry measurement of putative rabbit Treg functional assay T cells (inhibition/stimulation of proliferation by Treg dilutions).

Three concentrations of putative rabbit Tregs cells were used to suppress the proliferating CellTrace violet labelled Treg target cells (figure 2.4). These were  $10^6$ , 5 X  $10^5$  and  $10^5$  Tregs cells/ml mixed each with 5 X  $10^5$ cell/ml CellTrace violet labelled target cells in 3 separate groups, while the fourth group was considered as a negative control containing only CellTrace violet labelled cells. The positive control for target cells were those treated with 5 µg/ml ConA. All groups of cells were incubated for 4 days (wrapped with foil) in a CO<sub>2</sub> incubator at 37°C. Samples were fixed with 2% PFA at day zero, 2 and 4 days of incubation then analysed by flow cytometry.



Figure 2.4: Study protocol for the generation of putative rabbit Tregs and the suppression assay.

# 2.6 Molecular techniques specific to the putative rabbit Treg study

#### 2.6.1 Total RNA extraction for MLN and putative Treg cells

A portion of the putative rabbit Treg cells were kept for total RNA extraction at days 0 and 6 post-incubation. Cells were centrifuged and pelleted then counted and adjusted to  $1\times10^7$  cell/ml. A syringe with 21g needle was used for disruption and homogenisation in medium containing 10% βmercaptoethanol (β-ME) to remove the activity of RNases. The RNeasy Plus Mini Kit (74136 - Qiagen, Manchester, UK) was used to extract RNA. Also, RNase-Free DNase Set (79254 - Qiagen, Manchester, UK) was used to digest any DNases. Eventually, the final concentration was measured by the Nanodrop machine (Thermofisher Scientific, UK).

#### 2.6.2 cDNA synthesis

The Transcriptor First Strand cDNA Synthesis Kit (04897030001 – Roche, Burgess Hill, UK) was followed for 1  $\mu$ g/ml cDNA preparation. Random hexamer primers were used in the protocol. The samples were loaded in a thermal block cycler machine using a program set up for 25 °C for 10 min, 55 °C for 30 min and 85 °C for 5 min, then held at 4 °C.

#### 2.6.3 SDHA (Housekeeping) gene expression

Succinate dehydrogenase (SDHA) was considered as a reference gene. Rabbit DNA from MLN cells was used as a template for DNA amplification by using a Taq PCR Kit (E5000S - New England Biolabs, Hitchin, UK) (Table

2.12). Conventional (normal) PCR on thermal block cycler machine was set to a program listed in table 2.11 for DNA amplification. The PCR product was run on a 1.5% agarose gel for 70 minutes. The band was excised out by using a sterile clean scalpel then the band was purified by using QIAquick® Gel Extraction Kit (28704 - Qiagen, Manchester, UK). The amount of DNA extracted from the gel was 16.63 ng/ul as measured by Nanodrop.

PCR Phase	Cycles	Temperature & Timing
Initial Denaturation	1 x	95°C 5 mins
Denaturation	35x	94°C 30sec
Annealing	35x	60°C 30sec
Extension	35x	72°C 30 sec
Final extension	1 x	72°C 10mins
Hold	1 x	4°C hold

 Table 2.11: PCR program for rabbit MLN DNA amplification.

#### 2.6.4 Real Time - Polymerase chain reaction (RT-qPCR)

Gene expression assays were performed by running quantitative real-time PCR (RT-qPCR). LightCycler® 480 System (Roche, Burgess Hill, UK) was used to run cDNA samples.

Quantitative real time PCR (RT-qPCR) was performed to quantify FOXP3 gene expression. SDHA (reference gene) cDNA samples were diluted (10-fold serial dilutions) and loaded into qPCR plate. Primer and hydrolysis probe sequences (TaqMan) for both target genes (FOXP3) and reference gene (SDHA) were optimised by Nevi Parameswaran. SDHA was used along

with three other reference genes: glceraldehyde-3-phosphate dehydrogenase (GAPDH),  $\beta$ -actin and 18SRNA. Nevi Parameswaran noted that SDHA was expressed most consistently in all samples in rabbit lymphoid samples studied. Hydrolysis probes were dual labelled with fluorescent reporter dye 6 carboxyfluorescein (FAM) at the 5' end and quencher N,N,N,N'-tetramethyl-6-carboxyrhodamine (TAMRA) at the 3' end.

The RT-qPCR 96 well white colour plate was loaded (as duplicates) with 20 ul for each well containing the final working concentration as follows; 10ul of 1x Probes Master Kit (04707494001 - Roche, Burgess Hill, UK), 3.6 ul of 900 nM primers (for each forward and reverse) 0.1 ul of 175 nM Universal Probe Library (UPL probes) and finally 0.7 ul PCR grade water (table 2.12). No Template Control (NTC, everything except the cDNA) was included. 10-fold serial dilutions of samples were applied to give a standard curve(s) for all reference and target genes which was/were measured by LightCycler® 480 software (Roche). The RT-qPCR 96 well plate was covered by a heat-resistant seal (applied manually). The typical value for a standard efficiency curve should range between 1.95 – 2.05 (Roche) and this was achieved by following absolute quantification analysis for the reference gene SDHA, while the results of target gene expression were evaluated by applying relative quantification analysis (Goni et al., 2009, D'haene et al., 2010, Liu et al., 2010, Mayo et al., 2010, Weaver et al., 2010, Hindson et al., 2011, Pinheiro et al., 2011, Bolha et al., 2012, Ji et al., 2012, Long et al., 2013a, Hartshorne et al., 2014, Ma and Chung, 2014). Primers and hydrolysis probe sequences used in the experiments are listed in table 2.13.

Program name	Cycles	Temp	Duration	Analysis mode
1) Pre-incubation	1	95 °C	10 min	None
2) Amplification	45	Varied	90 min	Quantification
a) Denaturation	45	95 °C	10 sec	None
b) Annealing	45	60 °C	30 sec	Data collection
c) Extension (Optional)*	45	72 ℃	1 sec	None
3) Cooling (Optional)*		4 °C	1 min	None

## Table 2.12: RT-qPCR universal program followed for running samples (Roche).

\* Optional = RT-qPCR program can be run without including this step which is not affecting the quality or quantity of the amplification.

## Table 2.13: RT-qPCR primers and hydrolysis probes used for rabbit target and reference gene during the study\*

Gene	Forward primer	Reverse primer	Hydrolysis Probe (UPL)		
name	(5' - 3')	(5' - 3')	(5' - 3')		
Target gene	е				
FOXP3	CTCTGCACCTTCCCAAGC	CACTTGCACACGCCATTT	UPL number 56		
Reference gene					
SDHA	ACCGTGAAGGGCTCTGACT	TTTCTAGCTCGACCACAGAGG	UPL number 158		

\* Primers and probes were designated and normalised by Nevi Parameswaran (unpublished data).

The data obtained from running samples in a 96 RT-qPCR well plate were analysed according to Pfaffl method (formula) (Pfaffl, 2001, Pfaffl *et al.*, 2002, Bustin *et al.*, 2009, Taylor *et al.*, 2010). Below is the Pfaffl equation used for relative quantification:

Ratio = 
$$\frac{(E_{target})^{\Delta C_{T}, \text{ target (calibrator - test)}}}{(E_{ref})^{\Delta C_{T}, \text{ ref (calibrator - test)}}}$$

The ratio of the target gene is expressed for a sample compared to a control in comparison to a reference gene.  $E_{target}$  is the real-time PCR efficiency of the target gene transcript;  $E_{ref}$  is the real-time PCR efficiency of a reference gene transcript;  $\Delta CT_{target}$  is the CT deviation of control (calibrator) – sample of the target gene transcript;  $\Delta CT_{ref} = CT$  deviation of control (calibrator) – sample of the reference gene transcript.

#### 2.7 Bovine lymphocyte study

#### 2.7.1 Cytospin and Diff Kwick staining

The frozen bovine PBMCs were thawed from liquid nitrogen with prewarmed RPMI 1640 complete medium added dropwise, centrifuged at 300g for 10 min, supernatant was disposed of and cells re-suspended in 10 ml medium. 200ul of cell suspension was taken (containing cells that ranged between 1 X 10<sup>5</sup> to 2 X 10<sup>5</sup> cells) for cytocentrifugation at 500 x rpm for 5 minutes in the Shandon Cytospin®4 Cytocentrifuge (WD7020 -Thermofisher Scientific, Rugby, UK). The slides were dried overnight (left at room temperature) and stained with Diff Kwick stain (9990700 – Thermofisher Scientific, Rugby, UK).

Cover slips were applied, and then cells inspected under the light microscope: CTR500 Leica microscope (Leica Microsystems / Switzerland). This was used for imaging the PBMCs. LAS V3.8 software was utilised to obtain photos from the bright field (BF) area.

#### 2.7.2 PBMC fractionation

The main components of PBMCs are monocytes and lymphocytes plus their subsets. To isolate each cell subset, Magnetic Antibody Cell Sorting (MACS) was used. The whole process involves binding a specific antibody to magnetic microbeads to add to the PBMCs so that detected cells can be attracted to the sides of the tube by a magnet.

# 2.7.3 MACS protocol for fractionation of CD4, CD8, and $\gamma\delta$ T cells

The following protocol was designated by Miltenyi Biotec company to separate cells from a total of  $1 \times 10^7$  cells, thus the numbers, volumes and concentrations could be doubled or tripled when the starting count is doubled or tripled accordingly.

The primary step involved labelling cells with purified Abs (primary Abs) (Table 2.14). For the secondary step, secondary Abs specific for the primary antibody isotypes conjugated with microbeads were used.

Mouse anti-bovine	Isotype/	Working	Cat #
Ab (mAb†)	Clone	conc	(BIO-RAD,
			Watford, UK)
CD4	IgG1/ CC30	20 µg/ml	MCA834GA
CD8	IgG2a/ CC63	20 µg/ml	MCA837GA
Gamma delta	IgG2a/ CC15	20 µg/ml	MCA838G

Table	2.14:	Purified	Abs	used	in	the	primarv	stage	labelling.
							P		

+ mAb = Monoclonal Antibody

PBMCs (1 X 10<sup>7</sup> cells) were thawed from liquid nitrogen in pre-warmed RPMI 1640 complete medium (washed twice by spinning at 300 g for 10 min, supernatant discarded). A cell strainer was used to filter the cells to avoid having any clumps (clumps could block the columns). Cells were suspended in 95uL MACS running buffer (see appendix 8.1.3) which is made up by adding 4 ml ethylenediamintetraacetic acid (EDTA 500mM) + 5 gm bovine serum albumin (BSA) dissolved in 1 L PBS (commenced with

800 ml then finalised to 1 L). pH adjusted to 7.2 then sterile filtered and finally stored in the fridge (MACS running buffer). The cells were then mixed with 5uL purified mouse anti-bovine CD4 Ab (MCA834GA / BIO-RAD, Watford, UK) (dilution factor 1:20), and incubated in the fridge for 15 min. After that, cell suspension was spun at 300xg for 10 min, and the supernatant discarded. For the secondary step, 20uL goat anti-mouse IgG microbeads stock concentration (130-048-401 - Miltenyi Biotec, Woking, UK) + 80 uL MACS running buffer was added to the tubes and incubated in the fridge for 15 min. Cells were centrifuged at 300xg for 10 min, then the supernatant discarded and cell pellet re-suspended in 500uL Running buffer.

Magnetic separation of antibody-bound cells from others was achieved using the MidiMACS<sup>™</sup> Separator (130-042-302 - Miltenyi Biotec, Woking, UK) inside a class II microbiology safety cabinet. LS columns (130-042-401 - Miltenyi Biotec, Woking, UK) were rinsed with 3 ml of rinsing buffer (2 ml EDTA (500mM) dissolved in 1 L PBS (started with 800 ml then topped up to 1 L and labelled as MACS rinsing buffer, see appendix 8.1.4). pH adjusted to 7.2, then sterile filtered was loaded inside the column to make sure that the column was washed. A 15-ml falcon tube was put underneath the magnet (for collection of the negative selection). The 500uL cell suspension was transferred onto the column and left dripping. 3 ml of rinsing buffer was added into the column (repeat 3 times). The column was removed from the magnet then loaded with 5 ml rinsing buffer, the plunger was applied and pressed quickly to collect the positive selected (bound) cells in a new sterile 15 ml falcon tube (positive selection = CD4+ T cells). Fractionated (CD4) cells were counted by haemocytometer.

Similar steps were followed by re-using the negative selection collected from above method to separate the other cell subsets, using (MCA837GA-BIO-RAD, Watford, UK) and (MCA838G - BIO-RAD, Watford, UK) as primary antibodies to enrich for CD8 T cells and  $\gamma\delta$  T cells respectively (see table 2.14). The purity of each of the above fractionated subsets was tested by flow cytometry which is described in the next section.

#### 2.7.4 Flow cytometry

PBMCs or fractionated T cell subsets in RPMI 1640 (61870-010 Thermofisher Scientific, Rugby, UK) supplemented with 2% FCS were counted to 1 X 10<sup>6</sup> cell/45ul and placed in eppendorf tubes. A group of cells were left as non-stained controls (NS), while 5 ul of antibodies conjugated with fluorochromes was added to each tube on ice ( $\sim$ 0°C) containing 45uL cell suspension. Cells were labelled with the Abs listed in table 2.17, also see appendix 8.3. Generally, those Abs were added at 1:100 dilutions of stock according to manufacturer's instructions. The negative controls (Mouse IgG2a negative control and Mouse IgG2b negative control) were used at 1:100 as recommended by the supplier as well as titrated and tested before use.

 Table 2.15: Antibodies conjugated with fluorophores used to label bovine PBMCs.

Cell Target	Mouse anti-bovine (Ab)	Isotype/	Working	Cat # / BIO-RAD,
	specific antigen	clone	Conc	Watford, UK
Monocytes	CD11b FITC	IgG2b/ CC126	1 µg/ml	MCA1425F
B cells	CD21 FITC	IgG1/ CC21	1 µg/ml	MCA1424F
CD4 T cells	CD4 FITC	IgG2a/ CC8	1 µg/ml	MCA1653F
CD8 T cells	CD8 FITC	IgG2a/ 38.65	1 µg/ml	MCA2216F
γδ T cells*	WC1 (CC15 clone) FITC	IgG2a/ CC15	1 µg/ml	MCA838F
Mouse IgG2a	FITC (negative control for CD4,	IgG2a/ OX-34	1 µg/ml	MCA929F
Negative Control	CD8, γδ T cells & B cells)			
Mouse IgG2b	FITC (negative control for	IgG2b	1 µg/ml	MCA691F
Negative Control	monocytes)			

\*  $\gamma\delta$  T cells = gamma delta T cells.

After mixing different antibodies (fluorochromes conjugated) with the cell suspension, the mixtures were incubated in the fridge for 30 min, wrapped with foil to avoid light exposure. After incubation, cell suspensions were centrifuged at 300g for 5 min, supernatant removed, and the pellet was topped up with 500 ul PBS and analysed by a BD FACS CANTO II flow cytometer (Becton Dickinson, Bioscience, USA).

The FACS Diva software (BD Bioscience, USA) analysis tool was used to calibrate and run the samples. Non-stained cells (NS) were used as a guide to set up the proper relevant gate(s) for the group(s) of cells of interest. Forward scatter (FSC), and side scatter (SSC) were determined according to the non-stained cell distribution and density. In general, 10,000 events were selected for cell acquisition. CellQuest pro-software (BD Bioscience, USA) was used for data analysis.

Kaluza software, an advanced software for FACS analysis, version 1.5 from Beckman Coulter / Life Sciences was utilised to analyse samples labelled with multicolour panel of fluorophores conjugated antibodies.

#### 2.7.5 Major T cell subsets multicolour fluorophores panel

The major bovine T cell subsets were labelled within the pool of bovine PBMCs at the end of 3 days incubation and CD4, CD8, and  $\gamma\delta$  T cells were labelled (table 2.17) and also (see appendix 8.3) as follows; mouse anti bovine CD4:RPE (MCA1653PE - BIO-RAD, Watford, UK), mouse anti bovine CD8:Alexa Fluor® 647 (MCA837A647 - BIO-RAD, Watford, UK) and mouse anti bovine WC1:FITC (MCA929A647 - BIO-RAD, Watford, UK) respectively along with their isotype negative controls; mouse IgG2a negative control:RPE (MCA929PE- BIO-RAD, Watford, UK), mouse IgG2a negative control:

Alexa Fluor® 647 (MCA929A647 - BIO-RAD, Watford, UK), and mouse IgG2a negative control:FITC (MCA929F - BIO-RAD, Watford, UK) for CD4, CD8 and  $\gamma\delta$  T cells respectively.

Further to these 3 fluorophores, labelling with CellTrace<sup>™</sup> Violet Cell Proliferation Kit (C34557 - Thermofisher Scientific, Rugby, UK) or so called CFSE stain was done by mixing PBMCs at day 0 at 1X10<sup>6</sup> cells/ml with 1µL of 5 mM CellTrace Violet inside a class II microbiology safety cabinet and moved outside the hood and placed for 20 minutes in cold (4 °C) dark environment. 4x RPMI complete medium was added to wash out excess stain (to remove the toxic effect) and incubated in a dark cold environment for 5 minutes, then spun at 300 g for 5 minutes and supernatant discarded. Cells were resuspended in fresh RPMI complete medium and incubated for 3 days at 37°C wrapped with foil (protected from light) in a 5% CO<sub>2</sub> incubator. A fraction of cells was left non-stained as a control for the staining protocol. Labelled samples were harvested at the end of the incubation, washed and fixed with 2% PFA then washed and resuspended in 500ul PBS and analysed by flow cytometry.

Table 2.16: Fluorochrome-conjugated antibodies used for labelling major T cell subsets (CD4, CD8 and γδ T cells) within the pool of bovine PBMCs

Monoclonal antibody (mAb)	Stock conc	Working	Dilution	Isotype / Clone	Cat # / (BIO-RAD,
		conc	factor		Watford, UK)
MOUSE ANTI BOVINE CD4:RPE	0.1 mg/ml	1 µg/ml	1:100	IgG2a / CC8	MCA1653PE
MOUSE IgG2a NEGATIVE CONTROL:RPE	0.1 mg/ml	1 µg/ml	1:100	IgG2a / MRC OX-34	MCA929PE
MOUSE ANTI BOVINE CD8:Alexa Fluor®	0.05 mg/ml	0.5 µg/ml	1:100	IgG2a / CC63	MCA837A647
647					
MOUSE IgG2a NEGATIVE CONTROL:	0.05 mg/ml	0.5 µg/ml	1:100	IgG2a / MRC OX-34	MCA929A647
Alexa Fluor® 647					
MOUSE ANTI BOVINE WC1:FITC	0.1 mg/ml	1 µg/ml	1:100	IgG2a / CC15	MCA838F
MOUSE IgG2a NEGATIVE CONTROL:FITC	0.1 mg/ml	1 µg/ml	1:100	IgG2a / MRC OX-34	MCA929F

#### 2.8 Molecular techniques specific to Bovine

#### lymphocyte study

#### 2.8.1 Total RNA extraction

RNeasy plus mini kit (74134 – Qiagen, Manchester, UK) was used for total RNA extraction. This is a highly efficient, sensitive protocol for low amounts, and phenol-free method for RNA extraction with which up to 100 µg total RNA can be obtained for sensitive applications such as quantitative real-time PCR (Hanoux et al., 2007). Generally, 1 X 10<sup>7</sup> PBMCs (or any other cell type) were initially mixed with 600 ul of RLT buffer (cell lysis stage) and gently vortexed for 30 seconds (medium speed vortexing until clear translucent solution formed). A syringe with 21g needle was used for disruption and homogenisation in RLT buffer containing 10%  $\beta$ -mercaptoethanol ( $\beta$ -ME) to remove the activity of RNases. The lysates were transferred into a gDNA eliminator spin column (for genomic DNA removal) put on a 2-ml collection tube and spun at 8000 g for 30 seconds. The column was disposed of and the flow-through was collected and topped up by an equal volume of 70% ethanol, then pipetted up and down gently. 700 ul of the mixture was added to a RNeasy spin column which was spun at 8000 g for 15 seconds. The flowthrough was discarded and this step was repeated with the rest volume of RNA- alcohol mixture. After that, 350 ul of RWT solution (washing step) was added to the spin column and spun for 15 seconds at 8000 g. To digest the genomic DNA, 10 ul DNase I stock was mixed thoroughly and carefully with 70 ul Buffer RDD (both supplied in RNase-Free DNase Set 79254 - Qiagen, Manchester, UK) and loaded to the spin column then incubated for 30 minutes in the fridge then spun at 8000 g for 15 seconds. 350 ul of RWT was loaded to the spin column (washing step) and spun at 8000 g for 15 seconds. 500 ul

of RPE buffer was added to the spin column (to remove any undesired chemical or protein precipitates) then was spun at 8000 g for 15 seconds. The RPE washing step was repeated again but for 2 minutes. The spin column was put over 1.5 eppendorf tube and loaded with 40 ul RNase-free water for final elution which was spun at 8000 g for 60 seconds. All centrifugations were performed by using benchtop centrifuge.

#### 2.8.2 RNA quantification (Nanodrop®8000)

The Nanodrop®8000 machine (Thermofisher Scientific, UK) was used to measure the quality and quantity of RNA. It is basically a spectrophotometer that uses a UV light then absorbs the refracted beam light (from 220 to 750 nanometre wavelength). DNA (single and double stranded) as well as RNA samples can be measured at 260 nm wavelength. The machine is blanked with 1 ul PCR grade water, then the sample was loaded and measured accordingly. Up to 8 samples can be loaded at once. After measuring the concentration, RNA samples were stored in a -80 °C freezer.

#### 2.8.3 Complementary DNA synthesis

Complementary DNA (cDNA) was synthesised by following the Roche kit instructions: Transcriptor First Strand cDNA Synthesis Kit (04897030001 -Roche, Burgess Hill, UK) was followed to make cDNA at 1 ug/ul. Random hexamer primers were used in the protocol. Ice was used in this protocol to ensure cold environment. The components added are listed in table 2.17.

Component	Final conc.	Volume
Random hexamer primer	60 uM	2 µl
Transcriptor Reverse Transcriptase Reaction Buffer	1x (8mM MgCl <sub>2</sub> )	4 ul
Protector RNase Inhibitor	20 U	0.5 ul
Deoxynucleotide Mix	1 mM	2 ul
Transcriptor Reverse Transcriptase	10 U	0.5 ul
Total volume	-	9 ul
RNA volume (with or without water)	Varied	11 ul
Final working volume	-	20 ul

#### Table 2.17: cDNA synthesis kit reagents (<u>www.roche-applied-science.com</u>).

Sterile 0.2 ml nuclease-free, thin-walled PCR tubes were used which were loaded into a thermal block cycler machine (Alpha laboratories) using a program set up for 25 °C for 10 min, 55 °C for 30 min and 85 °C for 5 min, then held at 4 °C. cDNA samples were stored in the freezer at -20 °C.

#### 2.8.4 Polymerase chain reaction (PCR)

#### 2.8.4.1 Conventional PCR

Bovine PBMCs were subjected to RNA extraction and cDNA synthesis according to the above protocols. cDNA was used as a template to amplify DNA by using the Taq PCR Kit (E5000S - New England Biolabs, Hitchin, UK). Company instructions were followed as recommended. All kit components (except the enzyme "Taq") were thawed and placed on ice. Sterile 0.2 ml nuclease-free, thin-walled PCR tubes were used for this purpose. All the components were added inside PCR cabinets except the DNA which was added

outside the cabinet. A thermal block cycler (Alpha laboratories) was used to run the program. The components for each reaction and the cycling conditions are described in tables 2.18 and 2.19.

 Table 2.18: PCR components for bovine cDNA amplification.

	-	
PCR materials	For 1	x reaction
Nuclease free water	19.85	ul
10x buffer	2.5	ul
dNTPs (10mM)	0.5	ul
10mM Forward Primer (10 pmol/ul)	0.5	ul
10mM Reverse Primer (10 pmol/ul)	0.5	ul
DNA	1	ul
Taq (5u/ul)	0.15	ul
Total volume	25	ul

#### Table 2.19: PCR program for bovine cDNA amplification.

PCR Phase	Cycles	Tempe	erature & Timing
Initial Denaturation	1 x	95°C	2 mins
Denaturation	30x	94°C	30sec
Annealing	30x	60°C	30sec
Extension	30x	72°C	1 min
Final extension	1 x	72°C	5 mins
Hold	1 x	4°C	hold

#### 2.8.4.2 Gel electrophoresis

The amplicon resulting from PCR amplification was run on 2% gel electrophoresis (small size amplicon products require higher qel concentration), 5 ul of Nancy-520 solution (01494 - Sigma-Aldrich, Haverhill, UK) was added as a visualising agent and as a safer alternative DNA stain to ethidium bromide. The comb from the set gel was removed to load the samples. This included 1 ul of 6X gel loading dye (N3233S - New England Biolabs, Hitchin, UK) mixed with 5 ul DNA product (amplicon) on parafilm then 5 ul loaded into the well. 5 ul from Low molecular weight DNA ladder (N3233S - New England Biolabs, Hitchin, UK) was mixed with 1 ul 6X gel loading dye, then 5 ul was loaded to each terminal well (ladder). Tris-acetate-EDTA buffer (TAE Buffer) working solution of 40 mM Tris acetate (pH 8.5) which was (1X TAE) simply made of as a 50X TAE stock solution (see appendix 8.1.5) by dissolving 242gm Tris free base and 18.61 gm Disodium EDTA in 57.1 ml Glacial acetic acid all mixed with 700 ml double ionised distilled water then stirred gently and topped up to 1 L. To make up 1 L from 1X TAE buffer (see appendix 8.1.6), 20 ml 50X TAE was aspirated and topped up with double ionised distilled water to 1 L. The 1x TAE solution is 40mM Tris, 20mM Acetate and 1mM EDTA, pH approximately 8.5. The settings for the electrophoresis apparatus were 90V, 400 Amp for 90 min. The bands were viewed by the UV illuminator (63005650 - ImageQuant 300 Imager (Amersham Bioscience, Part of GE Healthcare, UK).

#### 2.8.4.3 Quantitative real-time PCR (RT-qPCR)

Gene expression assays were performed by running quantitative real-time PCR (RT-qPCR). The apparatus LightCycler® 480 System (Roche Applied Science, UK) was used to run the samples (cDNA samples) (Table 2.20). Primers and hydrolysis probe sequences (TaqMan) for both target genes (TLRs 1-10) and three reference genes; Glyceraldehyde 3-phosphate dehydrogenase (GAPDH), Ribosomal protein, large, P0 (RPLP0) and Ribosomal protein, large, P2 (RPLP2) were taken from previously published work which were used by other researchers (Werling *et al.*, 2006, Gibson *et al.*, 2012, Russell, 2012). Primers and probes were blasted online (Zhang *et al.*, 1997). Both primers and hydrolysis probes were ordered from Sigma Aldrich. Hydrolysis probes were labelled (dual labelled) with fluorescent reporter dye 6 carboxyfluorescein (FAM) at the 5' end and quencher N,N,N,N'-tetramethyl-6-carboxyrhodamine (TAMRA) at the 3' end.

The RT-qPCR 96 well white colour plate was loaded (as duplicates) with 20 ul for each well containing the final working concentration as follows; 10ul of 1x Probes Master Kit (04707494001 - Roche Life Science, UK), 3.6 ul of 900 nM primers (for each forward and reverse) 0.7 ul of 175 nM non-UPL probes and finally 0.7 ul PCR grade water (table 2.22). No Template Control (NTC) was included. 10-fold serial dilutions of samples were applied to give a standard curve(s) for all reference and target genes which were measured by LightCycler® 480 software (Roche). The RT-qPCR 96 well plate was covered by a heat-resistant seal (applied manually). The typical value for a standard efficiency curve should be ranged between 1.95 – 2.05 (Roche) and this was achieved by following absolute quantification analysis while the results of target gene expression were evaluated by applying relative quantification

analysis (Goni *et al.*, 2009, D'haene *et al.*, 2010, Liu *et al.*, 2010, Mayo *et al.*, 2010, Weaver *et al.*, 2010, Hindson *et al.*, 2011, Pinheiro *et al.*, 2011, Bolha *et al.*, 2012, Ji *et al.*, 2012, Long *et al.*, 2013b, Hartshorne *et al.*, 2014, Ma and Chung, 2014). Primers and hydrolysis probe sequences used in the experiments are listed in table 2.21

Program name	Cycles	Temp	Duration	Analysis mode
1) Pre-incubation	1	95 ℃	10 min	None
2) Amplification	45	Varied	90 min	Quantification
a) Denaturation	45	95 °C	10 sec	None
b) Annealing	45	60 °C	30 sec	Data collection
c) Extension (Optional)*	45	72 ℃	1 sec	None
3) Cooling (Optional)*		4 °C	1 min	None

## Table 2.20 RT-qPCR universal program followed for running samples (Roche).

\* Optional = RT-qPCR program can be run without including this step which is not affecting the quality or quantity of the amplification.

**Ch2: Materials & Methods** 

Gene name	Forward primer	Reverse primer	Hydrolysis Probe (non-UPL)				
	(5' - 3')	(5′ – 3′)	(5′ – 3′)				
Target genes							
BoTLR1	GCACCACAGTGAGTCTGGAA	GTACGCCAAACCAACTGGAG	TGTGTGCTTGATGATAATGGGTGTCCT				
BoTLR2	ACGACGCCTTCGTGTCCTAC	GCTCCTGGACCATGAGGTTC	CGAGCGGGATTCCTACTGGGTGG				
BoTLR3	AAAGAGTTCTCTCCTGGGTGTT	TGCTCAGGGACAGATTCTCA	CAATGCCAAGCTGAGCCCCA				
BoTLR4	TGGAGGACATGCCAGTGCT	CACCGACACACTGATGATCGT	AGTTTCAGGAACGCCACTTGTCAGCTG				
BoTLR5	CTAGACCTGGGTGGAAGTCAG	AGGGATGAAGGTAAAGACTCTGAA	TTCCTGTGGTCTCTCCGATGCTG				
BoTLR6	CCTGCCCATCTGTAAGGAAT	TAGGTGCAAGTGAGCAATGG	TTGGCAACTTGACCCAACTGAATTTC				
BoTLR7	GCTGAAGACTGTCCCTGAGA	TTTGAGCTGAGGTCCAGATG	TCCAACTGTTCCCGCAGCCTC				
BoTLR8	TCCACATTTGAAACGAAGACC	ACATCGGTCAGTCTGGGAAC	CCTGACGTTCAGATTTCTGTCCATC				
BoTLR9	CACCATCTTCAACGACCTGA	CTTCTCCAGGGACACCAGAC	TCCTTCGCCCACCTGCACCT				
BoTLR10	TGGTTGGATGGTCAGATTCA	CAGGGCAAATCAAAGTGGA	CCATTGTTGTCATGCTCGTTCT				
Reference genes							
BoGAPDH	CATGTTCCAGTATGATGATTC	GAGCTTCCCGTTCTCTGC	CGGCAAGTTCAACGGCAC				
BoRPLP0	CTGATTACACCTTCCCACTTGCT	AGCCACAAATGCAGATGGATCA	AAGGCCTTGACCTTTTC				
BoRPLP2	TCAACAAGGTCATCAGTGAGC	CCGATACCCTGAGCAATGA	CGTCCTCGATGTTCTTTCCGTGG				

#### Table 2.21: RT-qPCR primers and hydrolysis probes used for bovine target and reference genes during the study.
#### 2.8.4.4 RT-qPCR method for TLR quantification

RNA was extracted then reverse transcribed into cDNA which was synthesised accordingly then analysed using  $(2^{-\Delta\Delta CT})$  equation.

#### 2.8.4.5 Normalisation of bovine reference genes

Although published primers and hydrolysis probes were used, it was worth confirming that they were able to amplify the genes of interest producing a product represented by a band of nucleic acid on a gel. Thus, bovine reference genes (primers only were used but not the hydrolysis probes) were validated by running PBMC cDNA samples in the Thermal block cycler (Alpha laboratories) following the conventional PCR protocol mentioned above. Annealing temperatures were manipulated starting from 60 °C downward to 58°C to find out the best conditions for the selected primers (both forward and reverse). The Taq PCR Kit (E5000S / New England Biolabs, Hitchin, UK) was used for this purpose following the instructions in "Taq DNA Polymerase Guidelines for PCR Optimization" published by New England Biolabs Inc., UK on their website. Bands on the gels were of the expected size for the reference genes. These were not checked by sequencing though.

# 2.9 Cytokine measurements (ELISA)

Cytokines were measured by collecting the supernatants which were subjected to Enzyme Linked Immuno-Sorbent Assay (ELISA). All manufacturers' instructions for ELISA protocols were followed. In total, 7 cytokines were assayed by purchasing commercial ELISA kits. These were Bovine Interleukin 8 ELISA kit (DIY1028B-003, Kingfisher Biotech Inc, St. Paul, USA), Bovine IFN-aA ELISA kit (DIY0663B-003, Kingfisher Biotech Inc, St. Paul, USA), Bovine TNF-α ELISA kit (DIY0675B-003, Kingfisher Biotech Inc, St. Paul, USA) and Bovine IFN-γ ELISA development kit (3119-1H-6 / Mabtech, Cincinnati, USA), (see appendix 8.4).

Both IL-4 and IL-10 ELISA kits were purchased as "Matched Antibody Pairs" from BIO-RAD supplier. For IL-4, mouse anti bovine interleukin-4 (MCA2371, BIO-RAD, Watford, UK) was used as a capture Ab, while mouse anti bovine interleukin-4:Biotin (MCA2372B, BIO-RAD, Watford, UK) was used as a detection Ab. For IL-10, mouse anti bovine interleukin-10 (MCA2110, BIO-RAD, Watford, UK) was used as a capture Ab, while mouse anti bovine interleukin-10:Biotin (MCA2111B, BIO-RAD, Watford, UK) was used as a detection Ab. Finally, a TGFβ ELISA kit (ABIN996501, Antibodies-online, Aachen, Germany) was used to assay bovine TGF-beta.

#### 2.9.1 Generic materials and common steps

- ELISA washing buffer and washing procedure: the whole plate was washed 3 times with washing buffer which was made of 0.05% Tween<sup>™</sup>20 in PBS (pH 7.2). Fourth wash was done with PBS only. After each wash, the plate was dried off by gentle blotting against tissue paper.
- ELISA Blocking buffer and procedure: all the wells were blocked by adding 300 ul of blocking buffer which was made of 1% bovine serum albumin (BSA) in PBS (pH 7.2). Blocking buffer was prepared by weighing out 1 gm BSA dissolved in 800 ml sterile fresh PBS, then the volume was topped up to 1 L and sterile filtered with 0.22 µm millipore filter. Blocking duration was set for 1 hour.

- ELISA coating buffer consisted of 1.5 g Sodium carbonate (Na<sub>2</sub>CO<sub>3</sub>) plus
  2.93 g Sodium hydrogen carbonate (NaHCO<sub>3</sub>) in 800 ml DW, then topped up to I L and pH adjusted to 9.6.
- ELISA 96 micro-well plate (475094 Thermofisher Scientific, Rugby, UK).

#### 2.9.2 IL-8 (CXCL-8)

To detect IL-8 in the supernatants, Bovine IL-8 ELISA kit (DIY1028B-003, Kingfisher Biotech Inc, St. Paul, USA) includes capture Ab, standard protein and detection Ab. Sandwich ELISA method was used.

Briefly, samples and reagents were warmed up to room temperature before use. All wells in ELISA plate were loaded with 100 ul of capture Ab (Anti-Swine IL-8 Polyclonal Antibody, PB0143S-100, Kingfisher Biotech Inc, St. Paul, USA) in fresh sterile PBS at 2ug/ml, covered with plate sealer and incubated overnight at 25 °C.

The plate was washed by ELISA washing buffer (see section 2.9.1) and blocked by ELISA Blocking buffer (see section 2.9.1). 50 ul of samples were added in duplicates along with 50 ul of standard protein (Recombinant Bovine IL-8, RP0023B-005, Kingfisher Biotech Inc, St. Paul, USA), the standard protein (STD), dissolved in complete RPMI medium diluted as two-fold serial dilutions from 25 ng/ml down to 97.65 pg/ml whereas 2 wells were not loaded with STD where only complete RPMI medium was added.

Two more wells were loaded with PBS only (blank). The plate was sealed and incubated at 25 °C for 2 hours. The plate was washed with ELISA washing buffer (see section 2.9.1). Detection Ab (Biotinylated Anti-Swine IL-8 Polyclonal Antibody, PBB0266S-050, Kingfisher Biotech Inc, St. Paul, USA) was added at 0.3 ug/ml in blocking buffer (1%BSA) and loaded at 100 ul per well each then the plate was sealed and incubated at 25 °C for 1 hour. The plate was washed with ELISA washing buffer (see section 2.9.1).

50 ul of streptavidin-horseradish peroxidase (SA-HRP), the enzyme (AR0068-001, Kingfisher Biotech Inc, St. Paul, USA) was loaded as 1:25 diluted in blocking buffer (1%BSA) then sealed and wrapped with foil (protected from

light) and incubated at 25 °C for 30 minutes. The plate was washed with ELISA washing buffer (see section 2.9.1).

100 ul of ELISA Substrate 3,3',5,5'-tetramethylbenzidine (TMB) (DY999, R & D Systems, Minneapolis, USA) was added to each well by mixing equal volumes from pre-warmed vials A and B. These are stabilized hydrogen peroxide (895000, R & D Systems, Minneapolis, USA) for colour reagent A and stabilized tetramethylbenzidine, (895001, R & D Systems, Minneapolis, USA) for colour Reagent B. The plate was not sealed and incubated in the dark at 25 °C for up to 20 minutes. Once blue colour developed, all the wells were topped up with 100 ul of 0.18M sulphuric acid (H<sub>2</sub>SO<sub>4</sub>) as a stop solution changing the colour from blue to yellow.

Labtech spectrophotometer, a micro-plate reader, (LT-4000, Labtech, UK) was used to measure the absorbance (Optical Density which is abbreviated as "OD" values) at 450nm and 540nm wavelengths. OD values of all the samples were subtracted to omit background signal (corrected) as follows (Mean OD samples at 450nm –mean OD at 540nm). A standard curve line was plotted using recombinant bovine IL-8 (Figure 2.5). GraphPad Prism version 7 was used to interpolate the OD values into concentrations illustrated as ng/ml.



Figure 2.5: ELISA standard curve for Bovine IL-8. Mean OD values were determined at 2-fold serial dilutions of Recombinant Bovine IL-8 from 25 ng/ml down to 97.65 pg/ml concentrations. To generate the straight line, The X axis represents the log10 transformation of the standard concentrations expressed as a geometric mean and the Y axis the log10 of the O.D. expressed as a geometric mean. To get the sample concentrations the values are calculated from the standard curve (in Excel) and reverse transformed (backtransformed) to give a value in ng/ml or pg/ml.

#### 2.9.3 IFNa

Bovine IFN-aA ELISA kit (DIY0663B-003, Kingfisher Biotech Inc, St. Paul, USA) was used to measure IFN-aA in the supernatants. Sandwich ELISA method was used. Briefly, samples and reagents were warmed up to room temperature before use. All wells in ELISA plate were loaded with 100 ul of capture Ab (Anti-Bovine IFN-aA Polyclonal Antibody, PB0474B-100, Kingfisher Biotech Inc, St. Paul, USA) in fresh sterile PBS at 1.6ug/ml, covered with plate sealer and incubated overnight at 25 °C. The plate was washed by ELISA washing buffer (see section 2.9.1) and blocked by ELISA Blocking buffer (see section 2.9.1). 50 ul of samples were added in duplicates along with 50 ul of standard protein (Recombinant Bovine IFN-aA, RP0008B-005, Kingfisher Biotech Inc, St. Paul, USA), the standard protein (STD), dissolved in complete RPMI medium diluted as two-fold serial dilutions from 10 ng/ml down to 78.12 pg/ml whereas 2 wells were not loaded with STD

where only complete RPMI medium added. Two more wells were loaded with PBS only (blank). The plate was sealed and incubated at 25 °C for 2 hours. The plate was washed with ELISA washing buffer (see section 2.9.1). Detection Ab (Biotinylated Anti-Bovine IFN-aA Polyclonal Antibody, PBB0483B-050, Kingfisher Biotech Inc, St. Paul, USA) was added at 0.3 ug/ml in blocking buffer (1%BSA) and loaded at 100 ul per well each then the plate was sealed and incubated at 25 °C for 1 hour. The plate was washed with ELISA washing buffer (see section 2.9.1). 50 ul of SA-HRP, the enzyme (AR0068-001, Kingfisher Biotech Inc, St. Paul, USA), was loaded as 1:25 diluted in blocking buffer (1%BSA) then sealed and wrapped with foil (protected from light) and incubated at 25 °C for 30 minutes. The plate was washed with ELISA washing buffer (see section 2.9.1). 100 ul of ELISA Substrate 3,3',5,5'-tetramethylbenzidine (TMB) (DY999, R & D Systems, Minneapolis, USA) was added to each well by mixing equal volumes from prewarmed vials A and B. The plate was not sealed and incubated in the dark at 25 °C for up to 15 minutes. Once blue colour appropriately developed, all the wells were topped up with 100 ul of 0.18M sulphuric acid (H<sub>2</sub>SO<sub>4</sub>) as a stop solution changing the colour from blue to yellow. Labtech spectrophotometer, a micro-plate reader, (LT-4000, Labtech, UK) was used to measure the absorbance (Optical Density which is abbreviated as "OD" values) at 450nm and 540nm wavelengths. OD values of the all samples were subtracted to omit background signal (corrected) as follows (Mean OD samples at 450nmmean OD at 540nm). A standard curve line was plotted using recombinant bovine IFN-aA (Figure 2.6). GraphPad Prism version 7 was used to interpolate the OD values into concentrations illustrated as ng/ml.



Figure 2.6: ELISA standard curve for Bovine IFNa. Mean OD values were determined at 2-fold serial dilutions of Recombinant Bovine IFN-aA from 10 ng/ml down to 78.12 pg/ml concentrations. To generate the straight line, The X axis represents the log10 transformation of the standard concentrations expressed as a geometric mean and the Y axis the log10 of the O.D. expressed as a geometric mean. To get the sample concentrations the values are calculated from the standard curve (in Excel) and reverse transformed (backtransformed) to give a value in ng/ml or pg/ml.

#### 2.9.4 TNF-a

Bovine TNF-a ELISA kit (DIY0675B-003, Kingfisher Biotech Inc, St. Paul, USA) was used to measure TNF-a in the supernatants. Sandwich ELISA method was used. Briefly, samples and reagents were warmed up to room temperature before use. All wells in ELISA plate were loaded with 100 ul of capture Ab (Anti-Bovine TNF-a Polyclonal Antibody, PB0275B-100, Kingfisher Biotech Inc, St. Paul, USA) in fresh sterile PBS at 1.8ug/ml, covered with plate sealer and incubated overnight at 25 °C. The plate was washed by ELISA washing buffer (see section 2.9.1) and blocked by ELISA Blocking buffer (see section 2.9.1) and blocked by ELISA Blocking buffer (see section 2.9.1). 50 ul of samples were added in duplicates along with 50 ul of standard protein (Recombinant Bovine TNF-a, RP0055B-005, Kingfisher Biotech Inc, St. Paul, USA), the standard protein (STD), dissolved in complete RPMI medium diluted as two-fold serial dilutions from 10 ng/ml down to 78.12 pg/ml whereas 2 wells were not loaded with STD where only complete RPMI

medium added. Two more wells were loaded with PBS only (blank). The plate was sealed and incubated at 25 °C for 2 hours. The plate was washed with ELISA washing buffer (see section 2.9.1). Detection Ab (Biotinylated Anti-Bovine TNF-a Polyclonal Antibody, PBB0278B-050, Kingfisher Biotech Inc, St. Paul, USA) was added at 0.3 ug/ml in blocking buffer (1%BSA) and loaded at 100 ul per well each then the plate was sealed and incubated at 25 °C for 1 hour. The plate was washed with ELISA washing buffer (see section 2.9.1). 50 ul of SA-HRP, the enzyme (AR0068-001, Kingfisher Biotech Inc, St. Paul, USA) was loaded as 1:25 diluted in blocking buffer (1%BSA) then sealed and wrapped with foil (protected from light) and incubated at 25 °C for 30 minutes. The plate was washed with ELISA washing buffer (see section 2.9.1). 100 ul of ELISA Substrate TMB Core+ (BUF062, BIO-RAD, Watford, UK). The plate was not sealed and incubated in the dark at 25 °C for up to 15 minutes. Once blue colour appropriately developed, all the wells were topped up with 100 ul of 0.18M sulphuric acid as a stop solution changing the colour from blue to yellow. Labtech spectrophotometer, a micro-plate reader, (LT-4000, Labtech, UK) was used to measure the absorbance (Optical Density which is abbreviated as "OD" values) at 450nm and 540nm wavelengths. OD values of the all samples were subtracted to omit background signal (corrected) as follows (mean OD samples at 450nm – mean OD at 540nm). A standard curve line was plotted using recombinant bovine TNF-a (Figure 2.7). GraphPad Prism version 7 was used to interpolate the OD values into concentrations illustrated as ng/ml.



Figure 2.7: ELISA standard curve for Bovine TNFa. Mean OD values were determined at 2-fold serial dilutions of Recombinant Bovine IFN-aA from 10 ng/ml down to 78.12 pg/ml concentrations. To generate the straight line, The X axis represents the log10 transformation of the standard concentrations expressed as a geometric mean and the Y axis the log10 of the O.D. expressed as a geometric mean. To get the sample concentrations the values are calculated from the standard curve (in Excel) and reverse transformed (backtransformed) to give a value in ng/ml or pg/ml.

#### 2.9.5 IL-4

In order to measure bovine IL-4 cytokine in the supernatant, Matched Antibody Pairs from BIO-RAD was used. Sandwich ELISA method was used. ELISA plate was loaded with 100 ul of ELISA coating buffer (see section 2.9.1) containing 3ug/ml mouse anti bovine interleukin-4 (MCA2371, BIO-RAD, Watford, UK) which was used as a capture Ab. The plate was sealed and incubated in the fridge overnight. The plate was washed by ELISA washing buffer (see section 2.9.1) and blocked by ELISA Blocking buffer (see section 2.9.1). 50 ul of samples were added in duplicates along with 50 ul of standard protein (recombinant bovine interleukin-4, PBP010, BIO-RAD, Watford, UK), the standard protein (STD), dissolved in ELISA washing buffer diluted as two-fold serial dilutions from 10 ng/ml down to 78.12 pg/ml whereas 2 wells were loaded with zero conc of STD. Two more wells were loaded with PBS only (blank). The plate was sealed and incubated at 37 °C for 2 hours. The plate was washed with ELISA washing buffer (see section 2.9.1). 100 ul of 1 ug/ml

mouse anti bovine interleukin-4:Biotin (MCA2372B, BIO-RAD, Watford, UK) was used as a detection Ab in ELISA washing buffer then the plate was sealed and incubated at 37  $^{\circ}$ C for 1 hour. The plate was washed with ELISA washing buffer (see section 2.9.1). 50 ul of SA-HRP, the enzyme (AR0068-001, Kingfisher Biotech Inc, St. Paul, USA) was loaded as 1:25 diluted in ELISA washing buffer then sealed and wrapped with foil (protected from light) and incubated at 37 °C for 60 minutes. The plate was washed with ELISA washing buffer (see section 2.9.1). 100 ul of ELISA Substrate TMB Core+ (BUF062, BIO-RAD, Watford, UK) was added. The plate was not sealed and incubated in the dark at 25 °C for up to 25 minutes. Once blue colour appropriately developed, all the wells were topped up with 100 ul of 0.2M sulphuric acid as a stop solution. Labtech spectrophotometer, a micro-plate reader, (LT-4000, Labtech, UK) was used to measure the absorbance at 450nm wavelength. OD values of all the samples were blank subtracted to omit background signal (corrected) as follows (mean OD samples at 450nm -mean OD blank). A standard curve line was plotted using recombinant bovine interleukin-4 (Figure 2.8). GraphPad Prism version 7 was used to interpolate the OD values into concentrations illustrated as ng/ml.



Figure 2.8: ELISA standard curve for Bovine IL-4. Mean OD values were determined at 2-fold serial dilutions of Recombinant Bovine IL-4 from 10 ng/ml down to 78.12 pg/ml concentrations. To generate the straight line, The X axis represents the log10 transformation of the standard concentrations expressed as a geometric mean and the Y axis the log10 of the O.D. expressed as a geometric mean. To get the sample concentrations

# the values are calculated from the standard curve (in Excel) and reverse transformed (backtransformed) to give a value in ng/ml or pg/ml.

#### 2.9.6 IL-10

Bovine IL-10 cytokine was measured in the supernatant by purchasing Matched Antibody Pairs from BIO-RAD. A Sandwich ELISA method was used. ELISA plate was loaded with 100 ul of ELISA coating buffer (see section 2.9.1) containing 3ug/ml mouse anti bovine interleukin-10 (MCA2110, BIO-RAD, Watford, UK) which was used as a capture Ab. The plate was sealed and incubated in the fridge overnight. The plate was washed by ELISA washing buffer (see section 2.9.1) and blocked by ELISA Blocking buffer (see section 2.9.1). 50 ul of samples were added in duplicates along with 50 ul of standard protein (recombinant bovine interleukin-10, PBP016A, BIO-RAD, Watford, UK). The standard protein (STD), dissolved in ELISA washing buffer diluted as two-fold serial dilutions from 10 ng/ml down to 78.12 pg/ml whereas 2 wells were loaded with zero conc of STD. Two more wells were loaded with PBS only (blank). The plate was sealed and incubated at 37 °C for 2 hours. The plate was washed with ELISA washing buffer (see section 2.9.1). 100 ul of 1 ug/ml mouse anti bovine interleukin-10:Biotin (MCA2111B, BIO-RAD, Watford, UK) was used as a detection Ab in ELISA washing buffer then the plate was sealed and incubated at 37 °C for 1 hour. The plate was washed with ELISA washing buffer (see section 2.9.1). 50 ul of SA-HRP, the enzyme (AR0068-001, Kingfisher Biotech Inc, St. Paul, USA), was loaded as 1:25 diluted in ELISA washing buffer then sealed and wrapped with foil (protected from light) and incubated at 37 °C for 60 minutes. The plate was washed with ELISA washing buffer (see section 2.9.1). 100 ul of ELISA Substrate TMB Core+ (BUF062, BIO-RAD, Watford, UK) was added. The plate was not sealed

and incubated in the dark at 25 °C for up to 25 minutes. Once blue colour developed, all the wells were topped up with 100 ul of 0.2M sulphuric acid as a stop solution. Labtech spectrophotometer, a micro-plate reader, (LT-4000, Labtech, UK) was used to measure the absorbance (Optical Density which is abbreviated as "OD" values) at 450nm wavelength. OD values of the all samples were blank subtracted to omit background signal (corrected) as follows (mean OD samples at 450nm –mean OD blank). A standard curve line was plotted using recombinant bovine interleukin-4 (Figure 2.9). GraphPad Prism version 7 was used to interpolate the OD values into concentrations illustrated as ng/ml.



Figure 2.9: ELISA standard curve for Bovine IL-10. Mean OD values were determined at 2-fold serial dilutions of Recombinant Bovine IL-10 from 10 ng/ml down to 78.12 pg/ml concentrations. To generate the straight line, The X axis represents the log10 transformation of the standard concentrations expressed as a geometric mean and the Y axis the log10 of the O.D. expressed as a geometric mean. To get the sample concentrations the values are calculated from the standard curve (in Excel) and reverse transformed (backtransformed) to give a value in ng/ml or pg/ml.

#### 2.9.7 TGFβ

To detect bovine TGF-beta in the supernatants, Bovine TGFβ ELISA kit (E11T0058, Blue Gene) which was purchased as Cow TGF-beta (TGFb) ELISA Kit (ABIN996501, Antibodies-online.com, Aachen, Germany). This commercial kit applied competitive ELISA method. Instructions were literally followed. A pre-coated 96 well micro-titre plate (supplied in the kit as 8 X 12

strips) was used. 100 ul of standards A, B, C, D, E, and F vials (supplied) was loaded in duplicates containing (0, 50, 100, 250, 500 and 1000) pg/ml bovine TGF- $\beta$  standard protein. Two more wells (blank controls) were loaded with 100 ul PBS (Ph 7.2). 100 ul of samples were dispensed with 10 ul of Balance Solution (supplied) and gently mixed well. 50 ul of enzyme conjugate (supplied) was added to all wells (except the blank control) and gently mixed. The plate was covered with cling film and incubated for 60 minutes at 37 °C. The plate was washed with was washed 5 times with 1x washing solution (supplied as 10x and diluted to 1x with deionised DW then warmed up to room temperature). After each wash, the plate was dried off by gentle tapping against tissue paper. All residual fluid was drained properly. Substrate was added as 50 ul Substrate A (supplied) along with 50 ul Substrate B (supplied) to all wells including the blank control. The plate was covered with foil and incubated for 30 minutes at 37 °C in the dark. Labtech spectrophotometer, a micro-plate reader, (LT-4000, Labtech, UK) was used to measure the absorbance at 450nm wavelength. A mean OD value for each sample was obtained and subtracted by the mean value of blank control before data interpolation (blank correction). A standard curve line was plotted to determine TGF $\beta$  in the samples (Figure 2.10). GraphPad Prism version 7 was used to interpolate the OD values into concentrations illustrated as pg/ml.



Figure 2.10: ELISA standard curve for Bovine TGF $\beta$ . Mean OD values were determined at (0, 50, 100, 250, 500 and 1000) pg/ml concentrations of bovine TGF $\beta$  standard protein. To generate the straight line, The X axis represents the log10 transformation of the standard concentrations expressed as a geometric mean and the Y axis the log10 of the O.D. expressed as a geometric mean. To get the sample concentrations the values are calculated from the standard curve (in Excel) and reverse transformed (backtransformed) to give a value in ng/ml or pg/ml.

#### 2.9.8 IFNy

Bovine IFN-γ ELISA development kit (3119-1H-6 / Mabtech, Cincinnati, USA) was used to determine IFN-γ. A sandwich ELISA method was used. ELISA plate was coated with 100 ul of PBS containing 2ug/ml monoclonal antibody MT17.1 which was used as a capture Ab. The plate was sealed and incubated in the fridge overnight. The plate was washed twice by PBS and blocked by adding 200 ul to each well blocking buffer (PBS with 0.05% Tween 20 containing 0.1% BSA) for 1 hour then washed 5 times with PBS containing Tween 20. 50 ul of samples were added in duplicates diluted 1:1 in blocking buffer along with 100 ul of standard protein (recombinant bovine IFN-γ standard). The standard protein (STD) dissolved in blocking buffer diluted as two-fold serial dilutions from 1000 pg/ml down to 7.81 pg/ml and 2 wells were loaded with PBS (blank). The plate was sealed and incubated at room temperature for 2 hours, then washed 5 times with washing buffer (PBS in 0.05% Tween 20). 100 ul of 0.25 ug/ml Biotinylated monoclonal antibody MT307 was loaded as a detection Ab in blocking buffer then the plate was

sealed and incubated at 25  $^{\circ}$ C for 1 hour then washed 5 times with washing buffer. 100 ul of SA-HRP, the enzyme, was loaded as 1:1000 dilution in blocking buffer then sealed and wrapped with foil (protected from light) and incubated at 25 °C for 60 minutes then washed with washing buffer. 100 ul of ELISA Substrate TMB Core+ (BUF062, BIO-RAD, Watford, UK) was added. The plate was not sealed and incubated in the dark at 25 °C for up to 25 minutes. Once blue colour appropriately developed, all the wells were topped up with 100 ul of 0.2M hydrochloric acid (HCl) as a stop solution. Varioskan<sup>™</sup> Flash spectral scanning multimode reader (3001-1623 - Thermofisher Scientific, Rugby, UK), a micro-plate reader, was used to measure the absorbance (Optical Density which is abbreviated as "OD" values) at 450nm and 540 nm wavelengths. OD values of the all samples were subtracted to omit background signal (corrected) as follows (mean OD samples at 450nmmean OD samples at 540nm). A standard curve line was plotted using recombinant bovine IFN-y standard (Figure 2.11). GraphPad Prism version 7 was used to interpolate the OD values into concentrations illustrated as pg/ml.



Figure 2.11: ELISA standard curve for Bovine IFN gamma. Mean OD values were determined at 2-fold serial dilutions of Recombinant Bovine IFN- $\gamma$  from 1000 pg/ml down to 7.8125 pg/ml concentrations. To generate the straight line, The X axis represents the log10 transformation of the standard concentrations expressed as a geometric mean and the Y axis the log10 of the O.D. expressed as a geometric mean. To get the sample concentrations the values are calculated from the standard curve (in Excel) and reverse transformed (backtransformed) to give a value in ng/ml or pg/ml.

# 2.10 Statistical analysis

Data were analysed by using GraphPad Prism version 7. One-way analysis of variance (ANOVA) was used to compare the groups at a level of P<0.05 or less to show significance. Microsoft Office (MS) Excel 2016 and MS Word 2016 was used to input the raw data in groups or subgroups.

**Chapter Three** 

# Generation of putative rabbit regulatory

T cells (Tregs) in culture

# 3.1 Abstract

The initial task in this chapter was to refine techniques for the generation of putative rabbit Tregs in culture for the cattle study. Some work had been done in D Haig laboratory to generate putative rabbit Tregs for another study, therefore the objective was to complete this work with a view to publication. Attempts were made to generate rabbit putative nTregs (natural regulatory T cells) in vitro by adding ConA to stimulate T cell activation, IL-2, and TGF<sup>β</sup> in mesenteric lymph node (MLN) cells seeded in IMDM culture medium. At the end of incubation, cells were labelled with fluorochromes (FACS antibodies) to highlight the phenotypic changes of the cells. CD4+CD25hiFOXP3+ cells were gated by FACS software and considered as putative rabbit nTregs. To further confirm and identify the cells of interest (nTregs), a molecular detection of transcription factor FOXP3, a Treg marker, by RT-qPCR was performed. Putative Tregs were sorted from the cultures as CD4+CD25hi cells (all FOXP3+) and tested for activity in CFSE-labelled autologous MLN cells stimulated with ConA plus PMA. The results indicated that the putative rabbit nTregs did not inhibit autologous MLN cell proliferation.

## **3.2 Introduction**

Rabbits are a good laboratory model animal for immunological research studies. Rabbits have been utilised in many research fields (Mapara *et al.*, 2012), with a specific focus on infectious diseases (Peng *et al.*, 2015) and immunity (Schnupf and Sansonetti, 2012, Guo *et al.*, 2017) (Rittershaus *et al.*, 2000, Tsenova *et al.*, 2006, Cheng *et al.*, 2012, El Sayed *et al.*, 2016). From the point of view of my main thesis work, I hoped to gain experience in generating putative rabbit Tregs in the rabbit to complete a study and transfer the knowledge to generating cattle Tregs.

From an immunological point of view, the majority of regulatory T cells (in particular nTregs) are those cells phenotypically characterised by CD4+CD25hiFOXP3+ which are presented naturally at less than 1% *in vivo* but can be induced *in vitro* at a higher purity. Both natural and induced Tregs have been extensively studied in human and mice species (D'alessio *et al.*, 2009) but there are a lack of studies in rabbits.

Phenotypically, nTregs are a subset of T cells carrying both external markers CD4 and CD25 (IL-2a receptor) and transcription factor FOXP3, which is the essential intracellular key marker for Tregs (Mottet and Golshayan, 2007, Kelley and Parker, 2010). Functionally, the role of these cells could be very important in autoimmunity and cancer through their suppressive activity (Curiel, 2007, Von Boehmer and Daniel, 2013).

TGFβ, ConA, and IL-2 were titrated and added as a combination to mesenteric lymph node (MLN) cells seeded in IMDM culture medium in an attempt to generate putative rabbit Tregs. Preliminary work on this had been done by Nevi Parameswaran. Autologous MLN cells were stimulated

with a combination of ConA and PMA and used as target cells in a putative Treg functional assay.

# 3.3 Materials and methods

All reagents were titrated as described in sections 2.4.2 and 2.4.3 in the Chapter 2. Putative rabbit Tregs were generated *in vitro* (section 2.5.1) from the MLNs (collected from 5 White New Zealand breed both sexes - two male and three female - rabbits around 12 weeks of age) and FACS labelling (section 2.5.2) was performed after 6 days of incubation, then CD4+CD25hi cells were sorted by flow cytometry (section 2.5.3) to test function in the CFSE assay (section 2.5.4 and 2.5.5). At the end of 6 days incubation, RNA was extracted from a proportion of the cultured cells (section 2.6.1), and cDNA was made (section 2.6.2) for PCR (section 2.6.4) detection of FOXP3 gene expression (nTreg marker)

# Results

# **3.4 Preliminary results**

These included titrations of reagents and generation of putative rabbit Tregs and target T cells for putative Treg functional assay.

#### 3.4.1 TGF-β1

The concentration had been already titrated by Nevi Parameswaran (Haig lab postdoc) to 2 ng/ml for putative rabbit Treg generation.

#### 3.4.2 IL-2

This reagent was titrated as mentioned in the previous chapter, section 2.4.3.1. The results showed significant proliferation (5.225 X 10<sup>6</sup> MLN cells/ml) when 10<sup>6</sup> cells were treated with 50 ng/ml IL-2 for 4 days (figure 3.12) by comparison with the control group (0.75 X 10<sup>6</sup> cell/ml) at a level P<0.05. This concentration was considered for putative rabbit Treg generated from MLN cells experiments.



Figure 3.12: IL-2 titration for MLN cell proliferation. Proliferation was maximum when 50 ng/ml of IL-2 was used (P<0.05 by comparison with negative control untreated group). Cell count = cells/ml. Mean values  $\pm$  SEM for each group.

# 3.5 Con A, PHA and PMA titrations for stimulating T cells

Thirteen groups (G1-13) of 1 X 10<sup>6</sup> MLN cells/ml in IMDM medium were used for optimal stimulation of MLN cells to develop putative rabbit Tregs for Treg assay (see section 2.4.3.2). G1, G2 and G3 were treated with Con A (1  $\mu$ g/ml, 5  $\mu$ g/ml and 10  $\mu$ g/ml), while G4, G5 and G6 were treated with the same above concentrations of Con A plus 10 ng/ml PMA respectively. G7, G8 and G9 were treated with PHA (2  $\mu$ g/ml, 5  $\mu$ g/ml and 10  $\mu$ g/ml) an alternative T cell mitogen to Con-A (Barabas *et al.*, 2002), whereas G10, G11 and G12 were treated with the same concentrations of PHA plus 10 ng/ml PMA respectively. The control group contained only MLN cells without any stimulus.

The results illustrated that the fourth group (G4, 1 µg/ml Con A + 10 ng/ml PMA) showed significant stimulation (3.62 X  $10^7$  cell/ml) after 4 days incubation (figure 3.13) (P<0.05 by comparison with the unstimulated cell control group (1.595 X  $10^7$  cell/ml) when cells were counted by haemocytometer. This combination of ConA/PMA concentrations was used in Treg target cells experiments.



Figure 3.13: Con A, PHA, and PMA titration (Mean  $\pm$  SEM) for Treg assay target cell proliferation, using MLN cells. Cell count = cells/ml. G1, G2 and G3 were treated with Con A (1 µg/ml, 5 µg/ml and 10 µg/ml), while G4, G5 and G6 were treated with the same concentrations of Con A plus 10 ng/ml PMA respectively. G7, G8 and G9 were treated with PHA (2 µg/ml, 5 µg/ml and 10 µg/ml) an alternative T cell mitogen to Con-A (Barabas *et al.*, 2002), whereas G10, G11 and G12 were treated with the same concentrations of PHA plus 10 ng/ml PMA respectively. The control group contained only MLN cells without any stimulus.

The fourth group (1  $\mu$ g/ml Con A + 10 ng/ml PMA) revealed significant stimulation (3.62 X 10<sup>7</sup> cell/ml) at a significant level (P<0.05 after 4 days incubation by comparison with negative control group).

## 3.6 Phenotype analysis of MLN and cultured MLN

#### cells

Assessment of the generated putative rabbit Tregs was by flow cytometry as described in section 2.5.2. In general, putative rabbit Tregs were generated from MLN cells between 5% and 25.7% in 5 different animal MLN samples (Tr1, Tr2, Tr3, Tr4, and Tr5) on separate occasions. FACScan analysis was after 6 days in culture, as this was a consistent time to detect FOXP3 in positive samples (N Parameswaran, unpublished data). As an example, the MLN cell culture of Tr2 sample shown in figure 3.14 showed gated cells (A) that contained 87.3% CD4+ T cells (B), and 23.6% of the total cells were FOXP3 positive (C), which indicates that these latter cells might be putative rabbit Tregs. Then, 25% of the CD4+ T cells were FOXP3+ (D), and this accounts for all the FOXP3+ cells in the total MLN population. In addition, the majority of CD4+ T cells appeared to be positive for CD25 (83.8%) (E). Whereas 19.5% of CD25+ cells were FOXP3+ (F) which indicate that the overall CD4+CD25+FOXP3+ cells were about 20% of the population after 6 days of culture. All MLN cells showed less than 0.5% CD4+CD25hi FOXP3+ (nTregs) at day 0 (prior to setting up the experimental cultures). There was a total of five rabbit putative Treg cultures that have generated between 18 and 48% CD4+ CD25+ FOXP3+ cells after 6 days in culture.

After 6 days of culture in Con-A + IL-2 + TGF-beta (n=3) results showed a range of 5%-25.7% FOXP3 positive cells (single labelled) detected by specific antibody (figure 3.15).



Figure 3.14: Triple stained generated putative rabbit Treg cells in Tr2 sample at day 6. (A): gated cells, (B): CD4+ T cells, (C): FOXP3+ cells, (D): CD4+(x axis) FOXP3+ T cells, (E): CD4+(x axis) CD25+ T cells, (F): CD25+FOXP3+ T cells, (G): CD4 isotype control and (H): CD25 isotype control.



Figure 3.15: Validation FACscan for FOXP3 antibody-labelled cells in an MLN sample after 6 days incubation (Tr2 sample). (A): FSC /SSC gated cells (B): APC-labelled isotype control antibody labelled cells. (C): anti-FOXP3:APC labelled cells (25.7%).

# 3.7 Target cells for putative Treg functional assay

In this assay, MLN cells autologous with respect to the cultured putative Treg cells were stimulated by ConA and PMA and measured by CFSE as described in section 2.5.4. The Treg target MLN cells proliferated when  $5\times10^5$  MLN cells/ml labelled with 1 µL of 5 mM CellTrace Violet (CFSE) for each mL of cell suspension were stimulated with Con A (1 µg/ml) + PMA (10 ng/ml). This was to develop stimulation and culture conditions for these cells prior to their use as target cells for the cultured Treg cells in a functional assay.

The results indicated proliferation of the 5 animal MLN samples ranged between 6.2% to 35.9% after 4 days of incubation. Using Tr2 sample as an example, the analysis of Treg target cells showed proliferation in 13.1% and 35.9% of cells at 2 and 4 days post incubation respectively for the gated cells (figure 3.16). Note that we are looking for a decrease in the starting level of fluorescence (a shoulder or peak(s) of fluorescence to the left of the Day 0 peak) to indicate that daughter cells after each division contain a diminishing amount of fluorescent signal).



Figure 3.16: Treg target cell proliferation (labelled with CellTrace violet - CFSE, abbreviated as CTV). (A): Treg target cells dot plot at day 0. (B): Treg target cells dot plot at day 2. (C): Treg target cells dot plot at day 4. (D), (E) and (F): histograms derived from items (A), (B) and (C) respectively.

## 3.8 Putative rabbit Treg functional assay

This was done by sorting CD4+CD25hi cells from the cultured putative rabbit Tregs after 6 days incubation as described in section 2.5.3. These sorted cells (see figure 3.17) were mixed with Treg target cells (normal autologous MLN cells labelled with CFSE).

Unexpectedly, the results demonstrated no suppression of proliferating

MLN cells. (figure 3.18). This was repeated with the same result 5 times.



Figure 3.17: CD4+CD25hi cell sorting from putative rabbit Tregs generated in culture after 6 days incubation. (A): gated cells at day 6 post incubation (B): a quadrate gate to highlight CD4:FITC and CD25:PE double positives showing 44.6% frequency (the view before cell sorting). (C): purified sorted cells (CD4+CD25hi).



Figure 3.18: 5X10<sup>5</sup> cells/ml of Treg target cells (autologous MLN cells) labelled with CFSE. (A): Treg target cells dot plot at day 0. (B): histogram of A. (C): Treg target cells histogram at day 4. (D), (E), (F) Treg target cells mixed with 10<sup>3</sup>, 10<sup>4</sup>, and 10<sup>5</sup> cells/ml CD4+CD25hi sorted cells at day 4 respectively.

### 3.9 RT-qPCR for FOXP3 in the MLN cells

Total RNA was extracted from the five different animal MLN cells at day 0 (in duplicate) and putative rabbit Tregs generated from these at day 6 post incubation (see section 2.6.1). cDNA was made – reverse transcribed –as per Roche instructions (see section 2.6.2). All cDNA samples of generated putative rabbit Tregs were amplified in the Lightcycler 480® Roche (RT-qPCR machine) as explained in section 2.6.4, which showed significant expression of FOXP3 gene (the key marker for Tregs) at a level P<0.05 when compared with control group (figure 3.19).



cDNA of generated Treg samples

Figure 3.19: Average  $\pm$  range of FOXP3 gene expression levels in duplicate samples of each of the MLN from 5 animals on day 6 of Treg culture (Tr1-Tr5) compared to day 0 levels (control) These fold change values were determined against SDHA reference gene (optimised by Nevi Parameswaran).

#### 3.10 Discussion

In this study, rabbit mesenteric lymph node (MLN) cells were used to generate CD4+ CD25+, FOXP3+ putative rabbit nTreg in culture. The combination of Con-A (T cell mitogen), IL-2 and TGF-beta used to stimulate MLN cells was successful in developing a proportion of cultured cells (5%-27.5%) with a CD4+, CD25+ FOXP3+ phenotype representative of nTreqs (Yi et al., 2006). This finding, regarding expression of FOXP3 in the presence of these cytokines is in line with previous studies in mice and man (De La Rosa et al., 2004, Fontenot et al., 2005a, Cesana et al., 2006, Koreth *et al.*, 2011). In these species, TGFβ is required for developing and upregulating the intracellular nTreg marker FOXP3 (Nishioka et al., 2006, Bonelli et al., 2009, Kelley and Parker, 2010). Due to time constraints and the need to focus on the bovine T cell TLR expression work, this study was stopped at this stage so further characterisation of the cells was not done. The phenotype of Tregs has been recently extended in humans. CD127, which is the interleukin-7 receptor-a, was one of the external markers thought to correlate with an inhibitory role of Tregs in autoimmunity (Moradi et al., 2014, Wu et al., 2015a, Walter et al., 2016), whereas depletion of this molecule resulted in less functional Tregs in certain diseases such as diabetes and Guillain-Barré syndrome (a neuroimmunological disorder) (Marek-Trzonkowska et al., 2014, Wang et al., 2015). CD161 is expressed by natural Treg cells as well as induced Tregs (Duurland et al., 2017). CD161+Treg cells are thought to facilitate production of pro-inflammatory cytokines (Pesenacker et al., 2013).

In addition to time constraints, the challenges and limitations in this work for further phenotypic characterisation of putative rabbit Tregs were due to the scarcity of rabbit alloantigen-specific antibodies.

Several studies have confirmed the cross-reactivity of anti-FOXP3 mAb (clone FJK-16s) with many mammals (Schoenbrunn *et al.*, 2012, Groen *et al.*, 2013, Kronsteiner *et al.*, 2013). The results in this study demonstrated significant amplification of FOXP3 gene (P<0.05) by comparison with control negative groups in putative rabbit Tregs generated *in vitro*.

In this study, the putative rabbit Tregs did not show any suppressive activity on proliferating autologous MLN cells. The reasons for this are not known. In mouse and the human, the immuno-inhibitory role of Treg cells in regulating peripheral immune responses, autoimmune diseases and cancer have been studied extensively (Ducloux, 2014, Whelan et al., 2014). FOXP3 is the transcription factor expressed internally by Tregs which defines many Tregs including nTregs (Sakaguchi, 2005). Treg immunosuppressive function is mediated by several mechanisms such as: production of the immunosuppressive cytokines IL-10 and TGF-B; DC growth and functional impairment; apoptosis or cytolysis of target cells by releasing granzyme A, granzyme B and pore forming protein (perforin) (Vignali et al., 2008). Because of the scarcity of nTreqs in vivo (5% of overall T cells), it is important to grow them in culture so they can be characterised and studied further (Shevach, 2002). We found that putative rabbit Tregs can be successfully generated in culture (~25% of MLN cells after 6 days in the present experiments) as measured by flow cytometry, focussing on the putative rabbit Treg phenotype: CD4, CD25 and FOXP3 expression. This ratio is low compared to findings in human and murine

Tregs isolated from peripheral mononuclear blood cells (Wing *et al.*, 2002, Gołąb *et al.*, 2013), where frequencies of 40%- 66% could be seen.

We found that 19% of CD4+ T cells express CD25 (the alpha chain of the IL-2 receptor), but this is moderately higher than the human and murine findings where normally 10% (in humans) and 2-3% (in mice) of CD25+ can be noticed (Roncador *et al.*, 2005, Wang *et al.*, 2013). However, detection of FOXP3+ in the rabbit CD4+ T cells (representing about a quarter of MLN cells) is in line with some findings in human Tregs (Earle *et al.*, 2005, Godfrey *et al.*, 2005).

**Chapter Four** 

# **Bovine TLR expression on T cells**

# 4.1 Abstract

TLR expression has been detected by RT-qPCR in bovine macrophages, dendritic cells, epithelial cells, and neutrophils (Werling *et al.*, 2006, Russell, 2012, Russell *et al.*, 2012). The objective of this piece of work was to carry out in-depth detection of TLRs 1-10 gene expression in bovine PBMCs and the major bovine T cell subsets (CD4, CD8, and  $\gamma\delta$  T cells). PBMCs were fractionated by MACS into CD4, CD8, and  $\gamma\delta$  T cell subsets and purity determined by flow cytometry.

Results showed successful fractionation of  $\gamma \delta$  T cells, CD4+ T cells and CD8+ T cells from bovine PBMCs. Purity of these fractionated cells exceeded 90% for CD4 and  $\gamma \delta$  T cells but ranged between 83.4% - 87.1 in CD8 T cells. Reference and target genes were expressed in the majority of the PBMCs. All of TLRs 1-10 were expressed in CD4, CD8, and  $\gamma \delta$  T cell subsets to varying levels.

# 4.2 Introduction

TLRs have been described in most mammalian species including cattle (Kumar *et al.*, 2009, Seabury *et al.*, 2010, Pandey *et al.*, 2015).

In the next three chapters (4, 5, 6), expression of TLRs on PBMC and T cell subsets will be analysed by RT-qPCR (chapter 4); PBMC and T cell subsets will be stimulated with TLR agonists (PAMPs) and activation measured by CFSE proliferation functional assays (chapter 5) and cytokine production (chapter 6).

In this chapter, PBMC isolation and fractionation of T cell subsets is described and levels of TLR1-10 expression measured by RT-qPCR.

There has been no previous reports of measurement of TLR1-10 gene expression in bovine T cell subsets. PBMC cells were MACS fractionated into CD4, CD8, and  $\gamma\delta$  T cells. Purified cells were validated by flow cytometry. TLR-1-10 gene expression was then determined.

# 4.3 Materials and methods

All methods and procedures related to this chapter have been described in the general materials and methods (chapter 2). Bovine PBMCs were prepared from the blood of individual animals as described in section 2.3. CD4, CD8 and  $\gamma\delta$  T cells subsets were MACS fractionated (section 2.7.3). The purity of each subset was tested by flow cytometry (section 2.7.4). Molecular techniques involved in gene expression were described in section 2.8 and associated sub-sections.
#### Results

#### 4.4 PBMC separation

The bovine total PBMC counts and the mean PBMC count per 1 ml blood from each of the animals are shown in table 4.24. The total bovine PBMC count, (see general materials and methods section 2.3) ranged between  $1.429 \times 10^8 - 10.125 \times 10^8$  cells. A PBMC index was used to demonstrate the mean of isolated PBMCs per 1 ml blood collected from each adult cow blood sample (figure 4.20). This was done by dividing the total PBMC counts by the blood volume (ml).

Table 4.13:	Volumes of	bovine blood	showing	variable	<b>PBMC</b> counts.

Animal	Blood	Total PBMC	PBMC mean count/
identifier	volume	count	1 ml blood
	(ml)		
BOV4098	1000	10.125 X 10 <sup>8</sup>	10.125 X 10 <sup>5</sup>
Bov A (EA)*	500	1.781 X 10 <sup>8</sup>	3.562 X 10 <sup>5</sup>
Bov B (EA)	400	1.429 X 10 <sup>8</sup>	3.572 X 10 <sup>5</sup>
Bov 1 (EA)	1150	6.810 X 10 <sup>8</sup>	5.921 X 10 <sup>5</sup>
Bov 2 (EA)	1100	6.235 X 10 <sup>8</sup>	5.668 X 10 <sup>5</sup>
Bov 3 (EA)	900	5.901 X 10 <sup>8</sup>	6.557 X 10 <sup>5</sup>
Bov 4 (EA)	700	4.613 X 10 <sup>8</sup>	6.590 X 10 <sup>5</sup>
Bov 5 (EA)	700	2.968 X 10 <sup>8</sup>	4.240 X 10 <sup>5</sup>
Bov 6 (EA)	500	2.24 X 10 <sup>8</sup>	4.480 X 10 <sup>5</sup>
Bov 7 (EA)	500	4.08 X 10 <sup>8</sup>	8.160 X 10 <sup>5</sup>
Bov 8 (EA)	500	3.74 X 10 <sup>8</sup>	7.480 X 10 <sup>5</sup>
Mean			6.032 X 10 <sup>5</sup>

\* EA = Elliot Abattoir, Chesterfield, UK



Figure 4.20: PBMC mean cell count per 1 ml blood (PBMC index). \* EA= Elliot Abattoir

#### 4.5 PBMCs stained with Diff Kwick

Stained PBMC on slides (see general materials and methods section 2.7.1) were inspected under the light microscope showing predominantly lymphocytes and fewer monocytes (figure 4.21). Monocytes seemed quite scattered over the slides but recognised because of their large size and U-shape, bean-shape or occasionally irregular blue-stained nuclei that filled half the space of the cell.



Figure 4.21: PBMCs stained with Diff Kwick. Scale bar 100  $\mu$ m (20x lens/ Leica microscope). L = lymphocyte, M = monocyte.

#### 4.6 Flow cytometry

#### **4.6.1 Major PBMC components**

Bovine PBMCs were labelled with specific Abs conjugated with fluorochromes (section 2.7.4) to detect specific lymphocyte subsets. Mouse anti-bovine CD11b FITC (which labels monocytes, but also granulocytes, and a proportion of activated CD8+ T cells in PBMC), CD21 FITC (that labels B cells in PBMC), CD4 FITC (that labels CD4+ T cells), CD8 FITC (that labels CD8+ T cells) and WC1 (CC15 clone) FITC antibodies (that labels the predominant  $\gamma\delta$  receptor+ T cell population in PBMC) and isotype controls were used. The results of flow cytometry per 1 million PBMCs is shown in figures 4.22, 4.23. shows representative FACs profiles for a few PBMC samples.



Figure 4.22: FACS results for bovine PBMCs (Mean of 7 animal PBMC samples and range). \* = presence of significant differences at P<0.05 (multiple comparison between the means of each group with the mean of every other group). CD11b + monocytes only detected. CD21+ B cells only detected.



#### A) PBMCs (Bov A)

+

#### B) PBMCs (Bov B)



Figure 4.23. FACs dot plot analysis of PBMC samples from Bov A (A) and Bov B (B). (a) a gate was applied around the mononuclear fraction (low SSC) of bov A and B animal PBMC samples. (b) and (c) for (A) and (B) show the profiles for the isotype control (IgG2a and IgG2b respectively) FITC-labelled antibodies in Bov A and B respectively. A and B (d) show the proportion of cells of Bov A and Bov B respectively expressing CD11b. A and B (e) show the proportion of CD21+ B cells in Bov A and Bov B respectively. A and B (f) show the proportion of CD4+ T cells in Bov A and Bov B respectively. Figure A (g) shows the frequency of CD4+ T cells after MACS purification of Bov A cells. Figure A (h) and B (g) show the frequency of CD8+ T cells in Bov A and Bov B animal samples. Figure A (i) shows the frequency of CD8+ T cells in Bov A cells after MACS purification. Figure A (j) and (k) shows the frequency of WC1  $\gamma\delta$  T cells before and after MACS purification respectively. Figure B (h) and (i) show the frequency of WC1  $\gamma\delta$  T cells in Bov A sample before and after MACS purification respectively.

#### 4.7 PBMCs fractionated by MACS

CD4, CD8, and  $\gamma\delta$  T cells were fractionated from PBMCs by MACs (section 2.7.3). The results obtained from this method for each subset are shown in figure 4.24. The results of fractionation (out of 1 X 10<sup>8</sup> cells suspended in 10 ml counted by haemocytometer method), indicated that total fractionated  $\gamma\delta$  T cells was the dominant phenotype amongst the others (mean 4.236X10<sup>7</sup>, ranged between 3.716X10<sup>7</sup> – 4.160X10<sup>7</sup> cells/10 ml), followed by CD4 T cells (mean 3.443X10<sup>7</sup>, ranged between 3.280X10<sup>7</sup> – 3.740X10<sup>7</sup> cells/10 ml). Finally, the lowest value was with the CD8 T cell subset (mean 1.469X10<sup>7</sup>, ranged between 0.860X10<sup>5</sup> – 1.938X10<sup>7</sup> cells/10 ml).



Figure 4.234: CD4, CD8 and  $\gamma\delta$  T mean cell counts +/- sem (counted by haemocytometer) fractionated by MACS from 10<sup>8</sup> PBMC cells from 5 animals .

\* = presence of significant differences at P<0.05.

#### 4.7.1 Flow cytometry analysis of T cell subsets fractionated

#### by MACS (purity test)

The purity of fractioned CD4, CD8 and  $\gamma\delta$  T cells subsets separated by MACS method was checked by flow cytometry (see general materials and methods sections 2.7.3 and 2.7.4). The results illustrated high purity for the fractionated CD4 T cells and slightly lower purities for the CD8 and  $\gamma\delta$  T cells. The purity ranged between 90% - 96.9% in fractionated CD4 T cells, 82% - 87.1% in CD8 and 82% - 91.4% in  $\gamma\delta$  T cells (see figure 4.25).



Figure 4.245: MACS enrichment of CD4+ T cells, CD8+ T cells and WC1+  $\gamma\delta$  T cells in PBMC samples 'Bov A' and 'Bov B'. (a) gated mononuclear cell population, bov A. (b) isotype antibody-FITC control, bov A. (c) MACS enriched CD4+ T cell population, sample Bov A. (d) MACS enriched CD4+ T cell population, sample B. (e) and (f) MACS enriched CD8+ T cell population, sample A and sample B respectively. (g) and (h) MACS enriched WC1+  $\gamma\delta$  T cells in PBMC samples from Bov A' and Bov B' respectively.

#### 4.8 Bovine TLR gene expression

#### 4.8.1 Choice of reference genes for TLR expression analysis

#### by RT-qPCR

Gene expression by RT-qPCR for target genes must be done by comparing target genes together with reference genes (Pfaffl, 2001, Pfaffl and Hageleit, 2001, Bustin, 2002). Selecting the best reference gene depends on the consistency of its expression in cells with different treatments (Schmittgen and Livak, 2008, D'haene, 2013, Emam *et al.*, 2015, Svingen *et al.*, 2015).

GAPDH, RPLP0 and RPLP2 were selected as reference genes from the literature and from colleagues working with bovine cells (section 2.8.4.3). The RNA used in this experiment was extracted from 2X10<sup>7</sup> PBMCs and cDNA was synthesised accordingly (2.8.1, 2.8.2 and 2.8.3). The results showed that the best reference gene was RPLP2 which gave an excellent standard curve efficiency of 1.985 where amplification efficiency was 98.52% and the Cq value= 23.33 (figure 4.26). In addition, it gave consistent expression in all the samples analysed, whereas the other reference genes did not. In contrast, GAPDH and RPLPO were not as efficient as RPLP2, thus they were not used in further experiments. RPLP0 gave excellent standard curve efficiency of 1.987 (amplification efficiency 98.68%), but it showed only a fair Cq value= 26.98 (figure 4.27). In addition, it was not expressed consistently. Finally, GAPDH was the worst and excluded because of irregular standard curve efficiency of 1.904 (amplification efficiency 90.42%) and a very high Cq value = 33.21 as well as expression missing in some experiments (figure 4.28).





Figure 4.256: RPLP2 amplification and standard curves. (A) Amplification curve showing the Cq values starting from cycle 23 of the 10-fold serial diluted reference gene. (B) Standard curve efficiency showing the consistency of amplification through the straight line that is linking each diluted sample. Amplification efficiency =98.52%. (C) Mean Ct values  $(n=3) \pm$  SD for reference gene RPLP2 in the assayed animals.



Figure 4.267: RPLP0 amplification and standard curves. (A) Amplification curve showing the Cq values starting from cycle 26 of the 10-fold serial diluted reference gene. (B) Standard curve efficiency is showing the consistency of amplification through the straight line that is linking each diluted sample. Amplification efficiency =98.68%.



Figure 4.278: GAPDH amplification and standard curves. (A) Amplification curve showing the Cq values started from cycle 32 of the 10-fold serial diluted reference gene. (B) Standard curve efficiency showing the inconsistency of amplification through the straight line that is linking each diluted sample. Amplification efficiency =90.42%.

# 4.8.2 Gel analysis of TLRs and reference gene amplicons after expression measurement by RT-qPCR

RNA used in this experiment was extracted from 2X10<sup>7</sup> PBMCs and cDNA was synthesised accordingly (sections 2.8.1, 2.8.2 and 2.8.3). One microgram µg of cDNA was prepared from each RNA sample. After that, cDNA samples were used as templates for RT-qPCR as described in section 2.8.4.3. Three reference genes (GAPDH, RPLP0 and RPLP2) were selected to allow a comparison of the expression of 10 TLR target genes. Gel imaging (figure 4.29) was done (section 2.8.4.2) to show the molecular size of each band (ranged from ~70 bp to ~260 bp). The two terminal wells were loaded with low molecular weight DNA ladder reagent. All genes gave a clear band at the expected size, however amplification of GAPDH relatively poor.



Figure 4.289: DNA products of reference and target genes expressed in PBMCs, amplified by RT-qPCR and analysed on a 2% agarose gel. Amplicon size unit is bp (base pair); L= Low molecular weight Ladder; T1-T10= TLR1-TLR10; G= GAPDH; R0= RPLP0; R2= RPLP2 and NCT= Negative control.

#### **4.8.3 RT-qPCR results of TLR1-10 gene expression in PBMCs.**

#### 4.8.3.1 RT-qPCR results of TLR1-10 gene expression on PBMCs

RNA was extracted from PBMCs then cDNA was synthesised and run in the Lightcycler 480 (Real time PCR) machine and data were analysed as described in section 2.8 and its subsections.

The results of TLRs expressed on PBMCs were presented as log cDNA copy number per normalised reference gene (RPLP2) as described in general materials and methods (sections 2.8.4.3 and 2.8.4.4). Bovine PBMCs from seven animals showed different levels of expression of some target genes (TLR genes) without statistical significant differences between them at  $P \ge 0.05$  (figure 4.30). However, a few genes were not amplified in some blood samples. These were TLR5, TLR6 and TLR10 as illustrated in table 4.25.



Figure 4.30: TLR 1-10 gene expression by RT-qPCR from 7 animals of bovine blood presented as the Mean  $\pm$  standard error. The PBMCs expressed these TLR genes with no statistical significant differences between them, P>0.05.

TLR	BOV4098	Bov A (EA)	Bov B (EA)	Bov 1 (EA)	Bov 2 (EA)	Bov 4 (EA)	Bov 5 (EA)
TLR1	+	+	+	+	+	+	+
TLR2	+	+	+	+	+	+	+
TLR3	+	+	+	+	+	+	+
TLR4	+	+	+	+	+	+	+
TLR5	+	+	+	+	+	+	-
TLR6	+	+	+	+	+	-	-
TLR7	+	+	+	+	+	+	+
TLR8	+	+	+	+	+	+	+
TLR9	+	+	+	+	+	+	+
TLR10	+	+	+	+	-	+	-

### Table 4.14: Presence/absence of TLR target genes on PBMCs measured by RT-qPCR from7 animals.

+ = Gene amplified (presence),

- = No amplification (absence).

## 4.8.3.2 RT-qPCR results of TLR1-10 gene expression in the fractionated T cell subsets

RNA was extracted from MACS fractionated T cell subsets (CD4, CD8 and  $\gamma\delta$  T cells) then cDNA was synthesised and run in the Lightcycler 480 (Real time PCR) machine. Data were analysed as described in the previous section.

CD4 T cells revealed TLR 1-10 gene expression with no significant differences at  $P \ge 0.05$  compared to each other (figure 4.31). A few genes were not expressed in some animals. These were TLR5 and TLR6 (table 4.26).



Figure 4.291: TLR 1-10 gene expression by RT-qPCR from 6 animals of bovine blood PBMC shown as the Mean  $\pm$  standard error. Most CD4 T cells expressed genes with no statistical significant differences between them at P $\geq$ 0.05.

Table	4.15:	Presence/absence	of TLR	target	genes	on	CD4	Т	cells
measu	red by	RT-qPCR from 7 an	imals.						

TLR	BOV4098	Bov B (EA)	Bov 1 (EA)	Bov 2 (EA)	Bov 4 (EA)	Bov 5 (EA)
TLR1	+	+	+	+	+	+
TLR2	+	+	+	+	+	+
TLR3	+	+	+	+	+	+
TLR4	+	+	+	+	+	+
TLR5	-	-	+	+	+	+
TLR6	+	+	+	+	-	-
TLR7	+	+	+	+	+	+
TLR8	+	+	+	+	+	+
TLR9	+	+	+	+	+	+
TLR10	+	+	+	+	+	+

+ = Gene amplified (presence),

- = No amplification (absence).

CD8 T cells demonstrated TLR gene expression without significant differences at  $P \ge 0.05$  compared to each other (figure 4.32). A few genes were not expressed in some blood animals. These are TLR5, TLR6 and TLR10 (table 4.27).



Target genes on CD8 T cells

Figure 4.302: TLR 1-10 gene expression by RT-qPCR from 7 animals of bovine blood shown as the Mean  $\pm$  standard error. Most CD8 T cells expressed TLR genes with no statistical significant differences between them at P $\ge$ 0.05.

<b>TI D</b>	DOV/4000	Bov A	Bov B	Bov 1	Bov 2	Bov 4	Bov 5
ILK	BOV4098	(EA)	(EA)	(EA)	(EA)	(EA)	(EA)
TLR1	+	+	+	+	+	+	+
TLR2	+	+	+	+	+	+	+
TLR3	+	+	+	+	+	+	+
TLR4	+	+	+	+	+	+	+
TLR5	-	-	-	+	+	+	+
TLR6	+	+	+	+	+	-	-
TLR7	+	+	+	+	+	+	+
TLR8	+	+	+	+	+	+	+
TLR9	+	+	+	+	+	+	+

+

-

+

+

+

+

Table 4.16: Presence/absence of TLR target genes on CD8 T cellsmeasured by RT-qPCR from 7 animals.

+ = Gene amplified (presence),

+

**TLR10** 

- = No amplification (absence).

 $\gamma\delta$  T cells demonstrated higher TLR gene expression without significant differences at P $\geq$ 0.05 compared to each other (figure 4.33). Only TLR6 gene was not expressed in two blood animals of PBMC sample (table 4.28).



Target genes on gd T cells

Figure 4.313: TLR 1-10 gene expression by RT-qPCR from 7 animals of bovine blood shown as the Mean  $\pm$  standard error. Most  $\gamma\delta$  T cells expressed TLR genes with no statistical significant differences between them at P $\geq$ 0.05.

## Table 4.17: Presence/absence of TLR target genes on $\gamma\delta$ T cells measured by RT-qPCR from7 animals.

	Sera Lab	Elliot abattoir							
TLR	BOV4098	Bov A	Bov B	Bov 1	Bov 2	Bov 4	Bov 5		
	B074098	(EA)	(EA)	(EA)	(EA)	(EA)	(EA)		
TLR1	+	+	+	+	+	+	+		
TLR2	+	+	+	+	+	+	+		
TLR3	+	+	+	+	+	+	+		
TLR4	+	+	+	+	+	+	+		
TLR5	+	+	+	+	+	+	+		
TLR6	+	+	+	+	+	-	-		
TLR7	+	+	+	+	+	+	+		
TLR8	+	+	+	+	+	+	+		
TLR9	+	+	+	+	+	+	+		
TLR10	+	+	+	+	+	+	+		

+ = Gene amplified (presence),

- = No amplification (absence).

Graphs shown below (figure 4.34) are the same results of RT-qPCR but presented in the other way to show the difference between individual animals on the same graph. There are no significant differences.







0

PBMCS

co<sup>h</sup><sup>T</sup>ce<sup>115</sup>



co<sup>a</sup>rce<sup>ils</sup>

Cell subset

got cells











Figure 4.34. TLR expression in the studied cells (PBMCs, CD4 T cells, CD8 cells and gamma delta T cells) presented in groups to show the comparison. ANOVA was used for statistical analysis without any significant differences between cells at P>0.05 level. A: TLR1, B: TLR2, C: TLR3, D: TLR4, E:TLR5, F: TLR6, G: TLR7, H: TLR8, I: TLR9 and J: TLR10.

#### 4.9 Discussion

Successful isolation of PBMCs from bovine blood and fractionation of T cell subsets CD4+, CD8+ and gd T cells was achieved in this study. Furthermore, expression of TLRs 1-10 was seen in the PBMC from the different animals. PBMC would include cells other than CD4+ T cells, CD8+ T cells and gd T cells, - principally monocytes that would have contributed to the TLR expression data. The PBMC were then fractionated into T cell subsets for TLR expression analysis. Fractionation of bovine PBMCs generally is characterised by higher numbers of gamma delta T cells (a feature of ruminants) which is compatible with what we found in this study which was around 30% (Jutila et al., 1994, Herzig and Baldwin, 2009, Baldwin and Telfer, 2015). CD8 T cells are less frequent in bovine PBMCs (approximately 10-15%) compared to CD4 T cells (range roughly 20% -35%). Immunological status, age, gender and species could possibly affect these levels (Ayoub and Yang, 1996, Anderson et al., 1999, Hope et al., 2000, Sopp and Howard, 2001, Denholm et al., 2014, Kim et al., 2016). The T cell purities are adequate for CD4, gd T cells and CD8+ T cells to demonstrate TLR expression on these cells, although it would be preferable to have purities >98% in all cases to be confident that TLR expression is specific for the T cell subset under analysis. The purity of CD8 T cells was generally lower than for CD4+ T cells and  $\gamma\delta$  T cells.

PBMCs could be reduced to a very low count (less than 10%) in late pregnancy and during calving (Kimura *et al.*, 1999). However, administration of recombinant bovine granulocyte colony stimulating factor to the pregnant cows may lead to significant increase in the total count of PBMCs after birth but not during pregnancy (Harp *et al.*, 1991). PBMC

stimulated with concanavalin A isolated from Holstein and Brown Swiss cows and subjected to temperature from 39 to 43 °C showed decline in DNA synthesis of these PBMCs which reached to 22% in Brown Swiss cows and 40% Holstein cows less than normal in the two breeds (Lacetera *et al.*, 2006). Adding food supplement such as Selenium to the diet of heifers did not show any effect on the total number of PBMCs in calves (Brennan *et al.*, 2010). The effect of zinc supplementation (*in vitro*) on the cultured PBMC viability was ranged between 69% to 86% (Sheikh *et al.*, 2016).

TLR expression on innate immune cells has been extensively studied in humans and mice as sensory molecules for the detection of infection leading to induction of inflammatory responses (Skevaki *et al.*, 2015). Other studies, mainly in the human and mouse have underlined the importance of TLRs in innate immune recognition. Two subcategories of TLRs have been identified. Those that are expressed on the surface (e.g. TLR1, TLR2, TLR4, TLR5 and TLR6) and those that are intracellular (TLR3, TLR7, TLR8 and TLR9). Bacterial PAMPs can be recognised by TLR1, TLR2, TLR4, TLR5 and TLR6, whereas viral antigens are usually sensed by TLR3, TLR7, TLR8 and TLR9 (Filippi, 2015, Skevaki *et al.*, 2015), although there are overlaps.

Expression of TLRs in ruminants, and particularly bovine immune cells has been studied (Jungi *et al.*, 2011, Werling, 2012, Baldwin and Telfer, 2015). Werling and colleagues studying bovine TLRs indicated that these were very similar reagarding amino acid sequence homology and overall predicted structure to those of mice and humans (McGuire *et al.*, 2006). Werling *et al.* (2009) and Jungi *et al.*, (2011) highlighted the importance of TLRs in the bovine compared to those in other species regarding their
possible role as adjuvants for vaccines. The importance of bovine TLR in dendritic cells, and TLR comparisons in bovine species has been described (Bilgen *et al.*, 2016, Gibson *et al.*, 2016, Tombácz *et al.*, 2017, Werling *et al.*, 2017).

Peptidoglycan, double stranded RNA viruses, and LPS stimulated bovine PBMCs *in vitro* mediated by TLR2, TLR3, and TLR4 expression in the cells - mainly monocytes and lymphocytes (Doherty *et al.*, 2013, Feldman *et al.*, 2013, Singh *et al.*, 2014, Karthikeyan *et al.*, 2016, Martins *et al.*, 2016).

Molecular detection of TLR1-10 genes in two ruminant species, ovine (sheep) and bovine (cattle), PBMCs commenced in 2006 when Menzies and Ingham used RT-qPCR to scan for these TLRs in a trial to compare their structural sequences with human TLRs (Menzies and Ingham, 2006). We used the advanced model of Menzies and Ingham's method for TLR1-10 gene detection. It is a TaqMan method where primers and probes were used to ensure specific and accurate quantification.

Our findings in this study reveal TLR expression on bovine CD4 T cells. This has also been recorded under pathological conditions in cells expressed in parasitic infestation of cattle with the protozoan parasite *Neospora caninum* (Mineo *et al.*, 2010). Also, interestingly, in humans, TLR2 and TLR4 were expressed only under pathological conditions but not on cells from healthy individuals (Zastepa *et al.*, 2014, Zhang *et al.*, 2014, Borges da Silva *et al.*, 2015). The activation state of the cells in our cattle used in this study is not known, although all animals presented disease-free and healthy.

Possibly related to activation state /healthy status of our cattle, it is interesting to note that in humans and mice CD8 T cells did not express TLRs unless stimulated. Specifically, T cytotoxic cells (CD8+ T cells) responded to viral particles and tumours by expressing intracellular TLR3 and TLR7, 8 and 9 genes (Marcondes *et al.*, 2012, Melief, 2013, Mandraju *et al.*, 2014, Schölch *et al.*, 2015, Vanders *et al.*, 2015, Wu *et al.*, 2015b). In addition to intracellular TLRs, cytotoxic T cells (CD8+ T cells) have also been shown to express TLR4, which is involved in recognition of LPS in Gram negative bacteria (Komai-Koma *et al.*, 2009). Moreover, stimulation of human neonatal cytotoxic T cells with TLR agonists (Pam3Cys and flagellin) encouraged them to express TLR2 and TLR5 genes (responsible for detection of peptidoglycan and flagellins, both found in bacteria (McCarron and Reen, 2009). In summary, to date, human CD8+ T cells have the capability to synthesise extracellular TLR1, TLR2, TLR4, 5 and 6 plus intracellular TLR3 and TLR9 (Freeman *et al.*, 2013).

Finally, the MACS-separated  $\gamma\delta$  T cells showed expression of all TLR1-10 genes. This particular subset of T cells in cattle has not been studied in terms of TLR expression until now. However, human  $\gamma\delta$  T cells express intracellular TLR3 (Beetz *et al.*, 2008). In addition, human  $\gamma\delta$  T cells activated with lipid A (part of endotoxin of gram negative bacterial cell membranes) expressed TLR2 and TLR4 (Cui *et al.*, 2009). Pietschmann *et al.* (2009) observed expression of TLRs 1-9 in human  $\gamma\delta$  T cells at variable frequencies ranging from high expression of TLR1, TLR2, TLR6, and TLR7 genes, to lower expression of TLR3 and very low expression of TLR5 and TLR8.

This subset of T cells ( $\gamma\delta$ ) is of particular interest to us as it is thought to be an important regulatory T cell subset in cattle with a role in terminating immune responses (suppressive function) either instead of or in addition to CD4+CD25+FOXP3+ T cells (nTregs) (Dar *et al.*, 2014, Guzman *et al.*, 2014, Baldwin and Telfer, 2015). In conclusion, the expression of TLR mRNAs in T cell subsets in cattle has been demonstrated in this study. **Chapter Five** 

# Stimulation of PBMC and T cells with

## **TLR agonists**

#### 5.1 Abstract

In this chapter the functional status of TLR receptors on T cells was tested by stimulating them with their ligands (TLR agonists) and assessing cell proliferation (by the CFSE method) as a read-out. Two approaches were taken: firstly analysis of purified T cells (CD4+, CD8+ and  $\gamma\delta$  T cells) from each of the bovine PBMC samles and secondly to examine bovine CD4, CD8 and  $\gamma\delta$  T cell function within the unfractionated PBMC population upon stimulation with TLR agonists (i.e. analysis of each T cell subset proliferation within PBMC without prior fractionation). This was to determine if T cell TLR activation could be detected within the context of whole PBMC. This aim was achieved by using a panel of T cell subsetspecific antibodies conjugated to non-overlapping (signal) fluorophores: CD4:RPE, CD8:Alexa Fluor® 647, and WC1gd T cell:FITC.

TLR1-9 agonists (ligands) were used at 3 concentrations to stimulate bovine CFSE-labelled PBMCs and the major T cell subsets (CD4 T cells, CD8 T cells, and  $\gamma\delta$  T cells). There was then a focus on TLR3, TLR8, and TLR9 ligands to stimulate intracellular TLR3, TLR8, and TLR9 in CD4, CD8, and  $\gamma\delta$  T cells (MACS fractionated) separately. This selection prioritisation was because of the limited number of fractionated T cell subset cells (MACS method) obtained, and that intracellular TLR3, TLR3, TLR8, and TLR9 were prioritised as play a role in detecting viral particles, an interest of our research group.

Finally, the third objective was to further study whether  $\gamma\delta$  T cells had immunosuppressive activity on autologous T cells as has been described previously (Guzman *et al.*, 2014).

Regarding the two approaches to determining T cell TLR activation by measuring CFSE cell division, it was clear after several attempts with a selection of the different animal samples that the measurement of proliferation (cell division by CFSE) was giving inconsistent results. Therefore, some examples of this are given but a complete analysis of all animal samples was terminated in favour of measuring cytokine production as an indicator of T cell TLR activation by TLR agonists (Chapter 6).

In this limited study, there was a lack of CFSE-labelled PBMC cell division (proliferation) in any of the samples upon stimulation with TLR1-9 agonists compared to good proliferation in response to the T cell mitogen Con-A. Although generally inconclusive, purified CFSE-labelled CD4+ T cells revealed some proliferation when stimulated with TLR3, TLR8 and TLR9 agonists in Bov 1 sample, TLR3 and TLR8 agonists in Bov 5 sample and TLR8 agonist in Bov4 PBMCs (P<0.05). CD8+ T cells were activated /proliferated (CFSE assay) when stimulated with TLR3 and TLR9 ligands in Bov 1; TLR8 and TLR9 agonists in Bov 5 ; and TLR9 agonist in Bov 4 blood animals (P<0.05). Purified CFSE-labelled  $\gamma\delta$  T cells stimulated with TLR8 agonist in Bov 1 and TLR3 agonist in Bov A and Bov B showed P<0.05. For the T cell subset analysis within unfractionated PBMC, the CFSE-labelled PBMCs stimulated with TLR1-9 agonists gave inconclusive results for CD4, CD8, and  $\gamma\delta$  T cell proliferation.

Finally, and unexpectedly, MACS fractionated  $\gamma\delta$  T cells (both unstimulated and ConA stimulated) showed a stimulatory effect on CFSE-labelled PBMCs (both unstimulated and ConA stimulated).

#### **5.2 Introduction**

To provide a better understanding for TLR function in bovine T cell subsets, PBMC and T cells were stimulated by TLR agonists (Bhardwaj *et al.*, 2010). TLR agonists, or ligands, are molecules derived from certain microbes (infectious or non-infectious) which are responsible specifically for activation of TLR (upon exposure in the host cells), represented by TLR signalling pathways that lead to activation of the cells and the release of pro-inflammatory cytokines (e.g. IL-1 $\beta$ , IL-6, CXCL-8 and IFN $\gamma$ ) (Kawai and Akira, 2007).

One method for assessing the function of stimulated cells is the CFSE assay. Carboxyfluorescein succinimidyl ester is a DNA-binding fluorescent dye which is used to monitor the division of the cells (measured as peaks of fluorochrome signal of diminishing signal and amplitude as the cells divide) (Lyons *et al.*, 2001). This assay is commonly used and has been successfully used to monitor T cell division (Hawkins *et al.*, 2007, Quah and Parish, 2010).

The purpose of this study was to determine the proliferation (by CFSE measurement) of PBMC and purified T cell subsets upon stimulation with TLR agonists / ligands. Bovine CD4, CD8 and  $\gamma\delta$  T cell function was also to be studied within the pool of PBMCs in response to stimulation with TLR ligands, i.e. analysis of cell subset proliferation of T cell subsets in PBMC without prior fractionation. Finally, because bovine PBMCs showed an abundance of  $\gamma\delta$  T cells and based on previous publications (Guzman *et al.*, 2014, Baldwin and Telfer, 2015) who proposed that  $\gamma\delta$  T cells we

re-investigated the role of these  $\gamma\delta$  T cells (both resting and ConA activated) on CFSE-labelled PBMCs (both resting and ConA activated).

#### 5.3 Materials and methods

#### 5.3.1 Stimulation of bovine CFSE-labelled PBMCs with TLR

#### ligands

Bovine PBMCs were labelled with CellTrace<sup>™</sup> Violet Cell Proliferation Kit (C34557 - Thermofisher Scientific, Rugby, UK) as described in section 2.7.5, seeded in RPMI complete medium at 5 X 10<sup>5</sup> cells per ml in 24 well plates. CFSE-labelled cells were stimulated with 3 concentrations of TLR ligands; low (L), moderate (M) and high (H) as described in section 2.4.4, along with two control positive groups treated with 5 ug/ml Con A and 1 ug/ml anti-bovine CD3 monoclonal antibody (these has previously been tested in dose-response experiments using bovine PBMC and purified T cells by Nevi Parameswaran, a postdoctoral researcher in the Haig lab), while a fraction of cells was left untreated as a negative control. The plates were incubated for 3 days at 37°C in a humidified atmosphere of 5% CO2 in air and wrapped with foil to protect from light. After incubation, cells were fixed, stored and shipped for analysis as described in section 2.7.5.

# 5.3.2 Stimulation of bovine CFSE-labelled T cell subsets with TLR ligands

CD4, CD8 and  $\gamma\delta$  T cells were fractionated by MACS method (section 2.7.3) and labelled separately with CellTrace<sup>TM</sup> Violet Cell Proliferation Kit (C34557 - Thermofisher Scientific, Rugby, UK) as described in section 2.7.4 and seeded in RPMI complete medium at 2 X 10<sup>5</sup> cells/ml for CD4 T cells, 1 X 10<sup>5</sup> cells/ml for CD8 T cells and 3 X 10<sup>5</sup> cells/ml for  $\gamma\delta$  T cells in 24 well plates. CFSE-labelled cells were stimulated with 3 concentrations of 3 TLR ligands (3, 8, and 9), low (L), moderate (M) and high (H) as described in section 2.4.4 (with actual concentrations shown). 5 ug/ml Con A was added to the control positive group of the  $\gamma\delta$  T cells and 1 ug/ml anti-bovine CD3 monoclonal antibody was added to each of the CD4 and CD8 T cell control positive groups, while a fraction of cells was left untreated as a negative control. The plate was incubated for 3 days at 37°C in a humidified atmosphere of 5% CO2 in air and wrapped with foil to protect from light. After incubation, cells were fixed, stored and shipped for analysis as described in section 2.7.5.

# 5.3.3 Stimulation of CFSE-labelled PBMCs with TLR ligands and analysis of T cell activation to TLR ligands within the whole PBMC population.

Bovine PBMCs were labelled with CellTrace<sup>™</sup> Violet Cell Proliferation Kit (C34557 - Thermofisher Scientific, Rugby, UK) as described in section 2.7.5, seeded in RPMI complete medium at 5 X 10<sup>5</sup> cells per ml in 24 well plate. CFSE-labelled cells were stimulated with one concentration, high (H), of nine TLR ligands (1-9) as described in section 2.4.4, along with two control positive groups stimulated with 5 ug/ml Con A and 1 ug/ml antibovine CD3 monoclonal antibody, while a fraction of cells was left unstimulated as a negative control. The cells were incubated for 3 days at 37°C in a humidified atmosphere of 5% CO2 in air and wrapped with foil to protect from light. After incubation, cells were pelleted by spinning at 300 g for 5 minutes and the supernatant was stored at -20 °C for cytokine

assessment (in chapter 6). After incubation, cells were fixed, stored and shipped for analysis as described in section 2.7.5.

### 5.3.4 The effect of fractionated $\gamma\delta$ T cells on CFSE-labelled

#### bovine PBMC proliferation.

To attempt to highlight the functional role of bovine  $\gamma\delta$  T cells on suppression or not of autologous PBMC proliferation, this experiment planned to test both stimulated (Con-A) and unstimulated  $\gamma\delta$  T cells (MACS fractionated).

3.5 X  $10^4$  MACS fractionated (purified)  $\gamma\delta$  T cells were washed then mixed with 2 X  $10^5$  CFSE-labelled autologous PBMCs on day 0 in complete RPMI medium in a 24 well plate. Another 3.5 X  $10^4$  MACS fractionated  $\gamma\delta$  T cells were washed then stimulated with 5 ug/ml ConA, incubated for 2 days then washed and mixed with 2 X  $10^5$  CFSE-labelled autologous PBMCs and the mixture was harvested after 24 hours.

An additional 2 wells of CFSE-labelled PBMCs at day 0 were treated with 5 ug/ml and 25 ug/ml ConA to induce proliferation of the target cells, whereas a fraction of CFSE-labelled PBMCs was left as a control. After incubation, cells were fixed, stored and shipped for analysis as described in section 2.7.5.

# 5.3.4.2 Role of unstimulated purified $\gamma\delta$ T cells on activated bovine autologous PBMCs (stimulated by Con A)

PBMCs from the blood of two animals (Bov 6 and Bov 7) were seeded in RPMI complete medium in two separate 25 ml flasks and stimulated with 5 ug/ml Con A each then incubated for 3 days. After that, activated PBMCs

for each sample was washed to remove any remaining Con A and labelled with CFSE.

Various cell counts of MACS-fractionated  $\gamma \delta$  T cells (1 X 10<sup>5</sup>, 2 X 10<sup>5</sup>, 3 X 10<sup>5</sup> and 4 X 10<sup>5</sup> cells/ml) were mixed separately with a constant number of CFSE-labelled autologous PBMCs (5 X 10<sup>5</sup> cells /ml) at day 0 in a 24 well plate then incubated for 48 hours. A fraction of Con A activated PBMCs, which was washed, was again re-stimulated with 5 ug/ml Con A (control positive group) while a fraction of these PBMCs was left as a control negative group. The 24 well plate was wrapped with foil to protect from light and incubated for 48 hours. After incubation, cells were fixed, stored and shipped for analysis as described in section 2.7.5.

These analyses were performed at the end of the project and for some of these there was only a limited amount of TLR-ligand for the tested TLRs left in the kits. Furthermore fractionated T cell subsets were also limited in quantity for any given animal, which limited the analyses that could be done.

#### Results

# 5.4 PBMCs stimulated with TLR ligands and labelled with CFSE (proliferation assay)

Results demonstrated no dose-dependent responses of the PBMCs from bov 1, 2, 4,5, and 4098 when stimulated with a selection of the TLR1-9 ligands in comparison with control negative groups. Figure 5.35 shows the lack of PBMC response to TLRs 1-9 in bov 5 sample, using the high concentration of each of the agonists to show the CFSE plots. Figure 5.36 shows a lack of a significant response in PBMCs (bov 1 and bov 2) to a selection of TLR agonists (to TLRs 2,3, 8 and 9, which is to be the focus of the study, as explained in the introduction) used in a dose response of low, medium and high concentrations of each of the agonists. A final attempt to get a response in two other animal PBMC samples (bov 4 and 4098) plus a repeat of bovine 5 using only the high concentration of each of the agonists to TLRs 1-9 is shown in Figure 5.37. Proliferation was low and not significant ( $P \ge 0.05$ ) when PBMCs were stimulated with the TLR agonists compared to unstimulated cells, whereas increased (by comparison) proliferation was noticed with the Con A treated PBMC group (control positive). The anti-bovine CD3 monoclonal antibody did not stimulate the cells in this or any of the other experiments in this aspect of the study and the reason for this is not known. In the light of these results, further analyses of PBMC proliferation to TLR agonists was not undertaken.



Figure 5.325: CFSE-labelled PBMCs stimulated with TLR ligands showing gated proliferative cells upon stimulation (blood animal Bov 5). A1 & A2= dot plots showing non-stained cells gated at day 0. B1= a dot plot of CellTrace violet (CFSE) stained cells (abbreviated as CTV) at day 0 showing 99% of cells were labelled. B2= a histogram of B1. C= CFSE-stained cells at day 3 - control negative group (37%). D= anti-bovine CD3 monoclonal antibody treated group (control positive) at day 3 showing 47.29% proliferation. E= ConA treated group (control positive) at day 3 showing 68.42% proliferation. F, G, H, I, J, K, L, M, and N= PBMCs stimulated with TLR1, TLR2, TLR3, TLR4, TLR5, TLR6, TLR7, TLR8, and TLR9 agonists at day 3 showing 37.94%, 40.87%, 39.24%, 42.83%, 41.84%, 41.24%, 43.20%, 39.59%, and 35.17% proliferation rates respectively. The red colour indicates the peak of cells labelled at day 0 and the blue colour indicates putative PBMC division as indicated by the lower intensity of CFSE labelling. None of the TLR agonist stimulated cells showed any evidence of cell division over that of the negative control (P>0.05).



PBMCs treated with 3 concentrations of TLR agonists in 2 animals

Figure 5.336: Proliferation (as percentages CFSE+) of 2 animal CFSElabelled PBMCs stimulated with low (L), medium (M) and high (H) concentrations of TLR1-9 agonists as measured by flow cytometry. Each point represents the average of duplicates run in the assay. The reason for the limited analysis is due to the fact that these were performed at the end of the project and there was only a limited amount of TLR-ligand for the tested TLRs left in the kits



PBMCs treated with high conc of TLR agonists in 3 animals

Figure 5.347: Proliferation (percentages CFSE+) of 3 bovine CFSElabelled PBMCs stimulated with high concentrations only of TLR1-9 agonists as measured by flow cytometry. Each point represents the average of duplicates run in the assay. The reason for the limited analysis is due to the fact that these were performed at the end of the project and there was only a limited amount of TLRligand for the tested TLRs left in the kits

#### 5.5 Proliferation of T cell subsets of the PBMC

#### stimulated with TLR ligands

Bovine PBMCs were fractionated into the major T cell subsets (CD4, CD8 and  $\gamma\delta$  T cells) (section 2.7.3). Each subset was seeded in RPMI complete medium and stimulated with TLR ligands as described in section 5.3.2. Because of the limited number of fractionated T cell subset cells (MACS method), the intracellular TLR3, TLR8, and TLR9 were prioritised, which play a role in detecting viral particles, an interest of our research group.

#### 5.5.1 Results of CFSE-labelled CD4 T cells stimulated with

#### **TLR ligands**

A total of 5 bovine CD4+ T cell populations was analysed. The results showed dose-dependent proliferation of TLR3-agonist-stimulated Bov1 and Bov5 CD4 T cells compared with the control unstimulated group (figure 5.38, 5.39, 5.40). Bov 5 CD4 T cells also showed a dose-dependent response to TLR8 agonist compared to the control (Figure 5.40). None of the other bovine CD4 T cells responded to any of TLR 3, 8 or 9 (Figures 5.38, 5.39 and 5.40). Proliferation with the control positive group treated with anti-bovine CD3 monoclonal antibody was poor in comparison with the control negative group (figures 5.38, 5.39, 5.40).



Figure 5.358: CFSE-labelled CD4 T cells stimulated with TLR ligands showing gated proliferative cells upon stimulation (animal Bov1). A1= a dot plot showing CFSE-labelled cells gated at day 0. A2= a histogram of A1. B1= a dot plot of CFSE-stained cells at day 3 of control negative group. B2= a histogram of B1 showing 10.7% proliferation (gated as Prolif). C1= a dot plot of anti-bovine CD3 monoclonal antibody treated group (control positive) at day 3. C2=ahistogram of C1 showing 16.79% proliferation. D, E, and F= CD4 T cells stimulated with low, moderate, and high conc of TLR3 agonist at day 3 showing 36.17%, 38.85% and 45.23% proliferation respectively. G, H, and I = CD4 T cells stimulated with low, moderate, and high conc of TLR8 agonist at day 3 showing 16.47%, 18.66% and 25.91% proliferation frequencies respectively. J, K, and L= CD4 T cells stimulated with low, moderate, and high conc of TLR9 agonist at day 3 showing 14.10%, 18.20% and 20.95% proliferation respectively. The red colour indicates the peak of cells labelled at day 0 and the blue colour indicates putative PBMC division as indicated by the lower intensity of CFSE labelling.



CD4 T cells treated with 3 concentrations of TLR agonists in 2 animals

Figure 5.369: Proliferation (percentages) of bovine 1 and 2 CFSE-labelled CD4 T cells stimulated with 3 concentrations of TLR3, TLR8, and TLR9 agonists as measured by flow cytometry. Each point represents the average of duplicates run in the assay. The reason for the limited analysis is due to the fact that these were performed at the end of the project and



there was only a limited amount of TLR-ligand for the tested TLRs left in the kits

CD4 T cells treated with 2 concentrations of TLR agonists in 3 animals

Figure 5.40: Proliferation (percentages) of bovine 4,5 and 4098 CFSElabelled CD4 T cells stimulated with 2 concentrations of TLR3, TLR8, and TLR9 agonists as measured by flow cytometry. Each point represents the average of duplicates run in the assay. The reason for the limited analysis is due to the fact that these were performed at the end of the project and there was only a limited amount of TLR-ligand for the tested TLRs left in the kits

#### 5.5.2 Results of CFSE-labelled CD8 T cells stimulated with

#### **TLR ligands**

Results revealed in general no proliferation (CFSE assay) above unstimulated controls of TLR 3, 8 or 9 agonist-stimulated CD8 T cells from most of seven bovine animals (figures 5.41, 5.42, 5.43 and 5.44). However, bov1 CD8 T cells responded to high concentrations of TLR3 agonist and TLR9 agonist (Figure 5.41 F, L and Figure 5.42). Bov 5 CD8+ T cells responded to high concentrations of TLR8 agonist and TLR9 agonist (Figure 5.43). Once again, the anti-CD3 stimulation of the cells as a positive control was variable. In the analyses shown in figures 5.44 A and B, high and medium and high only concentrations of TLR-agonist only were used as the the high concentration of agonist in particular gave positive results in responders.



Figure 5.41: CFSE-labelled CD8 T cells stimulated with TLR ligands showing gated proliferative cells upon stimulation (blood animal Bov1). A1= a dot plot showing CFSE-labelled cells gated at day 0. A2= a histogram of A1. B1= a dot plot of CFSE-stained cells at day 3 of control negative group. B2= a histogram of B1 showing 4.98% proliferation (gated as Prolif). C1= a dot plot of anti-bovine CD3 monoclonal antibody treated group (control positive) at day 3. C2= a histogram of C1 showing 12.17% proliferation. D, E, and F= CD8 T cells stimulated with low, moderate, and high conc of TLR3 agonist at day 3 showing 5.26%, 6.70%, and 20.87% proliferation rates respectively. G, H, and I = CD8 T cells stimulated with low, moderate, and high conc of TLR8 agonist at day 3 showing 4.26%, 5.88%, and 4.19% proliferation rates respectively. J, K, and L= CD8 T cells stimulated with low, moderate, and high conc of TLR9 agonist at day 3 showing 3.54%, 3.67%, and 21.35% proliferation (CFSE+ cells) respectively. The red colour indicates the peak of cells labelled at day 0 and the blue colour indicates putative PBMC division as indicated by the lower intensity of CFSE labelling.



CD8 T cells treated with 3 concentrations of TLR agonists in 2 animals

Figure 5.42: Proliferation (percentages) of bovine CFSE-labelled CD8 T cells stimulated with high, medium and low concentrations of TLR3, TLR8, and TLR9 agonists as measured by flow cytometry. The samples were loaded in duplicates. The kit was not enough to run more samples (and that was a limitation), therefore only 2 animals were selected. It was not

possible to present the data results as Mean ± Standard error because the results were not repeated twice (only once) and that was also due to the shortness of kit materials.



CD8 T cells treated with 3 concentrations of TLR agonists in 2 animals

Figure 5.43: Proliferation (percentages) of bovine CFSE-labelled CD8 T cells stimulated with high, medium and low concentrations of TLR3, TLR8, and TLR9 agonists as measured by flow cytometry.



CD8 T cells treated with 2 concentrations of TLR agonists in 3 animals



CD8 T cells treated with high conc of TLR agonists in 2 animals

Figure 5.44. A) Proliferation (percentages) of bovine CFSE-labelled CD8 T cells stimulated with medium and high concentrations of TLR3, TLR8, and TLR9 agonists as measured by flow cytometry. B) CD8 T cell proliferation of the TLR agonists in 2 animals stimulated with high concentrations of TLR3, 8, and 9 agonists. There was only a limited amount of TLR-ligand

for the tested TLRs left in the kits. Furthermore, fractionated T cell subsets were also limited in quantity for any given animal, which limited the analyses that could be done.

#### 5.5.3 Results of CFSE-labelled $\gamma\delta$ T cells stimulated with TLR

#### ligands

Results of the TLR agonist-stimulated  $\gamma \delta$  T cells from a selection of bovines showed no proliferation above that of the control negative groups (Figures 5.45 and 5.46) with the exception of  $\gamma \delta$  T cells from bov 1 stimulated with the high concentration of TLR8 agonist (figures 5.45I and 5.46). The Con A treated group (control positive) showed the most proliferative response of the cells (figures 5.45 and 5.46).



Figure 5.375: CFSE-labelled  $\gamma\delta$  T cells stimulated with TLR ligands showing gated proliferative cells upon stimulation (animal Bov1). A1= a dot plot showing CFSE-labelled cells gated at day 0. A2= a histogram of A1. B1= a dot plot of CFSE-stained cells at day 3 of control negative group. B2= a histogram of B1 showing 6.41% proliferation (qated as Prolif). C1 = a dot plot of Con A treated group (control positive) at day 3. C2= a histogram of C1 showing 38.94% proliferation. D, E, and  $F = y\delta T$  cells stimulated with low, moderate, and high conc of TLR3 agonist at day 3 showing 3.30%, 5.12%, and 5.97% proliferation rates respectively. G, H, and I=  $\gamma\delta$  T cells stimulated with low, moderate, and high conc of TLR8 agonist at day 3 showing 4.42%, 10.45%, and 19.75% proliferation rates respectively. J, K, and L=  $\gamma\delta$  T cells stimulated with low, moderate, and high conc of TLR9 agonist at day 3 showing 4.74%, 4.16%, and 6.77% proliferation rates respectively. The green colour indicates the peak of cells labelled at day 0 and the blue/purple colour indicates putative PBMC division as indicated by the lower intensity of CFSE labelling.



gd T cells treated with 2 concentrations of TLR agonists in 3 animals



Figure 5.386: A) Proliferation (percentages) of bovine CFSE-labelled  $\gamma\delta$  T cells stimulated with medium and high concentrations of TLR3, TLR8, and TLR9 agonists as measured by flow cytometry. B) Proliferation of bovine CFSE-labelled  $\gamma\delta$  T cells stimulated with high concentrations of TLR3, TLR8, and TLR9 agonists. The samples were loaded in duplicates. The kit was not enough to run more samples (and that was a limitation), therefore only 3 animals were selected. It was not possible to present the data results as Mean  $\pm$  Standard error because the results were not repeated twice (only once) and that was also due to the shortness of kit materials.

# 5.6 Results of CFSE-labelled PBMCs stimulated with TLR ligands: T cell subset responses within the whole PBMC population

This experiment was designed to study responses to TLR ligands of subsets of T cells (CD4, CD8 and  $\gamma\delta$  T cells) within the context of unfractionated PBMC.

Bovine PBMCs were stimulated with TLR1-9 ligands and labelled with 4 fluorophores: CFSE at day 0 plus labelling major T cell subsets (CD4, CD8 and WC1  $\gamma\delta$  T cells) with mouse anti bovine CD4:RPE, mouse anti bovine CD8:Alexa Fluor® 647, mouse anti bovine WC1:FITC, at days 0 and 3 post-stimulation. Isotype controls were used as negative controls to calibrate the FACscan gates applied (mouse IgG2a negative control; RPE-mouse IgG2a negative control; Alexa Fluor® 647-mouse IgG2a negative control; FITC-IgG2a (as a control for FITC-CD4 antibody); RPE-CD8 ab, Alexa Fluor® 647  $\gamma\delta$  T cells labelled cells respectively) (sections 2.4.4 and 2.7.5). PBMCs stimulated with TLR ligands were seeded in 1 ml RPMI complete medium in 24 well plates along with two control positive groups treated with 5 ug/ml Con A and 1 ug/ml anti-bovine CD3 monoclonal antibody respectiveky, while a fraction of cells was left untreated as a negative control (see section 5.3.3).

Results for total PBMC proliferation were obtained by subtracting CFSElabelled cells at day 0 from day 3 gated cells. A gate named "Prolif", dark blue colour in the histograms, showed the total proliferation rate as a percentage (%). For major T cell subsets (CD4, CD8 and  $\gamma\delta$  T cells), CD4:PE (red colour in the histograms) was gated at day 0 and 3 then 0

subtracted from 3 to show the proliferation as a percentage (%). The same method for determining proliferative CD4 T cells was repeated with CD8:AlexaFluor<sup>®</sup> 647 (for CD8 T cells) and WC1:FITC (for  $\gamma\delta$  T cells).

PBMCs did not proliferate noticeably to TLR1-9 ligands at the high concentration in comparison with the control negative groups in any of the animal PBMCs analysed (figures 5.47 and 5.48). ConA treated cells showed the highest proliferation, while anti-CD3 mAb treated groups responded well in only Bov 5 and BOV4098 animal cells but not in the bov 4 animal (see figures 5.47 and 5.48).

For CD4:PE labelled cells; CD8:AlexaFluor® 647 labelled cells and  $\gamma\delta$  T cells (FITC labelled) no convincing CFSE responses were seen to any of the the TLR agonists (Figures 5.49, 5.50 and 5.51). Indeed, none responded convincingly to the Con A positive control group, so the assay did not work.





TLR1-9 ligands plus control positive groups (ConA and anti-CD3 mAb). A= dot plot of non-stained PBMCs at day 0. B1, B2= dot plots of CFSElabelled PBMCs (CFSE-stained) at day 0. B3, B4, B5= CD4, CD8 and  $\gamma\delta$  T cells labelled PBMCs with PE, AF\* and FITC respectively at day 0. C1, C2, C3= CD4, CD8 and  $\gamma\delta$  T cells labelled within the PBMCs with PE, AF and FITC respectively at day 3 in the control negative group. C4= total proliferation of PBMCs at day 3 in control negative group. D1, D2, D3= CD4, CD8 and  $\gamma\delta$  T cells labelled within PBMCs with PE, AF and FITC respectively at day 3 in anti-CD3 mAb treated group. D4= total proliferation of whole PBMCs at day 3 in anti-CD3 mAb treated group. E1, E2, E3= CD4, CD8 and  $\gamma\delta$  T cells labelled within PBMCs with PE, AF and FITC respectively at day 3 in ConA treated group. E4= total

F3= CD4, CD8 and  $\gamma\delta$  T cells labelled within PBMCs with PE, AF and FITC respectively at day 3 in high conc TLR1 agonist treated group. F4= total proliferation of whole PBMCs in high conc TLR1 agonist treated group. G1, G2, G3= CD4, CD8 and  $\gamma\delta$  T cells labelled within PBMCs pool with PE, AF and FITC respectively at day 3 in high conc TLR2 agonist treated group. G4= total proliferation of whole PBMCs in high conc TLR1 agonist treated group. H1, H2, H3= CD4, CD8 and  $\gamma\delta$  T cells labelled within PBMCs with PE, AF and FITC respectively at day 3 in high conc TLR3 agonist treated group. H4= total proliferation of whole PBMCs in high conc TLR3 agonist treated group. I1, I2, I3= CD4, CD8 and  $y\delta$  T cells labelled within PBMCs pool with PE, AF and FITC respectively at day 3 in high conc TLR8 agonist treated group. I4= total proliferation of whole PBMCs in high conc TLR8 agonist treated group. J1, J2, J3= CD4, CD8 and  $v\delta$  T cells labelled within PBMCs with PE, AF and FITC respectively at day 3 in high conc TLR9 agonist treated group. J4= total proliferation of PBMCs in high conc TLR9 agonist treated group. K= CD4:PE isotype control labelled cells at day 3. L= CD8:AF isotype control labelled cells at day 3.  $M = y\delta$ :FITC isotype control labelled cells at day 3. \*AF= Alexa Fluor® 647, conc= concentration, mAb= monoclonal antibody. Purple=CellTrace violet; blue=CD8 T cells; Green= Gamma delta T cells; Red=CD4 T cells.



Figure 5.408: Total proliferation (%) of CFSE-labelled PBMCs stimulated with ConA in Bov 7 and Bov 8 blood animals. The samples were loaded in duplicates. The samples left were very limited because I consumed most of them in above tests (and that was a limitation), therefore only 2 animals were selected. It was not possible to present the data results as





CD4 T cells within the pool of PBMCs treated with High conc of TLR agonists

Figure 5.419: Proliferation (%) of CD4 T cells (labelled with PE) within the pool of CFSE-labelled PBMCs stimulated with TLR1-9 ligands, anti-CD3 mAb and ConA in Bov 5 blood animal. The samples were loaded in duplicates. The samples left were very limited because I consumed most of them in above tests (and that was a limitation), therefore only 1 animal was selected. It was not possible to present the data results as Mean  $\pm$ Standard error because the results were not repeated twice (only once) and that was also due to the shortness of kit materials.



CD8 T cells within the pool of PBMCs treated with High conc of TLR agonists

Figure 5.50: Proliferation (%) of CD8 T cells (labelled with Alexa Fluor® 647) within the pool of CFSE-labelled PBMCs stimulated with TLR1-9 ligands, anti-CD3 mAb and ConA in Bov 4 and Bov 5 blood animals. The samples were loaded in duplicates. The samples left were very limited because I consumed most of them in above tests (and that was a limitation), therefore only 2 animals were selected. It was not possible to present the data results as Mean  $\pm$  Standard error because the results were not repeated twice (only once) and that was also due to the shortness of kit materials.



gd T cells within the pool of PBMCs treated with High conc of TLR agonists

Figure 5.51: Proliferation (%) of  $\gamma\delta$  T cells (labelled with FITC) within the pool of CFSE-labelled PBMCs stimulated with TLR1-9 ligands, anti-CD3 mAb and ConA in Bov 4 blood animal. The samples were loaded in duplicates. The samples left were very limited because I consumed most of them in above tests (and that was a limitation), therefore only 1 animal was selected. It was not possible to present the data results as Mean  $\pm$ Standard error because the results were not repeated twice (only once) and that was also due to the shortness of kit materials.

### 5.7 Functional role of bovine $\gamma\delta$ T cells without TLR-

#### agonist stimulation

To summarise: regardless whether  $\gamma\delta$  T cells were unstimulated or activated (by Con-A), autologous target PBMCs, either unstimulated or stimulated with Con-A showed enhanced proliferation (by CFSE assay) compared to control cells not incubated with  $\gamma\delta$  T cells.

# 5.7.1 Unstimulated and/or activated purified $\gamma\delta$ T cells have a stimulatory effect on resting bovine autologous PBMCs

#### (unstimulated)

As described in sections 2.7.3 and 5.3.4.1, 3.5 X 10<sup>4</sup> MACS fractionated  $\gamma \delta$ T cells were mixed with 2 X 10<sup>5</sup> CSFE-labelled autologous PBMCs from Bov 7 animal, while 6 X 10<sup>4</sup>  $\gamma \delta$  T cells were mixed with 3 X 10<sup>5</sup> CFSE-labelled autologous PBMCs from Bov 8 animal in complete RPMI medium. The  $\gamma \delta$  T cells were left unactivated or activated with 5 ug/ml ConA incubated for 2 days then washed and mixed with CFSE-labelled autologous PBMCs (unstimulated) in replicate wells of a 24 well plate. Control positive groups were both CFSE- labelled PBMCs treated with 5 ug/ml and 25 ug/ml ConA separately (the higher dose of Con-A to ensure maximal stimulation). Results revealed a stimulatory activity of the  $\gamma \delta$  T cells on PBMCs in both animals Bov 7 and Bov 8 after 3 days of incubation. Unstimulated purified  $\gamma \delta$  T cells induced 77.42% (Bov 7) and 74.37% (Bov 8) CFSE shift representative of cell division compared with the control negative groups 7.36% and 17.43% respectively (figures 5.52, 5.53 and 5.54). Whereas,

ConA activated  $\gamma\delta$  T cells also showed a stimulatory effect on PBMCs

60.89% and 56.63% in Bov7 and Bov 8 animals respectively, but it was lower than the effect of unstimulated  $\gamma\delta$  T cells (figure 5.54).



Figure 5.52: CFSE-labelled PBMCs from Bov 7 mixed with both unstimulated and activated  $\gamma\delta$  T cells separately. A1= dot plot of CFSEstained PBMCs at day 0. B1= dot plots of CFSE-labelled PBMCs in the control negative group at day 3. C1= dot plot of control positive group treated with 5 ug/ml ConA at day 3. D1= dot plot of control positive group treated with 25 ug/ml ConA at day 3. E1= 3.5 X 10<sup>4</sup> unstimulated purified  $\gamma\delta$  T cells mixed with 2 X 10<sup>5</sup> CFSE-labelled PBMCs at day 3. F1= 6 X 10<sup>4</sup> purified activated  $\gamma\delta$  T cells (stimulated with 5 ug/ml ConA for 2 days) mixed with 3 X 10<sup>5</sup> CFSE-labelled PBMCs at day 3. A2, B2, C2, D2, E2, and F2= histograms of A1, B1, C1, D1, E1 and F1. Green indicates the peak of cells initially labelled on day blue/purple, the CFSE expression of progeny cells having divided by day 3.


Figure 5.53: CFSE-labelled PBMCs from Bov 8 mixed with both unstimulated and activated  $\gamma\delta$  T cells separately. A1= dot plot of CFSEstained PBMCs at day 0. B1= dot plots of CFSE-labelled PBMCs in the control negative group at day 3. C1= dot plot of control positive group treated with 5 ug/ml ConA at day 3. D1= dot plot of control positive group treated with 25 ug/ml ConA at day 3. E1= 3.5 X 10<sup>4</sup> unstimulated purified  $\gamma\delta$  T cells mixed with 2 X 10<sup>5</sup> CFSE-labelled PBMCs at day 3. F1= 6 X 10<sup>4</sup> purified activated  $\gamma\delta$  T cells (stimulated with 5 ug/ml ConA for 2 days) mixed with 3 X 10<sup>5</sup> CFSE-labelled PBMCs at day 3. A2, B2, C2, D2, E2, and F2= histograms of A1, B1, C1, D1, E1 and F1. Green indicates the peak of cells initially labelled on day blue/purple, the CFSE expression of progeny cells having divided by day 3.



Figure 5.54: Proliferation (CFSE diminuition peak percentages) of CFSElabelled PBMCs mixed with both unstimulated and activated  $\gamma\delta$  T cells separately in Bov 7 and Bov 8 showing the stimulatory role of  $\gamma\delta$  T cells. Unstimulated  $\gamma\delta$  T cells are potent stimulants for PBMCs (77.42% and 74.37% for bov 7 and bov 8 respectively) in comparison with control negative groups: (7.36% and 17.43%). The samples were loaded in duplicates. The samples left were very limited because I consumed most of them in above tests (and that was a limitation), therefore only 2 animals were selected. It was not possible to present the data results as Mean ± Standard error because the results were not repeated twice (only once) and that was also due to the shortness of kit materials.

#### 5.7.2 Purified unstimulated $\gamma \delta$ T cells have stimulatory

#### effect on activated bovine autologous PBMCs (stimulated)

As described in section 5.3.4.2 in this chapter, a total of  $1 \times 10^5$ ,  $2 \times 10^5$ ,  $3 \times 10^5$  and  $4 \times 10^5$  cells/ml  $\gamma \delta$  T cells were mixed separately with  $5 \times 10^5$  cells /ml CFSE-labelled autologous PBMCs from 2 animals (Bov 6 and Bov 7, which were previously activated with 5 ug/ml ConA in 25 ml flasks for 3 days and washed) in wells of a 24 well plate then incubated for 48 hours. Control positive group was re-stimulated with 5 ug/ml Con A.

Because this experiment was done in 2 stages and lasted for 5 days (PBMCs were stimulated for 3 days with ConA then washed and mixed with  $\gamma\delta$  T cells for 2 days), day 3 and day 5 were used to describe the duration of stimulation after the initial step of stimulation.

Results demonstrated a dose-dependent stimulatory role of  $\gamma\delta$  T cells on Con A-activated PBMCs. This was represented by elevated CFSE peaks of proliferation (lesser fluorescence than initially labelled cells) that ranged between 58.47% to 83.35% and 68.63% to 80.88% in the mixtures of (1X10<sup>5</sup>  $\gamma\delta$  T cells + 5X10<sup>5</sup> PBMCs) and (4X10<sup>5</sup>  $\gamma\delta$  T cells + 5X10<sup>5</sup> PBMCs) compared with control negative groups (13.86%) and (56.62%) in Bov 6 and Bov 7 animals respectively (see figures 5.55, 5.56, and 5.57).



and 5X10<sup>3</sup> PBMCs at day 5. E1= dot plot of a mixture of 2X10<sup>3</sup>  $\gamma\delta$  T cells and 5X10<sup>5</sup> PBMCs at day 5. F1= dot plot of a mixture of 3X10<sup>5</sup>  $\gamma\delta$  T cells and 5X10<sup>5</sup> PBMCs at day 5. G1= dot plot of a mixture of 4X10<sup>5</sup>  $\gamma\delta$  T cells and 5X10<sup>5</sup> PBMCs at day 5. A2, B2, C2, D2, E2, F2, and G2 = histograms of A1, B1, C1, D1, E1, F1, and G1. Green indicates the peak of cells initially labelled on day; blue/purple, the CFSE expression of progeny cells having divided by day 3.



cells having divided by day 3.

histograms of A1, B1, C1, D1, E1, F1, and G1. Green indicates the peak of cells initially labelled on day; blue/purple, the CFSE expression of progeny



ConA activated CFSE-labelled PBMCs mixed with unstimulated gd T cells

Figure 5.437: Proliferation (percentages) of CFSE-labelled PBMCs (5 ug/ml ConA activated for 3 days) mixed with unstimulated purified  $\gamma\delta$  T cells (for two days of the assay) separately in Bov 6 and Bov 7 blood animals showing the stimulatory role of  $\gamma\delta$  T cells. Proliferative PBMCs were 83.35% and 80.88% in the mixture of  $4X10^5 \gamma \delta T$  cells and  $5X10^5$ PBMCs in bov 6 and bov 7 respectively in comparison with control negative groups (13.86% and 56.62% respectively. The samples were loaded in duplicates. The samples left were very limited because I consumed most of them in above tests (and that was a limitation), therefore only 2 animals were selected. It was not possible to present the data results as Mean  $\pm$  Standard error because the results were not repeated twice (only once) and that was also due to the shortness of kit materials.

#### 5.8 Discussion

The functional assessment of PBMCs, MACS fractionated CD4, CD8, and  $\gamma\delta$  T cells upon stimulation with TLR 1-9 agonists by CFSE assay (a cell proliferation read-out) was overall unsatisfactory. PBMCs did not proliferate significantly in response to TLR-agonist stimulation above unstimulated control levels. This was in spite of the Con-A positive control of stimulated cells that gave a good response by comparison with the control negative group. ConA provides a potent mitogenic activity for T cells through its binding to TCR receptor glycoproteins (Balch *et al.*, 1984, Luzyanina *et al.*, 2007, Quah and Parish, 2012). On the other hand, the anti-CD3 positive stimulation control gave mixed results. Both Con-A and anti-CD3 antibody had been previously tested by dose response on cattle PBMC and T cells and the concentrations used were optimal for 3-7 day cultures.

Bovine CD4+ T cells (5 animals studied) generally did not divide/ proliferate (CFSE assay) in reponse to any of the TLR-agonists, with the exception of bov 1 CD4 T cells that responded to the TLR3 agonist and bov 5 cells that responded to both the TLR-3 agonist and the TLR-8 agonist. Similarly for CD8+ T cells, where only those (seven animal cell preparations were studied) from bov1 responded to TLR3-agonist and TLR-9 agonist, and from bov 5 that responded to TLR8-agonist and TLR9 agonist. For  $\gamma\delta$  T cells, only those from bov1 responded to TLR8 agonist. As some stimulation was seen in a few bovine purified T cells, it was decided to try another assay where T cell subsets within PBMCs from the different animals were labelled differentially with fluorescinated antibodies and then looking for the effect of PBMC proliferation of the different cell

types. This avoided isolating purified populations of T cell subsets. This was unsuccessful. The difficulties encountered with the proliferation assays for measuring cell responses to TLR-agonists led us to the conclusion that troubleshooting further would take up limited time with no guarantee of success, and it was decided to proceed to measuring cytokine expression as a readout for TLR-agonist activation of the PBMC and T cells within (Chapter 6). This means that the results of this chapter are preliminary in nature and further work is required.

The reasons for the lack of success are not known, but could include: (1) the assays were not optimised. The TLR-ligands have been shown to stimulate bovine cells at the concentrations used (Werling *et al.*, 2004, Mookherjee *et al.*, 2006, Lee *et al.*, 2008, Arsenault *et al.*, 2009, Koets *et al.*, 2010, Larsson, 2010, Nelson, 2010, Chu *et al.*, 2011, Olivia, 2012, Porcherie *et al.*, 2012, Feldman *et al.*, 2013, Price et al., 2013, Berghuis *et al.*, 2014, Dhanasekaran *et al.*, 2014, Sei *et al.*, 2014, van der Vlugt *et al.*, 2014) but the concentrations used may not have been appropriate for stimulating T cells. It is interesting that where TLR-ligands stimulated a response it was usually the medium and high concentrations that were active. Extending the concentration up may reveal better reactivity.

(2) Different animals may have been immunologically compromised such that their cells were refractile to TLR-agonist stimulation. The animals used were healthy and clinically-disease-free when they presented prior to euthanasia, but as these are outbred animals in a microbial rich environment there may have been sublimal effects. Short of performing these experiments in specific pathogen free animals, it is difficult to get round this. This is discussed further in the general discussion. Interestingly

bov 1 and bov 5 animal T cells responded to some TLR-agonists differentially (TLR 3 agonist, TLR-8 agonist and TLR 9 agonist), whereas other animal T cells did not respond to the TLR-agonists. This is discussed further in the general discussion.

Finally,  $\gamma \delta$  T cells (both unstimulated and ConA activated) showed stimulatory but not inhibitory activity for autologous bovine CFSE-labelled PBMCs (unstimulated or ConA stimulated). This is different from other studies, for example (Hoek et al., 2009), the first authors who claimed an immunosuppressive role of  $\gamma\delta$  T cells, and evidenced *in vitro* functionallyimmunosuppressive bovine IL-10 secreting  $\gamma\delta$  T cells but not CD4+CD25hi FOXP3+ T cells (conventionally known as nTregs). They speculated the presence of biological variation between bovine and non-ruminant species with regards to this particular subpopulation of T cells ( $\gamma\delta$  T cells) where at least a proportion of which expressed IL-10 (the immunosuppressive cytokine). Guzman *et al.*, (2014) has also demonstrated that bovine  $\gamma\delta$  T cells were an immunosuppressive population, more dominant than CD4+CD25hi FOXP3+ cells (nTreg) and also detected IL-10 and TGF<sup>β</sup> cytokines (known to have immune-inhibitory role on other immune cells) in the supernatant of co-culture of  $\gamma\delta$  and autologous CFSE-labelled mononuclear cells which showed significant regression in their number upon incubation ex vivo in culture. The reasons why  $\gamma\delta$  T cells were stimulatory for autologous PBMC is not known and requires further study.  $\gamma\delta$  T cells in cattle consist of several subtypes.

The clone WC1 gamma delta T cells is most likely existed not only in bovine family, but it also discovered in pigs, goats, sheep, mice, horses as well as humans (Wijngaard *et al.*, 1992). WC1  $\gamma\delta$  T cells are subdivided in cattle

into 3 further clones which are WC1.1, WC1.2, and WC1.3 (Rogers *et al.*, 2005, Rogers *et al.*, 2006). In general, bovine WC1  $\gamma\delta$  T cells are phenotypically divided into 2 main categories which are: WC1–CD2+CD3+  $\gamma\delta$  T cells that located in the red pulp of the spleen and the intestinal canal, whilst the second phenotype WC1+CD2–CD3+  $\gamma\delta$  T cells that existed mainly in the peripheral blood (Baldwin *et al.*, 2019).

Additional markers have been noted on human, murine and bovine  $\gamma\delta$  T cells which included CD2, CD4, CD5, CD6, and CD8 but functionally these markers are not known to play a role in these cells, or these markers are less beneficial to these  $\gamma\delta$  T cells (Machugh *et al.*, 1997, Hayday, 2000).

The function of  $\gamma\delta$  T cells can be basically explained by their lack to CD4 or CD8 (CD4-CD8- T cells), therefore these cells counterpart the invasive microbes without conjugated with either MHC I or MHC II as described in Chapter one. The second function is that  $\gamma\delta$  T cells can recognise lipid antigens and phosphorylated microbial metabolites (Baldwin *et al.*, 2019). Third function is that these cells produce mediators (perforin and granzyme) involved in cellular cytotoxicity such as in TB infection (Dieli *et al.*, 2000, Dieli *et al.*, 2001). The last function is that cytokine production where  $\gamma\delta$  T cells yield IFN- $\gamma$  and TNF- $\alpha$ , macrophage-colony stimulating factor, IL-17, and IL-21, IL-4, IL-10, IL-11, keratinocyte growth factor or connective tissue growth factor (Wesch *et al.*, 2001, O'Brien *et al.*, 2007, Beetz *et al.*, 2008).

In this study we used the WC1  $\gamma\delta$  T cell antibody that recognises the majority of bovine  $\gamma\delta$  T cells and is excellent in methods to bind to and hence purify  $\gamma\delta$  T cells from the tissues.

Although human  $\gamma\delta$  T cells may not necessarily be the same as that of bovine counterparts (Baldwin and Telfer, 2015, Silva-Santos *et al.*, 2015) (neither react with MHC class I or II on APC cells), human  $\gamma\delta$  T cells have been described in the literature to possess immunosuppressive effects in host responses to cancer (Dar *et al.*, 2014) upon TLR agonist activation. In the same context, Peng *et al.* (2007) investigated the functional role of  $\gamma\delta$  T cells which demonstrated immunosuppression of T cells and dendritic cells in breast cancer *in vitro* and *in vivo*, during a response to TLR8 ligands. Finally, recent studies on human  $\gamma\delta$  T cells confirmed their defensive role against cancer and autoimmune diseases as well as in fighting some infectious diseases and play a role in clinical and medical future immunotherapy (Lawand *et al.*, 2017). **Chapter Six** 

## TLR-ligand activated PBMC and T cell

## subsets: Cytokine production

#### 6.1 Abstract

The objective was to measure cytokine responses from TLR-ligandstimulated PBMC and T cell subsets. ELISA techniques were used to measure 7 cytokines, known to be produced, differentially, in various celltypes after TLR-ligand stimulation. These were: CXCL-8, IFNa, IFNy, TNFa, TGF $\beta$ , IL-4, and IL-10. There was no significant cytokine response at the entire group level for any cytokine, but individual samples did give a significant response for some analyses. In general, the results showed variable responses that were PBMC and T cell subset dependent. TLR1-9 agonists stimulated PBMC production of CXCL-8 in most samples. Purified CD4, CD8, and  $\gamma\delta$  T cells produced CXCL-8 in response to TLR3L, TLR8L and TLR9 ligand. Type I interferon (IFNa) was not reliably detected. IFN $\gamma$ was produced in some PBMC samples to a range of TLRLs. However, purified CD4+ T cells, CD8+ T cells and  $\gamma\delta$  T cells did not produce any significant quantities of IFN $\gamma$  in any sample to the restricted range of TLR3L, TLR8L and TLR9Ls.

TNF- $\alpha$  production was not significantly produced (compared to controls) in PBMC or T cell subsets (except one PBMC sample stimulated with a low concentration of TLR3 agonist). TGF $\beta$  also was not significantly produced after TLR-ligand stimulation of PBMC or associated T cell subsets.

IL-10 was increased in some TLR-ligand stimulated PBMCs and some CD4+ T cells and  $\gamma\delta T$  cells (in response to TLR3, TLR8, and TLR9 ligands) IL-4 was produced in some PBMC samples and in CD4 T cells predominantly within the T cell subsets (to TLR3L, TLR8L and TLR9L).

#### 6.2 Introduction

Bovine TLR expression was detected in PBMCs and T cells (Chapter 4) and a CFSE proliferation assay used as a readout for activation of the receptors by TLR-ligand (Chapter 5). As not all activated cells necessarily divide or proliferate, it is important to measure some other indicators of cell activation. For TLR activation, there are well-described signalling pathways described (Kabelitz, 2007). The outcome from this leads to production of pro-inflammatory cytokines in antigen-presenting cells, epithelial and endothelial cells and lymphocytes, but with the likelihood of some cellspecific cytokine production (e.g. IL-4 and IFNy in T cells) (Reynolds and Dong, 2013). The cytokines produced include IL-1β, IL-6, IL-12, CXCL-8, TNFa, type I interferons (IFNa and  $\beta$ ), IFNy, G-CSF and GM-CSF (Takeda and Akira, 2004, Kawai and Akira, 2006, Kawai and Akira, 2007, Kawasaki and Kawai, 2014). Upon stimulation with TLR agonists, T lymphocytes responded to external stimuli and as a result some further sets of cytokines have been noted such as IL-4 (van Panhuys et al., 2008), TGFB (Takebayashi et al., 2009), and IL-10 (Saraiva and O'Gara, 2010). Detailed information about each cytokine has been mentioned in chapter one (introduction), and this information included the cells that produce each cytokine, its function and the target cells.

Previous studies on bovine monocytes and macrophages stimulated with TLR ligands *in vitro* revealed secretion of abundant amounts of CXCL-8 (Russell, 2012, Russell *et al.*, 2012, Smith, 2012). We will use ELISA to measure bovine CXCL-8, IFNa, IFN $\gamma$ , TNFa, TGF $\beta$ , IL-4, and IL-10 in the supernatants of PBMCs, CD4, CD8, and  $\gamma\delta$  T cells co-cultured with TLR agonists.

## 6.3 Materials and methods

BMCs and major T cell fractions (CD4, CD8, and  $\gamma \delta$  T cells) were stimulated with TLR ligands in the presence/absence of CFSE stain as described in sections 2.4.4 and 2.7.5 (for FACS protocols). Seven cytokines were measured in response to stimulation with TLR ligands by ELISA (see section 2.9 and its branches 2.9.1 to 2.9.8), which were specific for:IL-8 (CXCL-8), IFNa, TNF-a, IL-4, IL-10, TGF $\beta$ , and IFN $\gamma$ .

The pro-inflammatory chemokine (CXCL-8) secretion from the stimulated cells was considered the principal initial readout for TLR signalling pathways because many of the TLR+ cells have been shown to produce this (Kabelitz, 2007, Brasier, 2010, Tornatore *et al.*, 2012). Other above listed cytokines also have been selected based on the recommendations collected from the literature.

Because the T cell subsets were shown to express TLR3, TLR8, and TLR9 and there is a limited supply of T cell subsets from any PBMC for testing, the focus was on these.

#### Results

# 6.4 CXCL-8 production from TLR-ligand stimulated PBMCs and T cell subsets.

#### 6.4.1 CXCL-8 results for PBMCs stimulated with TLR ligands

Figure 6.58 shows the mean of CXCL-8 (IL-8) responses in the supernatants of the seven PBMC samples studied and stimulated with TLRligands in a dose response. The comparison between PBMC TLR ligandstimulated groups and unstimulated controls (Mean  $\pm$  Standard error) showed no significant differences at the group level, but there were individual animal samples that exhibited a significant response and these are highlighted below. The PBMCs secreted IL-8 in response to high concentrations of TLR1 ligand in Bov 2 animal and also medium concentrations in Bov B and Bov4098 animals by comparison with the control negative group (P=0.041, P=0.0477 and P= 0.0473 respectively Figure 6.58a). In Figure 6.58b, responses to TLR 2 agonists at high concentration in Bov 1 and to low concentrations in Bov 2 and Bov 4098 animals were observed (P=0.03755, P=0.0398 and P=0.0406). Figure 6.58c showed a significant increase in CXCL-8 concentrations in response to high concentrations of TLR3 agonist in Bov 2 and Bov 4098 animals (P=0.00238 and P=0.00287) plus another significant increase in CXCL-8 in response to medium concentrations of TLR3 agonist in Bov 1 and Bov 4098 (P=0.0312 and P=0.0348). A significant increase in CXCL-8 was recorded in response to TLR4 ligand at high concentration in 3 animals

(Bov 1, Bov2, Bov 4098) by comparison with control negative group (figure 6.58d, P=0.0363, P=0.0329 and P=0.0373).In Figure 6.58e, high concentrations of TLR5 ligands gave significant increases of CXCL-8 in 3 animals (Bov 1, Bov 2 and Bov4098 animals) compared with the control group (P=0.0421, P=0.0407 and P=0.0441). Figure 6.58f shows a marked secretion of CXCL-8 in response to TLR6 ligands at high and low concentrations in 3 animals (Bov1, Bov2, and Bov 4098) by comparison with the control group (P=0.0414, P=0.0411 and P=0.0416). PBMCs in Figure 6.58g responded significantly by comparison with the control group to TLR7 agonists at high concentrations in 5 animals (Bov1, Bov2, Bov4, Bov5 and Bov4098) (P=0.0333, P=0.0302, P=0.0275, P=0.0228 and P=0.0341 respectively) and also when stimulated with low concentrations in 3 animals (Bov1, Bov2 and Bov4098) (P=0.0401, P=0.0442 and P=0.0403). In figure 6.58h, 3 animals (Bov1, Bov2, and Bov4098) produced CXCL-8 at significant levels upon stimulation with high concentrations of TLR8 agonists (P=0.0418, P=0.0433 and P=0.0439 respectively) when compared to control unstimulated group). Finally, high concentrations of TLR9 agonists stimulated a significant increase in CXCL-8 in 5 animals (Bov1, Bov2, Bov4, Bov5 and Bov4098) as shown in figure 6.58i (P=0.0407, P=0.0442, P=0.0488, P=0.0492 and P=0.0405 respectively) compared to unstimulated control group).



#### a) PBMCs: CXCL8 (IL-8) TLR1-ligand



b) PBMCs: CXCL8 (IL-8) TLR2-ligand



TLR-ligand concentration/ stimulus



#### c) PBMCs: CXCL8 (IL-8) TLR3-ligand



d) PBMCs: CXCL8 (IL-8) TLR4-ligand



TLR-ligand concentration/ stimulus



#### e) PBMCs: CXCL8 (IL-8) TLR5-ligand

TLR-ligand concentration/ stimulus

f) PBMCs: CXCL8 (IL-8) TLR6-ligand



TLR-ligand concentration/ stimulus



TLR-ligand concentration/stimulus

h) PBMCs: CXCL8 (IL-8) TLR8-ligand



TLR-ligand concentration/ stimulus



i) PBMCs: CXCL8 (IL-8) TLR9-ligand

TLR-ligand concentration/stimulus

Figure 6.448: CXCL-8 concentrations measured in the supernatant of PBMCs cultured in response to TLR1-9 agonists in 7 animals. a) TLR1 ligand b) TLR2 ligand c) TLR3 ligand d) TLR4 ligand e) TLR5 ligand f) TLR6 ligand g) TLR7 ligand h) TLR8 ligand i) TLR9 ligand TLR ligand concentrations are shown as high (H), medium (M) and low (L), where individual TLR ligand concentration values have been listed previously (chapter 2 paragraph 2.4.4).

# 6.4.2 CXCL-8 results of T cell subsets: CD4 T cells stimulated with TLR ligands

The comparison between CD4 T cell TLR ligand-stimulated groups and unstimulated controls (Mean ± Standard error) showed no significant differences at the group level, but there were individual animal samples that exhibited a significant response and these are highlighted below. Samples of CD4 T cells from 4 animals in total were stimulated with TLR3-, TLR8-, and TLR9 agonists. It was not possible to analyse all blood sample T cell subsets and the full inventory of TLR-ligands, so these were chosen (see introduction section 6.2 for reasons). The results are shown in Figure 6.59a. CD4+ T cells of Bov 1 sample showed significant quantities (P=0.0409 compared to unstimulated control) of CXCL8 (IL-8) when medium and high concentrations of TLR3 agonist were used. Also, CXCL8 was secreted significantly (P=0.0483 compared to unstimulated control) when CD4 T cells were stimulated with a medium concentration of TLR3 ligand in Bov 4 animal. In figure 6.59b, high and medium concentrations of TLR 8 agonists induced a significant increase of CXCL8 in Bov 1 while only the medium concentration of TLR8 ligand stimulated a significant increase of CXCL8 in animal Bov 4 (P=0.0246 compared to unstimulated control). Finally, figure 6.59c shows a significant increase in the levels of CXCL8 in response to stimulation with high concentrations of TLR9 ligands in 2 animals (Bov1 and Bov2, P=0.0452 and P=0.0487 compared to unstimulated control), whereas 3 animals (Bov1, Bov2 and Bov5) yielded significant amounts of CXCL8 when stimulated with medium concentrations of TLR9 ligands (P=0.044, P=0.0461 and P=0.0455 respectively).



TLR-ligand concentration/ stimulus

b) CD4 T cells: CXCL8 (IL-8) TLR8-ligand



**TLR-ligand concentration/ stimulus** 



TLR-ligand concentration/ stimulus

Figure 6.459: CXCL-8 concentrations measured in the supernatant of CD4 T cells cultured in response to TLR3, TLR8, and TLR9 agonists in 4 animals. a) 2 animals yielded significant amounts of IL-8 in response to 3 concentrations of TLR3 ligands. b) 2 animals yielded significant amounts of IL-8 in response to 3 concentrations of TLR8 ligands. c) 3 animals yielded significant amounts of IL-8 in response to 3 concentrations of TLR9 ligands. H, M, L explained as before (chapter 2, paragraph 2.4.4).

#### 6.4.3 CXCL8 results of CD8 T cells stimulated with TLR ligands

CD8 T cells from the PBMC of 4 animals were stimulated with TLR3, TLR8, and TLR9 agonists.

Figure 6.60a shows significant responses (P=0.0316 and P=0.0384) of CXCL8 in the CD8 T cells of Bov 1 and Bov 2 stimulated with high concentrations of TLR3 ligands by comparison with the unstimulated control. Figure 6.60b shows significant responses (P=0.0251 and P=0.0272) of CXCL8 in the CD8 T cells of Bov 1 and Bov 2 stimulated with high concentrations of TLR8 ligands by comparison with the unstimulated

control. Also, significant responses were seen in the Bov1, Bov2, and Bov5 T cells when stimulated with medium concentrations of TLR8 ligands (P=0.0372, P=0.0353 and P=0.0333 respectively). Finally, figure 6.60c shows significant responses (P=0.0386 and P=0.0403) of CXCL8 in the CD8 T cells of Bov 1 and Bov 2 stimulated with high concentrations of TLR9 ligands by comparison with the unstimulated control. Also, significant responses were detected in the T cells from animals Bov1, Bov2, and Bov4 when stimulated with medium concentrations of TLR9 ligands (P=0.0433, P=0.0442 and P=0.0399 respectively).



**TLR-ligand concentration/ stimulus** 

### c) CD8 T cells: CXCL8 (IL-8) TLR9-ligand



TLR-ligand concentration/ stimulus

## Figure 6.60: CXCL-8 concentrations measured in the supernatant of CD8 T cells cultured in response to TLR3, TLR8, and TLR9 agonists in 4 animals. a) TLR3 agonist. b) TLR8 agonist. c) TLR9 agonist.

#### 6.4.4 CXCL8 results of $\gamma\delta$ T cells stimulated with TLR ligands

There was no significant difference between M±SEM of  $\gamma\delta$  T cells of the treated animal samples over unstimulated controls at the group level, but there were individual samples that were significantly different from controls. Stimulation with TLR3, TLR8, and TLR9 agonists of  $\gamma\delta$  T cells isolated from the PBMC of 4 animals were analysed (Bov 1, Bov 2, Bov 4, and Bov 5).

Figures 6.61a and b show significant production (P=0.0147 and P=0.0158) of CXCL8 in the  $\gamma\delta$  T cell supernatants of Bov1 and Bov4 stimulated with high concentrations of TLR3 and TLR8 agonists. Figure 6.61c shows a significant increase of CXCL8 in the  $\gamma\delta$  T cell supernatants of Bov 1 and Bov4 stimulated with high concentrations of TLR9 ligands (P=0.0126 and

P=0.0136). In addition, CXCL8 in the  $\gamma\delta$  T cell supernatants of Bov 1 Bov 2 and Bov 4 in response to medium concentrations of TLR9 ligands were significant (P=0.0176, P=0.0159 and P=0.0211 respectively) by comparison with the unstimulated control.

## a) Gamma delta T cells: CXCL8 (IL-8) TLR3-ligand



**TLR-ligand concentration/ stimulus** 

## b) Gamma delta T cells: CXCL8 (IL-8) TLR8-ligand



**TLR-ligand concentration/ stimulus** 



#### c) Gamma delta T cells: CXCL8 (IL-8) TLR9-ligand

**TLR-ligand concentration/ stimulus** 

Figure 6.61: CXCL-8 concentrations measured in the supernatant of  $\gamma\delta$  T cells cultured in response to TLR3, TLR8, and TLR9 agonists in 4 animals. a) TLR3 agonist. b) TLR8 agonist. c) TLR9 agonists.

# 6.5 IFN-α, TNF-α, and TGF-beta in TLRL-stimulated PBMCs and T cell subsets

There was no significant difference between M±SE of TGF- $\beta$  in PBMCs or T cells of the TLRL -treated groups over unstimulated controls. However, only cells from two animals were analysed (due to the capacity of the kit and a lack of time at the end of the project to complete the analysis) (Figure 6.65). There was no significant response in PBMCs or T cells to any TLRL stimulation for IFN-alpha, or TNF-alpha at the group level compared to unstimulated controls. However individual responses were significant compared to controls. Bov2 and bov5 PBMCs stimulated with the highest concentration of TLR2L and TLR3L showed significantly enhanced IFN-

alpha compared to controls. PBMCs stimulated with TLR7L and TLR8L did not produce significant amounts of IFN-alpha compared to controls. CD8 T cells stimulated with high concentrations of TLR3 ligands yielded significant amount of IFNa in 2 animals Bov1 and Bov 2 (P=0.0425 and P= 0.0464 respectively) (Figure 6.63a), but other TLRLs were not tested. Gammadelta T cells of bovs 4 and 5 stimulated with TLR9-L also showed a significant response compared to controls. On the other hand, individual PBMCs yielded significant amounts of TNFa when stimulated with high concentrations of TLR3 ligands in 5 animals Bov1, Bov2, Bov5, Bov A and Bov4098 (P=0.0482, P=0.0488, P=0.0482, P=0.0466, and P=0.0471) (Figure 6.64c). Finally, individually, PBMCs of Bov5 stimulated with the high concentration of TLR9 ligand gave a significant increase in TNFa (P=0.0468) compared to controls (Figure 6.64i).



b) PBMCs: IFN alpha TLR3-ligand



**TLR-ligand concentration/ stimulus** 



**TLR-ligand concentration/ stimulus** 

Figure 6.62(a-d): IFN-a concentrations measured in the supernatant of PBMCs cultured in response to TLR1-9 agonists in 7 animals.

## a) CD8 T cells: IFN alpha TLR3-ligand







c) Gamma delta T cells: IFN alpha TLR9-ligand



**TLR-ligand concentration/ stimulus** 

Figure 6.63 (a-c): IFN-a concentrations measured in the supernatant of T cell subsets cultured in response to TLR3 and TLR9 agonists from CD8 T
cells of 4 animals and  $\gamma\delta$  T cells of 5 animals. CD4 T cells produced nil concentrations.



TLR-ligand concentration/ stimulus

b) PBMCs: TNF alpha TLR2-ligand



**TLR-ligand concentration/ stimulus** 



TLR-ligand concentration/ stimulus



TLR-ligand concentration/ stimulus



TLR-ligand concentration/ stimulus

h) PBMCs: TNF alpha TLR8-ligand



TLR-ligand concentration/ stimulus



Figure 6.64 (a-i): TNF-a concentrations measured in the supernatant of PBMCs cultured in response to TLR1-9 agonists. CD4, CD8 and  $\gamma\delta$  T cells did not produce any TNF-a in the supernatant.

# i) PBMCs: TNF alpha TLR9-ligand



b) PBMCs: TGFb TLR2-ligand



TLR-ligand concentration/ stimulus



d) PBMCs: TGFb TLR4-ligand



TLR-ligand concentration/ stimulus



f) PBMCs: TGFb TLR6-ligand



TLR-ligand concentration/ stimulus



h) PBMCs: TGFb TLR8-ligand



TLR-ligand concentration/ stimulus





k) CD4 T cells: TGFb TLR8-ligand



TLR-ligand concentration/ stimulus

236



TLR-ligand concentration/ stimulus

m) CD8 T cells: TGFb TLR3-ligand



TLR-ligand concentration/ stimulus



TLR-ligand concentration/ stimulus

o) CD8 T cells: TGFb TLR9-ligand



TLR-ligand concentration/ stimulus





**TLR-ligand concentration/ stimulus** 

Figure 6.65: TGF $\beta$  concentrations measured in the supernatant of PBMCs (figures a-i) and T cell subsets (figures j-r) cultured in response to TLR agonists.

## 6.6 IL-4

## 6.6.1 IL-4 results of PBMCs and T cell subsets stimulated with

#### **TLR ligands**

The comparison between PBMC TLR ligand-stimulated groups and unstimulated controls (Mean  $\pm$  Standard error) showed no significant differences at the group level, but there were individual animal samples that exhibited a significant response and these are highlighted below. Figure 6.66a shows that the PBMC from 7 animals stimulated with TLR1 agonists, gave significant increases in IL-4 in 2 animals only (Bov1 and Bov2) at P=0.0458 and P=0.0441 by comparison with unstimulated controls. PBMCs treated with TLR2 and TLR3 agonists did not show significant increases of IL-4 (figures 6.66b and c). Low concentrations of TLR4 and TLR5 agonists stimulated a significant increase in IL-4 in 2 animals (Bov1 and Bov2) as shown in figure (6.66d and e P=0.0471 and P=0.0458). There was no enhanced production of IL-4 from cells stimulated with 3 concentrations of TLR6 agonists (figure 6.64f). Stimulation of PBMCs with high concentrations of TLR7 ligands caused a significant increase in IL-4 in 2 animals (Bov A and Bov 5, P=0.0438 and P=0.0447 figure 6.66g) by comparison with the unstimulated control. Also stimulation with low concentrations of TLR7 agonists induced a significant increase in IL-4 in 2 animals (Bov 1 and Bov2, P=0.0488 and P=0.0496 compared to unstimulated control). In figure 6.66h, stimulation with high concentrations of TLR 8 agonists induced a significant increase in IL-4 in 2 animals (Bov 5 and Bov A, P=0.0467 and P=0.0453) compared to unstimulated control) while stimulation with medium concentrations of TLR8 agonists caused a significant increase in IL-4 in 2 animals (Bov1 and Bov2 P=0.0489 and P=0.0471 vs control). Finally, figure 6.66i shows that stimulation with the high concentration of TLR9 agonists caused a significant increase in IL-4 in 3 animals (Bov 2, Bov 4 and Bov 5, P=0.0472, P=0.0459 and P=0.0483 respectively vs control).



TLR-ligand concentration/ stimulus

b) PBMCs: IL-4 TLR2-ligand



**TLR-ligand concentration/ stimulus** 





d) PBMCs: IL-4 TLR4-ligand



TLR-ligand concentration/ stimulus



**TLR-ligand concentration/ stimulus** 

f) PBMCs: IL-4 TLR6-ligand



TLR-ligand concentration/ stimulus



TLR-ligand concentration/ stimulus



Figure 6.66: IL-4 concentrations measured in the supernatant of PBMCs cultured in response to TLR1-9 agonists in 7 animals. (a) TLR1 ligand. (b) TLR2 ligand. (c) TLR3 ligand. (d) TLR4 ligand. (e) TLR5 ligand. (f) TLR6 ligand. (g) TLR7 ligand. (h) TLR8 ligand. (i) TLR9 ligand.

#### 6.6.2 IL-4 results of CD4 T cells stimulated with TLR ligands

The comparison between CD4 T cell TLR ligand-stimulated groups and unstimulated controls (Mean  $\pm$  Standard error) showed no significant differences at the group level, but there were individual animal samples that exhibited a significant response and these are highlighted below. An analysis of the T cell subsets (figure 6.67) showed that an increase of IL-4 secreted from CD4 T cells was seen when stimulated with TLR3-, TLR8and TLR9 agonists. Stimulation of CD4 T cells with low concentrations of TLR3 ligands caused significant production of IL-4 in 2 animals (Bov1 and Bov2) over control negative group (P=0.0493 and P=0.0484, Fig 6.67a). In figure 6.67b stimulation with low concentrations of TLR8 ligand induced significant production of IL-4 in 1 animal only (Bov2) over the unstimulated control (P=0.0491). In figure 6.67c stimulation with medium and high concentrations of TLR9 ligand induced significant production of IL-4 in 1 animal only (Bov1) compared to the unstimulated control (P=0.0488).





**TLR-ligand concentration/ stimulus** 

Figure 6.67: IL-4 concentrations measured in the supernatant of CD4 T cells cultured in response to TLR3, TLR8, and TLR9 agonists in 4 animals. (a) TLR3 ligand. (b) TLR8 ligand. (c) TLR9 ligand.

## 6.6.3 IL-4 results of CD8 T cells stimulated with TLR ligands

The comparison between CD8 T cell TLR ligand-stimulated groups and unstimulated controls (Mean  $\pm$  Standard error) showed no significant differences at the group level, but there were individual animal samples that exhibited a significant response and these are highlighted below. Figure 6.68 shows that for the CD8 T cell subset, no significant increases over controls of IL-4 production (P>0.05) was seen in samples stimulated with TLR3, TLR8 and TLR9 agonists by comparison with control unstimulated groups (Figures 6.68a, b and c).





**TLR-ligand concentration/ stimulus** 



**TLR-ligand concentration/ stimulus** 

Figure 6.468: IL-4 concentrations measured in the supernatant of CD8 T cells cultured in response to TLR3, TLR8, and TLR9 agonists in 4 animals. (Figures a, b and c respectively).

### 6.6.4 IL-4 results of $\gamma\delta$ T cells stimulated with TLR ligands

The comparison between  $\gamma\delta$  T cells TLR ligand-stimulated groups and unstimulated controls (Mean ± Standard error) showed no significant differences at the group level, but there were individual animal samples that exhibited a significant response and these are highlighted below. Figure 6.69a shows that  $\gamma\delta$  T cells produced IL-4 at significant levels over controls (P<0.04) in 2 samples, to high and low TLR3 agonist (bov 2) and low concentration (bov 1) of by comparison with control negative groups. In figure 6.69b, high concentrations of TLR8 ligands induced a significant increase of IL-4 production in 2 animals (Bov 1 and Bov 5) over controls (P=0.0471 and P=0.0462). Finally figure 6.69c shows no significant production of IL-4 from the stimulated cells with TLR9 agonist over controls.



# a) Gamma delta T cells: IL-4 TLR3-ligand





TLR-ligand concentration/ stimulus



# c) Gamma delta T cells: IL-4 TLR9-ligand

Figure 6.479: IL-4 concentrations measured in the supernatant of  $\gamma\delta$  T cells cultured in response to TLR3, TLR8, and TLR9 agonists in 5 animals (Figures 6.69 a, b and c respectively.

# 6.7 IL-10 results of PBMCs and T cell subsets stimulated with TLR ligands

The comparison between PBMC TLR ligand-stimulated groups and unstimulated controls (Mean  $\pm$  Standard error) showed no significant differences at the group level, but there were individual animal samples that exhibited a significant response and these are highlighted below. Figure 6.70a) shows a significant increase in IL-10 in animals stimulated with the high concentration of TLR1 ligands (Bov 2 and Bov 4098 P=0.0478 and P=0.0482 compared to unstimulated control) and significant increase in IL-10 in those stimulated with medium concentrations of TLR 1 in 2 animals (Bov1 and Bov2) over controls (P=0.0461 and P=0.0453). Figures 6.70b and c) show significant increases of IL-10 in those cells stimulated with the high concentration of TLR2 and TLR3 ligands in 2 animals (Bov2 and Bov 4098) over controls (P=0.0452 and P=0.0466). Figures 6.70d and e show significant increase of IL-10 in those stimulated with low concentrations of TLR4 ligand in 2 animals (Bov1 and Bov 2) over controls (P=0.0447 and P=0.0419). Figure 6.70e shows secretion of IL-10 from cells stimulated with low concentrations of TLR5 ligands (P=0.0422 and P=0.0438) vs unstimulated control in 2 animals (Bov1 and Bov2). Figure 6.70f shows secretion of IL-10 from cells stimulated with medium concentrations of TLR6 agonist in 2 animals (Bov1 and Bov 2) over controls (P=0.0456 and P=0.0436). In figure 6.70g, cells from 3 animals (Bov 1, Bov2, Bov A) gave significant increases in IL-10 stimulated with the high concentration of TLR7 ligand compared to unstimulated controls (P=0.0438, P=0.0377 and P=0.0426). The low concentration of TLR7

256

ligand caused a significant increase in IL-10 over controls in 2 animals (Bov1 and Bov2, P=0.0468 and P=0.0479). In figure 6.70h, only one sample from Bov 1 gave a significant IL-10 response to TLR8 ligand at medium concentration over controls (P=0.0475). Finally, figure 6.70i shows only one sample (Bov 2) responded significantly to stimulation with high concentration of TLR9 over controls (P=0.0443).



TLR-ligand concentration/ stimulus

b) PBMCs: IL-10 TLR2-ligand



TLR-ligand concentration/ stimulus



TLR-ligand concentration/ stimulus



TLR-ligand concentration/ stimulus



TLR-ligand concentration/ stimulus

h) PBMCs: IL-10 TLR8-ligand



TLR-ligand concentration/ stimulus


**TLR-ligand concentration/ stimulus** 

# Figure 6.70: IL-10 concentrations measured in the supernatant of PBMCs cultured in response to TLR1-9 agonists in 7 animals. (a) LR1 ligand. (b) TLR2 ligand. (c) TLR3 ligand. (d) TLR4 ligand. (e) TLR5 ligand. (f) TLR6 ligand. (g) TLR7 ligand. (h) TLR8 ligand. (i) TLR9 ligand.

Figure 6.71 (a) shows (that there was no significant production of IL-10 secretion by the CD4 T cells to any concentration of TLR3 agonist by comparison with the control negative group. The low concentration of TLR8 ligand significantly induced IL-10 in 1 animal (Bov1) over controls (P=0.0452). In figure 6.71c, CD4 T cells stimulated with high concentrations of TLR9 ligand showed a significant increase of IL-10 in 1 animal (Bov4) over controls (P=0.0476).

Figure 6.71d reveals that CD8 T cells stimulated with medium concentration of TLR3 ligand induced IL-10 in Bov 1 when stimulated with medium concentrations of TLR3 ligand over controls (P=0.041 compared to control). Figure 6.71e, shows no significant increase of IL-10 from CD8 T cells stimulated with TLR8 ligand at 3 concentrations.

Figure 6.71f shows the response in CD8 T cells of 2 animals (Bov1 and Bov 5) stimulated with high and medium concentrations of TLR9 ligand respectively by comparison with the control group (P=0.0477 and P=0.0433).

Figure 6.71g, shows gamma delta T cells stimulated with the high concentration of TLR3 ligand which significantly induced IL-10 in 2 animals (Bov1 and Bov2) over controls (P=0.0448 and P=0.0461). Figure 6.71h, shows a significant increase of IL-10 secreted from gamma delta T cells stimulated with medium concentrations of TLR8 ligand in 2 animals (Bov1 and Bov4) over controls (P=0.0482 and P=0.0466). Finally, figure 6.71i, shows no significant production of IL-10 from gamma delta T cells stimulated with TLR9 ligand at 3 concentrations by comparison with controls.







TLR-ligand concentration/ stimulus





#### h) Gamma delta T cells: IL-10 TLR8-ligand





Figure 6.71: IL-10 concentrations measured in the supernatant of T cell subsets cultured in response to TLR3, TLR8, and TLR9 agonists in 4 animals (for CD4, figures a-c, and CD8 T cell samples – figures d-f) and 5 animals (for  $\gamma\delta$  T cell samples – figures g-i).

# 6.8 IFNγ results of PBMCs and T cell subsets stimulated with TLR ligands

#### 6.8.1 IFNy results of PBMCs stimulated with TLR ligands

The comparison between PBMC TLR ligand-stimulated groups and unstimulated controls (Mean  $\pm$  sem) showed no significant differences at the group level, but there were individual animal samples that exhibited a significant response and these are highlighted below. Figure 6.72a shows that a significant IFNy response was detected in the PBMCs of only 1 animal (Bov 5) (P=0.0339 compared to controls) stimulated with high concentrations of TLR1 ligands and in bov A and Bov5 for medium TLR1 stimulation (P<0.04). Figure 6.72b shows a significant IFNy response in Bov 5 stimulated with either high or medium concentrations of TLR2 agonists over controls (P<0.316). In figure 6.72c and d, 2 animals yielded significant amounts of IFNy in response to simulation with high and medium concentrations of TLR3 and TLR4 agonists (Bov B and Bov 5), (P=0.0349 and P=0.0356 compared with controls). In figure 6.72e, f and g, PBMCs yielded significant amounts of IFNy in response to stimulation with high and medium concentrations of TLR5, TLR6 and TLR7 agonists in 3 animals (Bov 5, Bov A and Bov B) respectively by comparison with control negative groups (P=0.0361, P=0.0377 and P=0.0395 respectively).

Figure 6.72h, shows significant production of IFN $\gamma$  in 3 animals (Bov5, Bov A and Bov B) stimulated with high and medium concentrations of TLR8 ligands by comparison with the unstimulated control groups (P=0.0348, P=0.0366 and P=0.0371). Finally, figure 6.72i reveals a significant production of IFN $\gamma$  in the supernatant of PBMCs stimulated with high and medium concentrations of TLR9 ligands in 3 animals (Bov 5, Bov A and Bov

B). by comparison with control negative group (P=0.0327, P=0.0355 and P=0.0387 respectively).



TLR-ligand concentration/ stimulus

b) PBMCs: IFN gamma TLR2-ligand



**TLR-ligand concentration/ stimulus** 



TLR-ligand concentration/ stimulus

d) PBMCs: IFN gamma TLR4-ligand



**TLR-ligand concentration/ stimulus** 



TLR-ligand concentration/ stimulus

f) PBMCs: IFN gamma TLR6-ligand



**TLR-ligand concentration/ stimulus** 



TLR-ligand concentration/ stimulus

h) PBMCs: IFN gamma TLR8-ligand



**TLR-ligand concentration/ stimulus** 



**TLR-ligand concentration/ stimulus** 

Figure 6.48 (a-i, TLR ligands 1-9 respectively): IFN $\gamma$  concentrations measured in the supernatant of PBMCs cultured in response to TLR1-9 agonists in 7 animals.

#### 6.8.2 IFNy results of T cell subsets stimulated with TLR

#### ligands

IFNγ was not secreted by neither CD4 T cells, nor CD8 T cells stimulated with TLR3, TLR8, and TLR9 agonists, while gamma delta T cells produced some IFNγ (figure 6.73).

The comparison between  $\gamma\delta$  T cell TLR ligand-stimulated groups and unstimulated controls (Mean ± Standard error) showed no significant differences at the group level, but there were individual animal samples that exhibited a significant response and these are highlighted below.





TLR-ligand concentration/ stimulus

#### b) Gamma delta T cells: IFN gamma TLR8-ligand



**TLR-ligand concentration/ stimulus** 





Figure 6.73. Response of  $\gamma\delta$  T cells to TLR-agonists (a-c, TLRs 3, 8 and 9 agonists respectively.

#### 6.9 Discussion

In this chapter, TLR activation by TLR-ligand stimulation was assessed in PBMC and the T cell subsets (CD4, CD8,  $\gamma\delta$  T cells) by the production of a range of cytokines. Some analyses are incomplete owing to time constraints, but there are interesting results from what has been done. CXCL8 has been used most often as a read out for TLR activation in a variety of cell types (Allavena *et al.*, 2008, Fonseca *et al.*, 2011, Mills, 2011, Russell *et al.*, 2012, Lee *et al.*, 2015, Karthikeyan *et al.*, 2016, Salvi *et al.*, 2016, Tombácz *et al.*, 2017).

Secretion of CXCL-8 from bovine APCs such as monocytes, macrophages, dendritic cells and B cells upon stimulation with different microbes has previously been recorded (Widdison *et al.*, 2008, Taubert *et al.*, 2009, Metcalfe *et al.*, 2010, Düvel *et al.*, 2014).

What was immediately apparent was that only a subset of cell samples from the bovine animals gave a response in the assays. This meant that there was no significant cytokine response at the entire group level for any cytokine, but that individual samples did give a significant response for some analyses (expanded below). In particular cell samples from bovs 1,2, 4, 5 and 4098 were generally more responsive to a range of TLRLs for cytokine production that the other samples. The reasons for this are not known. However it could be that the immune status of the animal may have been responsible. All animals were checked for clinical disease and overall good health prior to entering to the abattoir, but the immune system of outbred animals in the environment are constantly being challenged and this is normal. We were unable to perform a thourogh analysis of this.

In this study CXCL8 was the cytokine amongst those tested that gave the most comprehensive results. Both PBMC and T cell subsets of a proportion of the animal samples responded to the TLR-Ls (PBMC) or a subset of them (see below) respectively to give a CXCL8 response. In this study ELISAs specific for the bovine animal were used whereas in the above references, assays cross -reactive with bovine molecules but using antibody raised against murine or human cytokines were used. PBMC contain cells other than T cells that can produce TLR-activated cytokines. These include monocytes and B cells and probably NK cells. The PBMCs as in the work of previous chapters was a positive control for TLR-ligand activity and TLR engagement (although this latter was not specifically molecularly measured). Using PBMCs as target cells for TLR-agonist stimulation measured by CXCL8 production confirmed that the TLR agonists were active on bovine cells, although whether the optimum concentration range was used or not was not further explored. Within the PBMC, monocytes, B cells, possibly NK cells as well as T cells amongst other cell types are all capable of responding to the TLR agonists.

Using the purified T cell subsets, only TLR3L, TLR8L and TLR9L were used to stimulate the cells for cytokine analyses to accommodate the capacity of the ELISA kits and also due to the fact that these analyses were performed at the end of the study period, where there was not time to do a comprehensive analysis. They were prioritised as they are involved in anti-viral immunity, an interest of our research group. CD4+ T cells, CD8+ T cells and  $\gamma\delta$  T cells from bov 1, bov 4 responded to TLR3L, TLR8L and TLR9L in all the T cell subsets to produce CXCL8, and bov 2 and bov 5 to specific TLR agonists to produce CXCL8. Why these animals were

responsive in these assays and not (for the most part) other animals is not known. This is discussed further in the general discussion.

Amongst the other cytokine responses studied, both PBMC and T cell subsets of a proportion of animal samples responded to TLRLs to produce IL-4, and IL-10, and TNF-alpha and IFN-alpha to a much lesser extent. TGF-beta was not detected in a preliminary analysis (although the number of samples analysed was too low to draw any conclusions).

Within the T cell subsets, the production of IL4 and IL10 by a proportion of the samples analysed is interesting given that T cell subset (CD4+ T cells, CD8+ T cells,  $\gamma\delta$  T cells) production of IFN-gamma was not detected. Interferons are a family of low molecular weight proteins secreted by many host cells in response to intracellular bacteria, viruses, or against tumours. We noticed an absence or low levels of type I interferons (IFNa) in the supernatants of PBMC or CD4+ or CD8+ T cell subset samples stimulated with TLR ligands. It had been recorded by Charleston *et al.* (2001) who found confusing results, with IFNa produced from cells infected with cytopathic BVD virus but not produced when non-cytopathic BVD virus was used. Herath *et al.* (2009) studied the expression of TLRs1-10 in the uteri of fertile and non-fertile cattle and discovered that IFNa was at similar levels in both, unlike IL1a, IL1 $\beta$ , or IL10 which were elevated in the infertile cattle compared to fertile cattle.

Interestingly PBMCs and T cell subsets of some animals produced IL-4 to TLR-ligand stimulation. IL-4 is difficult to detect in bovine samples (except in some specific disease states). This is also of interest as IFNγ responses were not prevalent. The reason for this is not clear, but could reflect a Th2 dominance of immune response (IL-4 mediated) over a Th1 response (IFN-

y mediated) in the T cells from these animals. However, this is highly speculative with no supporting evidence.

IFNy is a Th1 cytokine which is derived from CD4+ T cells amongst other cells (Ike et al., 2005) and was secreted in PBMCs, but not CD4+, CD8+ or  $\gamma\delta$  T cells stimulated with TLR agonists. IFN $\gamma$  is known to be produced by activated NK cells, CD4 T cells (particularly T helper 1 cells), and CD8 T cells predominantly (Harty *et al.*, 1992, Ike *et al.*, 2005).  $\gamma\delta$  T cells also produce this cytokine (Skeen and Ziegler, 1995) as do monocytes/ macrophages under some circumstances (Munder et al., 1998). All of above mentioned potentially IFN<sub>Y</sub> secreting cells are part of the PBMC pool. The lack of IFNy by TLR-ligand stimulated T cells remains to be explained. One of the major cytokines released by cytotoxic T cells is IFNy (Kennedy et al., 2002, Wherry and Ahmed, 2004, Shin et al., 2005, Stabel et al., 2007). Baldwin *et al.* (2000) detected IFNy secreting  $\gamma\delta$  T cells (phenotype WC1+, which is the same as the cells we use, fractionated using the WC1specific CC15 mAb) by two methods; flow cytometry and ELISA. Vesosky et al. (2004) expanded purified  $\gamma\delta$  T cells co-cultured with cell wall particles of Mycobacterial mycolylarabino-galactan (a peptidoglycan molecule similar to TLR2 agonist we used). These  $\gamma\delta$  T cells produced significant levels of IFNy. This was also seen by Price and Hope, (2009) but using another species of Mycobacterium in an APC (M. bovis-infected dendritic cells) instead. Later studies continued to support the concept of secreting IFNy by  $\gamma\delta$  T cells upon stimulation (Toka *et al.*, 2011, Plattner *et al.*, 2013).

T cells, mainly Th2 cells, are a source of IL-4 (they also respond to IL-4 stimulation), but this is not a generic rule (Nelms *et al.*, 1999, Chatila,

2004). We stimulated PBMCs (that includes T cells) with TLR1-9 agonists but recorded IL-4 secretion in the supernatant only in a proportion of the bovine samples. IL-4 secretion was detected from purified T cell subsets stimulate with TLR-agonists (novel and original observation). In this regard, Waldvogel *et al.*, (2000) measured very low but inconsistent levels of IL-4 and IFNγ by ELISA in PBMCs stimulated with non-cytopathic BVD virus in pregnant and non-pregnant cattle although the mRNA of both cytokines was positively detected by RT-qPCR method.

Sopp and Howard (2001) detected IL-4 secretion (by flow cytometry) from lymphocytes (collected from bovine lymph nodes and peripheral blood) *in vitro* activated by PMA (a T cell mitogen). They found that only CD4+ T cells produced IL-4 but not CD8+ nor  $\gamma\delta$  T cells. In this study IL-4 was secreted by  $\gamma\delta$  T cells from a couple of animal cells in response to TLRs 3 and 8 but not at all by CD8+ T cells.

IL-10 was detected in some PBMC samples and in some TLR3L, TLR8L and TLR9L stimulated CD4 T cells, CD8 T cells and  $\gamma\delta$  T cells (bov 1 bov 2, bov 4 and bov 5 in particular). This cytokine is known to possess immunoinhibitory activity, notably in autoimmune disease or cancer. This raises the possibility that Under some circumstances (not defined) TLR-agonist engagent could lead to T cell expression of IL-10 and a suppressive local environment in vivo for ongoing immune responses.

In conclusion, it is difficult to draw robust conclusions from this analysis, but there are indications of interesting results that should be followed up with a more incisive study.

**Chapter Seven** 

General discussion,

conclusions and recommendations

#### 7.1 General overview

In this study, the expression and function of TLRs on bovine T cells has been researched. The T cell subsets studied included CD4+ T cells, CD8+ T cells and  $\gamma\delta$  T cells from peripheral blood samples of cattle. We originally wished to include Tregs as well, but the work on rabbit T cells (to inform cattle Treg development in culture) indicated that this would be time consuming and the cells did not in any case show suppressive activity, so this was dropped.

TLRs1-10 were analysed on the T cell subsets by RT-qPCR. Some did not express TLR5 and TLR6, but the other samples did express all of the TLRs. This is the first description of this that we are aware of in cattle. Cell purities were adequate to allow allocation of TLRs to the T cell subsets but with the caveat that the few non T cells in the preparations could have contributed to the analysis. There is a lack of bovine-specific TLR blocking antibodies to confirm TLR-ligand specificity for the TLR, but there are conclusions that can drawn, nevertheless. Differences between whole PBMCs and the T cell subsets with respect to TLR expression are likely attributable in the main to the presence in the PBMCs of monocytes, NK cells and B cells, all of which are known to express TLRs in cattle (and other animals) (Werling *et al.*, 2006, Werling *et al.*, 2017).

An important observation is that differences in TLR expression and more obviously function (see below) were seen in the different animals of PBMC and their derived T cell subsets. The reasons for this are not known but may indicate previous immune activation status of the donor animals or involvement of other co-factors or molecules for TLR expression and

function not present in PBMC or the T cell subsets *in vitro*. This is discussed further below.

Functionally, the CFSE assay revealed proliferation in response to TLR agonists in T cell subsets in a non-consistent way but not in PBMCs. TLR3agonist, TLR8-agonist, and TLR9-agonist stimulation of CD4+, CD8+ and  $\gamma\delta$  T cells in 3-4 animals induced measurable cell division /proliferation by the CFSE assay. However, this aspect of the study requires further validation of techniques and time constraints did not allow this, focussing instead on cytokine production as a read out for TLR-agonist activated cells. The cytokine assays revealed some interesting results, were a better read out of TLR-ligand activation of the cells and indicated differential expression of the cytokines at the T cell level and the TLR-agonist used. This is discussed below. Time constraints meant that some of the work is incomplete and in particular for T cell subset activation studies the number of TLR-ligands used was reduced from all those stimulating all 9 TLRs to those targeting TLRs 3 (dsRNA as ligand), 8 (single stranded viral RNA ligand) and 9 (unmethylated CpG oligodeoxynucleotide DNA) that are important in anti-virus responses, an interest of our research group.

#### 7.2 Putative Rabbit Tregs

Phenotypic properties of regulatory T cells in animals and humans have been mostly identified as (CD4+CD25hiFOXP3+). They have been investigated in laboratory animals such as mice (Fontenot *et al.*, 2005b, Ziegler, 2006, Murai *et al.*, 2009, Waight *et al.*, 2015), guinea pigs (Shang *et al.*, 2011, Kato-Maeda *et al.*, 2012, Clark *et al.*, 2015), rats (Donia *et al.*, 2009, Fujiki *et al.*, 2010, Jin *et al.*, 2013) but not in rabbits. We attempted to generate putative Tregs in rabbit MLN cells by adding titrated levels of ConA, IL-2, and TGF $\beta$  to cells in IMDM medium and we successfully obtained up to 25% of putative rabbit nTreg cells in the total cell population after 6 days of culture (detected by flow cytometry and confirmed by RT-qPCR as FOXP3+). Therefore, we documented the phenotype for the first time in rabbits.

Our results show that rabbit MLN cells responded well to IL-2 which is essential for development of CD25+ external marker (Zheng *et al.*, 2007), and we can attribute FOXP3+ (a key feature and intracellular marker of nTregs) in the generated cells to TGF $\beta$ , agreeing with (Chai *et al.*, 2005), while ConA provided a non specific T cell proliferation (Wang *et al.*, 2012, Ando *et al.*, 2014).

With regard to the functional role of the putative rabbit Tregs, there was no suppression of the proliferation of autologous MLN cells. This is at variance with the generally accepted view that Tregs function as immunosuppressive cells through secretion of immune-inhibitory cytokines (mostly IL-10) (Anderson *et al.*, 2007, Murai *et al.*, 2009, Nylén *et al.*, 2007, Sakaguchi *et al.*, 2009). We did not measure this cytokine in the supernatant.

There has been a description of human FOXP3+ gene detected in some T effector cells and also in CD4+CD25+ Tregs that did not show any suppressive action or inhibitory cytokine production (Allan *et al.*, 2007). Liu *et al.* (2006) found that CD127+ was essential for CD4+FOXP3+ human Tregs function. We did not look for this. Tran *et al.* (2007) tested the effect of TGF $\beta$  on FOXP3 expression which was enhanced (FOXP3+) in Tregs but

these cells were not inhibitory in spite of the secretion of immunosuppressive cytokines.

Finally, it was found that induced human Tregs (FOXP3+) in the intestine play different role represented by anti-inflammatory role rather than immune-suppression (Round and Mazmanian, 2010). In summary, we discontinued the Treg culture approach for bovine Tregs.

#### 7.3 Bovine T cell TLR gene expression

We reported a higher frequency of  $\gamma \delta$  T cells compared to other T cell subsets in the bovine blood samples from seven animals. This along with the lower frequencies of B cells, CD8+ T cells and CD4+ T cells is consistent with the results of others, who also point out that young ruminants have the highest frequency of  $\gamma \delta$  T cells that decline in number with age (Hayday and Tigelaar, 2003, Telfer and Baldwin, 2015, Gillespie *et al.*, 2017).

The expression of TLR1-10 genes in bovine PBMCs was demonstrated, as well as in CD4+, CD8+, and  $\gamma\delta$  T cells, which was demonstrated in this study for the first time. However, TLR5 and TLR6 were not expressed in 2 animals. These results are roughly in line with studies on T cell TLR expression in other species (mostly humans and mice). In these studies TLR2, TLR3, TLR4, TLR7, and TLR8 expression was generally most prevalent on T cells amongst the TLRs analysed (Peng *et al.*, 2005, Kabelitz, 2007, González-Navajas *et al.*, 2010, Reynolds *et al.*, 2010, Flaherty and Reynolds, 2016, Pacheco *et al.*, 2016).

In this study, bovine CD8+ T cells expressed TLR1-10 genes. Observations on cytotoxic T cells (in humans and mice) have revealed intracellular TLR3, and TLR7-9 genes (Kabelitz, 2007, Vanders *et al.*, 2015, Wu *et al.*, 2015b)

while recently, purified CD8+ T cells were found to express some extracellular TLRs (TLR1, TLR2, and TLR4-6 genes) in certain disease conditions (McCarron and Reen, 2009, Hammond *et al.*, 2010, Freeman *et al.*, 2013, Portou *et al.*, 2015).

TLR1-10 genes were detected in  $\gamma\delta$  T cells. Research on human  $\gamma\delta$  T cells has indicated the presence of TLR1-9 genes (Pietschmann *et al.*, 2009, Bonneville *et al.*, 2010, Hannani *et al.*, 2012, Zheng *et al.*, 2013, Dar *et al.*, 2014). In conclusion, this study demonstrates that in common with other studied mammalian species, bovine T cells express a range or TLRs, conferring on them the potential to respond directly to TLR-agonists.

## 7.4 The activation status of bovine TLRs assessed by CFSE and cytokine assays and functional aspects of yō T cells

The non responsiveness of PBMCs in the CFSE assay could be due to preferential uptake of the TLR agonists by monocytes, B cells and other non T cells, that would not be stimulated to proliferate, or in the case of monocytes an activation involving immunosuppression (e.g by IL-10 production). Activated monocytes can cause death of T cells (apoptosis) (Beck *et al.*, 2011, Sepulcre *et al.*, 2011). This is highly speculative though. In contrast, CD4+, CD8+, and  $\gamma\delta$  T cells showed propagation in response to TLR3, TLR8, and TLR9 agonists in certain animals which indicates that these TLR are functional on these cells. TLR3, TLR8, and TLR9 agonists consist of dsRNA, ssRNA, and DNA nucleic acids (Bhardwaj *et al.*, 2010).

Recently, based on the literature (Hoek *et al.*, 2009, Guzman *et al.*, 2014), it is clear that  $\gamma\delta$  T cells may function as regulatory T cells which could work alongside or instead of Tregs CD4+CD25hi FOXP3+ T cells in cattle. However, in this study, the  $\gamma\delta$  T cells were stimulatory for autologous T cells (both unstimulated and ConA activated. It is possible that this is due to residual Con-A in the assay although steps were taken to wash this away. Alternatively, the  $\gamma\delta$  T cells may need to be stimulated in a different way, are the wrong subset or the target cells are inappropriate. There was not time to explore these various alternatives.

#### 7.5 Cytokine responses

Proliferation requires particular signalling pathways and is not the only measure of cell activation. We could have looked at signalling pathways, but the easiest method available to us was to look for cytokine production from the T cell subsets as a measure of TLR activation by agonists (CXCL-8, IFNa, IFN $\gamma$ , TNFa, TGF $\beta$ , IL-4, and IL-10). This also allows us to speculate on differential function (mediated by different cytokines).

Results of CFSE assays (proliferation assay) were incomplete and although some information was obtained from a few animal T cell samples (bov 1 bov 4 and bov 5 for CD4 T cell and CD8 T cell responses to TLR3 ligand and TLR8 ligand, and bov A and bov B and bov 1 for  $\gamma\delta$  T cells stimulated with TLR8 ligand and TLR3 ligand) time constraints meant that refining this assay was abandoned in favour of developing the cytone assays as a read out for T cell subset responses to TLR-ligands.

The pro-inflammatory CXCL-8 is used as a major readout of TLR signalling processes (Fonseca *et al.*, 2011, Mills, 2011, Russell, 2012, Russell *et al.*,

2012, Smith, 2012). CXCL8 production was observed in TLRs 1-9 ligandstimulated PBMCs from 3 to 5 of seven cattle. Bov 1, 2, 4 and 5 T cell subsets responded to TLR 3, 8 and 9 ligands to produce CXCL8. In these animals this is good evidence of TLR-ligand activation of the cells.

Secretion of the other (above-mentioned) cytokines might indicate some further specific functional role. Of interest was the production of IL-4 by CD4+ T cells and  $\gamma\delta$  T cells (but not CD8+ T cells) from bov 1 and bov 2 animals, to TLR3, 8 and 9 ligands (CD4 T cells) and to TLR 3 and 8 ligands (CD8 T cells). This could indicate a Th2 bias in the responder animals by T cells, and this result should be validated and followed up. However the production of IFN- $\gamma$  by PBMCS in 3 animals (bov 5, bov B and bovine A) needs to be explained.

IL-10 production was produced by T cell subsets to TLR3 Ligand and differentially to TLR8 and TLR9 ligand in bovs 1, 2 and 4 and 5 (to a lesser extent). This suggests that, in these animals T cells could have a local suppressive function. Once again this is highly speculative. TNF-a and IFN-a responses to TLR-ligands in PBMC and T cells was a very limited study and the results are equivocal. TGF- $\beta$  analyses were pilot only and limited to two animals. Time constraints limited this aspect of the study.

#### 7.6 Conclusions and recommendations

Putative rabbit nTregs phenotypically identified as CD4+CD25hiFOXP3+ cells were generated in culture. The putative rabbit Tregs lacked immunosuppressive activity and equivalent cattle Tregs were not developed for inclusion in the main objectives of the study. Importantly, this study has identified for the first time the expression of TLRs on

fractionated CD4+ T cells, CD8+ T cells and  $\gamma\delta$ + T cells. Functional studies (cytokine production by the T cell subsets to TLR agonist stimulation) indicated that this approach has value and should be followed up.

The most important outcome was that only a proportion of animal T cells responded to TLR-ligands. It was interesting to note that cells from same animals consistently gave a result in the various functional assays. These were: bov 1, bov 2, bovine 4098, and to a lesser extent bovine 4, bovine 5 then bovine A and bovine B. The other animals were poor or non responders. The two main reasons for this are likely to be (A) that the TLRligands are not active on bovine cells and (B) that the immune status of the animals is vaiable within the cohort. We can prett well rule out (A) because the CXCL8 responses of responder animal PBMCs to all 9 TLRligands were positive. It is still possible that the concentration range used may not be optimal. More likely is that the animal cells were differentially responsive to the TLR-ligands as a result of ongoing inflammatory or immune responses that may have inhibited responses to the TLR-ligands. The animals listed above as good all round responders could not be differentiated from any of the others at the clinical level as all were passed as clinically disease-free and healthy. To get round this, in future in may be prudent to use pathogen-free animals of a given breed, gender and age. Future work would be to refine the techniques and study the consequences of TLR agonist stimulation of T cells in the context of immune responses to antigen in vitro and possibly later in vivo.

Appendices

## 8.1 Preparations of generic reagents, media, and compounds

#### 8.1.1 RPMI complete medium

Material	Catalogue number/supplier
500 ml RPMI 1640 medium	11544526 – Fisher Scientific,
	Loughborough, UK
55 ml heat inactivated foetal calf	10073772 – Fisher Scientific,
serum (FCS)	Loughborough, UK
5.5 ml of 10,000 (U/ml) Penicillin/	15140122 – Fisher Scientific,
10 (mg/mL) Streptomycin	Loughborough, UK
1 ml Gibco Amphotericin B	15290-018 – Fisher Scientific,
(Fungizone®) 0.25 µg/ml	Loughborough, UK
5M of L-glutamine	G7513 - SIGMA-ALDRICH, Dorset,
	UK

#### 8.1.2 Freeze-mix medium

Material	Catalogue number/supplier
45 ml heat inactivated foetal calf	10073772 – Fisher Scientific,
serum (FCS)	Loughborough, UK
5 ml DMSO	276855 - SIGMA-ALDRICH,
	Dorset, UK

#### 8.1.3 MACS running buffer

This was made up by mixing 4 ml ethylenediaminetetra-acetic acid (EDTA 500mM) + 5 gm bovine serum albumin (BSA) dissolved in 1 L PBS (commenced with 800 ml then finalised to 1 L). pH adjusted to 7.2 then sterile filtered and finally stored at  $4^{\circ}$ C in the fridge.

#### 8.1.4 MACS rinsing buffer

This was made up by following the same steps in above section (8.1.3) to prepare MACS running buffer, but without adding BSA.

#### 8.1.5 Tris Acetate EDTA buffer (50X TAE)

It was made up by mixing 242g of Tris base plus 57.1 ml of glacial acetic acid plus 0.5M EDTA in 800 ml of distilled water (DW) then topped up to form a total volume of 1L and stored at room temperature.

#### 8.1.6 Tris Acetate EDTA buffer (1X TAE)

It was made up by diluting 50X TAE (described in above section 8.1.5) as follows: to prepare 1L of 1X TAE, 20 ml of 50X TAE dissolved in 980 ml DW.

#### 8.1.7 Erythrocyte lysis buffer (ELB)

To make up 1L of ELB, a mixture of 8.3 g ammonium chloride (NH<sub>4</sub>Cl at 0.15M working conc) plus 1 g potassium carbonate (KHCO3 at 10mM working conc) was dissolved in 200 ul of 0.5M EDTA at 0.1mM working conc, all dissolved in 800 ml double distilled water (ddH2O), and pH adjusted to 7.2 - 7.4, then volume adjusted to 1 Litre with ddH2O and sterile filtered then stored in the fridge.

### 8.2 Antibodies used in putative rabbit Treg project

Antibody	Catalogue number/supplier
Mouse IgG2a isotype control:FITC	MCA929F- BIO-RAD, Watford, UK
Mouse anti rabbit CD4:FITC	MCA799F - BIO-RAD, Watford, UK
Mouse IgG2b purified CD25	MG2b00 - Thermofisher Scientific,
isotype control	Rugby, UK
Mouse anti rabbit CD25	MCA1119GA - BIO-RAD, Watford,
(unconjugated)	UK
Goat anti-Mouse IgG2b Human	M32404 – Life technologies,
Adsorbed R-PE	Carlsbad, USA
Anti-Mouse/Rat FOXP3 APC	17-5773-82 - Thermofisher
	Scientific, Rugby, UK
Rat IgG2a K Isotype Control APC	17-4321-41 - Thermofisher
	Scientific, Rugby, UK

## 8.3 Antibodies used in bovine TLR project

Antibody	Catalogue
	number/supplier
Mouse anti-bovine CD4 T cells	MCA834GA - BIO-RAD,
	Watford, UK
Mouse anti-bovine CD8 T cells	MCA837GA - BIO-RAD,
	Watford, UK
Mouse anti-bovine γδ T cells	MCA838G - BIO-RAD,
	Watford, UK
Goat anti-mouse IgG microbeads	130-048-401 - Miltenyi
	Biotec, Woking, UK
Mouse anti-bovine CD11b FITC	MCA1425F - BIO-RAD,
	Watford, UK
Mouse anti-bovine CD21 FITC	MCA1424F - BIO-RAD,
	Watford, UK
Mouse anti-bovine CD4 FITC	MCA1653F - BIO-RAD,
	Watford, UK
Mouse anti-bovine CD8 FITC	MCA2216F - BIO-RAD,
	Watford, UK
Mouse anti-bovine WC1 (CC15 clone) FITC	MCA838F - BIO-RAD,
	Watford, UK
Mouse IgG2a Negative Control	MCA929F - BIO-RAD,
	Watford, UK
Mouse IgG2b Negative Control	MCA691F - BIO-RAD,
	Watford, UK
MOUSE ANTI BOVINE CD4:RPE	MCA1653PE - BIO-RAD,
	Watford, UK
MOUSE IgG2a NEGATIVE CONTROL:RPE	MCA929PE - BIO-RAD,
	Watford, UK
MOUSE ANTI BOVINE CD8: Alexa Fluor® 647	MCA837A647 - BIO-
	RAD, Watford, UK
MOUSE IgG2a NEGATIVE CONTROL:	MCA929A647 - BIO-
Alexa Fluor® 647	RAD, Watford, UK

### **8.4 ELISA kits used for cytokines' measurements**

Kit name	Catalogue number/supplier
Bovine Interleukin 8 ELISA kit	DIY1028B-003, Kingfisher Biotech
	Inc, St. Paul, USA
Bovine IFN-aA ELISA kit	DIY0663B-003, Kingfisher Biotech
	Inc, St. Paul, USA
Bovine TNF-a ELISA kit	DIY0675B-003, Kingfisher Biotech
	Inc, St. Paul, USA
Bovine IFN-γ ELISA development	3119-1H-6 / Mabtech, Cincinnati,
kit	USA
TGFβ ELISA kit	ABIN996501, Antibodies-online,
	Aachen, Germany
"Matched Antibody Pairs" used f	or detection of IL-4 and IL-10
plus their standards (recombina	nt proteins)
Mouse anti bovine interleukin-4	MCA2371, BIO-RAD, Watford, UK
Mouse anti bovine interleukin-4:	MCA2372B, BIO-RAD, Watford, UK
Biotin	
Recombinant bovine interleukin-4	PBP010, BIO-RAD, Watford, UK
Mouse anti bovine interleukin-10	MCA2110, BIO-RAD, Watford, UK
Mouse anti bovine interleukin-10:	MCA2111B, BIO-RAD, Watford, UK
Biotin	
Recombinant bovine interleukin-10	PBP016A, BIO-RAD, Watford, UK

## 8.5 Kits used for gene expression and molecular work

Kit name	Catalogue number/supplier
RNeasy Plus Mini Kit	74136 - Qiagen, Manchester, UK
RNase-Free DNase Set	79254 - Qiagen, Manchester, UK
Transcriptor First Strand cDNA	04897030001 – Roche, Burgess
Synthesis Kit	Hill, UK
Taq PCR Kit	E5000S - New England Biolabs,
	Hitchin, UK
QIAquick <sup>®</sup> Gel Extraction Kit	28704 - Qiagen, Manchester, UK
Probes Master Kit (for qPCR)	04707494001 – Roche, Burgess
	Hill, UK
Low molecular weight DNA ladder	N3233S - New England Biolabs,
	Hitchin, UK
Gel loading dye	N3233S - New England Biolabs,
	Hitchin, UK
References

## **Bibliography**

- ABBAS, A. K., LICHTMAN, A. H. & PILLAI, S. 2014. *Cellular and Molecular Immunology: With Student Consult Online Access*, Elsevier Health Sciences.
- ADEREM, A. & ULEVITCH, R. J. 2000. Toll-like receptors in the induction of the innate immune response. *Nature*, 406, 782-7.
- ADEREM, A., HAYASHI, F., SMITH, K. D., UNDERHILL, D. M. & OZINSKY, A. 2014. Toll-like receptor 5 ligands and methods of use. Google Patents.
- ADHIKARI, A., XU, M. & CHEN, Z. 2007. Ubiquitin-mediated activation of TAK1 and IKK. *Oncogene*, 26, 3214-3226.
- AGGARWAL, B. B., GUPTA, S. C. & KIM, J. H. 2012. Historical perspectives on tumor necrosis factor and its superfamily: 25 years later, a golden journey. *Blood*, 119, 651-65.
- AKIRA, S. 2003. Mammalian Toll-like receptors. *Current Opinion in Immunology*, 15, 5-11.
- AKIRA, S., UEMATSU, S. & TAKEUCHI, O. 2006. Pathogen Recognition and Innate Immunity. *Cell*, 124, 783-801.
- AKUTHOTA, P., WANG, H. B. & WELLER, P. F. 2010. Eosinophils as antigenpresenting cells in allergic upper airway disease. *Current Opinion in Allergy and Clinical Immunology*, 10, 14-19.
- ALEXOPOULOU, L., HOLT, A. C., MEDZHITOV, R. & FLAVELL, R. A. 2001. Recognition of double-stranded RNA and activation of NF-κB by Toll-like receptor 3. *Nature*, 413, 732-738.
- ALIPRANTIS, A. O., YANG, R.-B., MARK, M. R., SUGGETT, S., DEVAUX, B., RADOLF, J. D., KLIMPEL, G. R., GODOWSKI, P. & ZYCHLINSKY, A. 1999. Cell activation and apoptosis by bacterial lipoproteins through toll-like receptor-2. *Science*, 285, 736-739.
- ALLAN, S. E., CROME, S. Q., CRELLIN, N. K., PASSERINI, L., STEINER, T. S., BACCHETTA, R., RONCAROLO, M. G. & LEVINGS, M. K. 2007. Activationinduced FOXP3 in human T effector cells does not suppress proliferation or cytokine production. *International Immunology*, 19, 345-354.

- ALLAVENA, P., GARLANDA, C., BORRELLO, M. G., SICA, A. & MANTOVANI, A. 2008. Pathways connecting inflammation and cancer. *Current Opinion in Genetics & Development*, 18, 3-10.
- AMULIC, B., CAZALET, C., HAYES, G. L., METZLER, K. D. & ZYCHLINSKY, A. 2012. Neutrophil function: from mechanisms to disease. *Annual Review of Immunology*, 30, 459-89.
- ANDERSON, B., WATSON, D. & COLDITZ, I. 1999. The effect of dexamethasone on some immunological parameters in cattle. *Veterinary Research Communications*, 23, 399-413.
- ANDERSON, C. F., OUKKA, M., KUCHROO, V. J. & SACKS, D. 2007. CD4+ CD25– Foxp3– Th1 cells are the source of IL-10–mediated immune suppression in chronic cutaneous leishmaniasis. *Journal of Experimental Medicine*, 204, 285-297.
- ANDO, Y., YASUOKA, C., MISHIMA, T., IKEMATSU, T., UEDE, T., MATSUNAGA, T.
   & INOBE, M. 2014. Concanavalin A-mediated T cell proliferation is regulated by herpes virus entry mediator costimulatory molecule. *In Vitro Cellular & Developmental Biology-Animal*, 50, 313-320.
- ARESCHOUG, T. & GORDON, S. 2009. Scavenger receptors: role in innate immunity and microbial pathogenesis. *Cell Microbiology*, 11, 1160-9.
- ARIKAWA, E., SUN, Y., WANG, J., ZHOU, Q., NING, B., DIAL, S. L., GUO, L. & YANG, J. 2008. Cross-platform comparison of SYBR® Green real-time PCR with TaqMan PCR, microarrays and other gene expression measurement technologies evaluated in the MicroArray Quality Control (MAQC) study. *BMC Genomics*, 9, 328.
- ARMSTRONG, L., MEDFORD, A., HUNTER, K., UPPINGTON, K. & MILLAR, A. 2004. Differential expression of Toll-like receptor (TLR)-2 and TLR-4 on monocytes in human sepsis. *Clinical & Experimental Immunology*, 136, 312-319.
- ARSENAULT, R. J., JALAL, S., BABIUK, L. A., POTTER, A., GRIEBEL, P. J. & NAPPER, S. 2009. Kinome analysis of Toll-like receptor signaling in bovine monocytes. *Journal of Receptors and Signal Transduction*, 29, 299-311.
- ARTIS, D. & SPITS, H. 2015. The biology of innate lymphoid cells. *Nature*, 517, 293.
- ASEA, A., REHLI, M., KABINGU, E., BOCH, J. A., BARÉ, O., AURON, P. E., STEVENSON, M. A. & CALDERWOOD, S. K. 2002. Novel signal transduction pathway utilized by extracellular HSP70 role of Toll-like

receptor (TLR) 2 and TLR4. *Journal of Biological Chemistry*, 277, 15028-15034.

- AYOUB, I. A. & YANG, T. 1996. Age-dependent changes in peripheral blood lymphocyte subpopulations in cattle: a longitudinal study. *Developmental* & Comparative Immunology, 20, 353-363.
- BABU, S., BLAUVELT, C. P., KUMARASWAMI, V. & NUTMAN, T. B. 2006. Cutting edge: diminished T cell TLR expression and function modulates the immune response in human filarial infection. *The Journal of immunology*, 176, 3885-3889.
- BALCH, C., DOUGHERTY, P., CLOUD, G. & TILDEN, A. 1984. Prostaglandin E2mediated suppression of cellular immunity in colon cancer patients. *Surgery*, 95, 71-77.
- BALDWIN, C. L. & TELFER, J. C. 2015. The bovine model for elucidating the role of  $\gamma\delta$  T cells in controlling infectious diseases of importance to cattle and humans. *Molecular Immunology*, 66, 35-47.
- BALDWIN, C., SATHIYASEELAN, T., ROCCHI, M. & MCKEEVER, D. 2000. Rapid changes occur in the percentage of circulating bovine WC1+ γδ Th1 cells. *Research in Veterinary Science*, 69, 175-180.
- BALDWIN, C. L., YIRSAW, A., GILLESPIE, A., LE PAGE, L., ZHANG, F., DAMANI-YOKOTA, P. & TELFER, J. C. 2019. γδ T cells in livestock: Responses to pathogens and vaccine potential. *Transboundary and emerging diseases*.
- BANCHEREAU, J. & STEINMAN, R. M. 1998. Dendritic cells and the control of immunity. *Nature*, 392, 245-52.
- BARABAS, E., KEMENY, B., GONZALEZ, R., VARKONYI, V., NAGY, K. & HORVATH, A. 2002. A comparison of the lymphoproliferative capacity of pokeweed mitogen (PWM) and phytohaemagglutinin (PHA) in CCR-5 wild-type and heterozygous HIV-infected and uninfected subjects. *International Journal* of STD & AIDS, 13, 691-697.
- BARRY, M. & BLEACKLEY, R. C. 2002. Cytotoxic T lymphocytes: all roads lead to death. *Nature Reviews Immunology*, 2, 401-409.
- BARTON, G. M. & MEDZHITOV, R. 2003. Toll-like receptor signaling pathways. *Science*, 300, 1524-1525.
- BECK, B., DÖRFEL, D., LICHTENEGGER, F. S., GEIGER, C., LINDNER, L., MERK, M., SCHENDEL, D. J. & SUBKLEWE, M. 2011. Effects of TLR agonists on

maturation and function of 3-day dendritic cells from AML patients in complete remission. *Journal of Translational Medicine*, 9, 151.

- BECKMANN, M. P., COSMAN, D., FANSLOW, W., MALISZEWSKI, C. R. & LYMAN, S. D. 1992. The interleukin-4 receptor: structure, function, and signal transduction. *Interleukins: Molecular Biology and Immunology*. Karger Publishers.
- BEETZ, S., WESCH, D., MARISCHEN, L., WELTE, S., OBERG, H.-H. & KABELITZ, D. 2008. Innate immune functions of human γδ T cells. *Immunobiology*, 213, 173-182.
- BEISSERT, S., SCHWARZ, A. & SCHWARZ, T. 2006. Regulatory T cells. *Journal of Investigative Dermatology*, 126, 15-24.
- BEKEREDJIAN-DING, I., INAMURA, S., GIESE, T., MOLL, H., ENDRES, S., SING, A., ZÄHRINGER, U. & HARTMANN, G. 2007. Staphylococcus aureus protein A triggers T cell-independent B cell proliferation by sensitizing B cells for TLR2 ligands. *The Journal of Immunology*, 178, 2803-2812.
- BELARDELLI, F., FERRANTINI, M., PROIETTI, E. & KIRKWOOD, J. M. 2002. Interferon-alpha in tumor immunity and immunotherapy. *Cytokine & Growth Factor Reviews*, 13, 119-134.
- BELLA, J., HINDLE, K. L., MCEWAN, P. A. & LOVELL, S. C. 2008. The leucine-rich repeat structure. *Cellular and Molecular Life Sciences*, 65, 2307-2333.
- BELLOCCHIO, S., MONTAGNOLI, C., BOZZA, S., GAZIANO, R., ROSSI, G., MAMBULA, S. S., VECCHI, A., MANTOVANI, A., LEVITZ, S. M. & ROMANI, L. 2004. The contribution of the Toll-like/IL-1 receptor superfamily to innate and adaptive immunity to fungal pathogens *in vivo*. *The Journal of Immunology*, 172, 3059-3069.
- BERGHUIS, L., ABDELAZIZ, K. T., BIERWORTH, J., WYER, L., JACOB, G., KARROW, N. A., SHARIF, S., CLARK, M. E. & CASWELL, J. L. 2014.
   Comparison of innate immune agonists for induction of tracheal antimicrobial peptide gene expression in tracheal epithelial cells of cattle. *Veterinary Research*, 45, 1-10.
- BERNASCONI, N. L., ONAI, N. & LANZAVECCHIA, A. 2003. A role for Toll-like receptors in acquired immunity: up-regulation of TLR9 by BCR triggering in naive B cells and constitutive expression in memory B cells. *Blood*, 101, 4500-4504.
- BEUTLER, B. 2004. Innate immunity: an overview. *Molecular Immunology*, 40, 845-859.

- BHARDWAJ, N., GNJATIC, S. & SAWHNEY, N. B. 2010. TLR agonists: are they good adjuvants? *Cancer Journal (Sudbury, Mass.)*, 16, 382.
- BIDGOOD, S. R., TAM, J. C., MCEWAN, W. A., MALLERY, D. L. & JAMES, L. C. 2014. Translocalized IgA mediates neutralization and stimulates innate immunity inside infected cells. *Proceedings of the National Academy of Sciences*, 111 (37), 13463-13468.
- BIEBACK, K., LIEN, E., KLAGGE, I. M., AVOTA, E., SCHNEIDER-SCHAULIES, J., DUPREX, W. P., WAGNER, H., KIRSCHNING, C. J., TER MEULEN, V. & SCHNEIDER-SCHAULIES, S. 2002. Hemagglutinin protein of wild-type measles virus activates toll-like receptor 2 signaling. *Journal of Virology*, 76, 8729-8736.
- BILGEN, N., CINAR KUL, B., OFFORD, V., WERLING, D. & ERTUGRUL, O. 2016. Determination of Genetic Variations of Toll-Like Receptor (TLR) 2, 4, and 6 with Next-Generation Sequencing in Native Cattle Breeds of Anatolia and Holstein Friesian. *Diversity*, 8, 23.
- BLOHMKE, C. J., VICTOR, R. E., HIRSCHFELD, A. F., ELIAS, I. M., HANCOCK, D. G., LANE, C. R., DAVIDSON, A. G. F., WILCOX, P. G., SMITH, K. D. & OVERHAGE, J. 2008. Innate immunity mediated by TLR5 as a novel antiinflammatory target for cystic fibrosis lung disease. *The Journal of Immunology*, 180, 7764-7773.
- BOLHA, L., DUŠANIĆ, D., NARAT, M. & OVEN, I. 2012. Comparison of methods for relative quantification of gene expression using real-time PCR. *Acta Agriculturae Slovenica*, 100, 97-106.
- BOMAN, H. G. 2000. Innate immunity and the normal microflora. *Immunological Reviews*, 173, 5-16.
- BONELLI, M., SAVITSKAYA, A., STEINER, C.-W., RATH, E., SMOLEN, J. S. & SCHEINECKER, C. 2009. Phenotypic and functional analysis of CD4+ CD25– Foxp3+ T cells in patients with systemic lupus erythematosus. *The Journal of Immunology*, 182, 1689-1695.
- BONIZZI, G. & KARIN, M. 2004. The two NF-kappaB activation pathways and their role in innate and adaptive immunity. *Trends in Immunology*, 25, 280-8.
- BONNEVILLE, M., O'BRIEN, R. L. & BORN, W. K. 2010. γδ T cell effector functions: a blend of innate programming and acquired plasticity. *Nature Reviews Immunology*, 10, 467.

- BORGES DA SILVA, H., FONSECA, R., ALVAREZ, J. M. & D'IMPÉRIO LIMA, M. R. 2015. IFN-γ Priming Effects on the Maintenance of Effector Memory CD4. *Journal of Immunology Research*. Volume 2015, Article ID 202816, 1-8.
- BOTOS, I., SEGAL, D. M. & DAVIES, D. R. 2011. The structural biology of Tolllike receptors. *Structure*, 19, 447-459.
- BRASIER, A. R. 2010. The nuclear factor-κB-interleukin-6 signalling pathway mediating vascular inflammation. *Cardiovascular Research*, 86, 211-218.
- BRENNAN, K. M., BURRIS, W. R., BOLING, J. A. & MATTHEWS, J. C. 2010. Effects of selenium source on blood selenium content, blood cell counts and peripheral blood mononuclear cell mRNA profiles in maturing beef heifers. Federation of American Societies for Experimental Biology.
- BREZSKI, R. J. & MONROE, J. G. 2008. B-Cell Receptor. In: Sigalov A.B. (eds) Multichain Immune Recognition Receptor Signaling. *Advances in Experimental Medicine and Biology*, volume 640. Springer, New York.
- BRINKMANN, M. M., SPOONER, E., HOEBE, K., BEUTLER, B., PLOEGH, H. L. & KIM, Y.-M. 2007. The interaction between the ER membrane protein UNC93B and TLR3, 7, and 9 is crucial for TLR signaling. *The Journal of Cell Biology*, 177, 265-275.
- BSIBSI, M., RAVID, R., GVERIC, D. & VAN NOORT, J. M. 2002. Broad expression of Toll-like receptors in the human central nervous system. *Journal of Neuropathology & Experimental Neurology*, 61, 1013-1021.
- BUCHTA, C. M. & BISHOP, G. A. 2014a. Toll-like receptors and B cells: functions and mechanisms. *Immunologic Research*, 1-11.
- BUCHTA, C. M. & BISHOP, G. A. 2014b. TRAF5 negatively regulates TLR signaling in B lymphocytes. *The Journal of Immunology*, 192, 145-150.
- BUSTIN, S. 2002. Quantification of mRNA using real-time reverse transcription PCR (RT-PCR): trends and problems. *Journal of Molecular Endocrinology*, 29, 23-39.
- BUSTIN, S. A., BENES, V., GARSON, J. A., HELLEMANS, J., HUGGETT, J.,
  KUBISTA, M., MUELLER, R., NOLAN, T., PFAFFL, M. W. & SHIPLEY, G. L.
  2009. The MIQE guidelines: minimum information for publication of quantitative real-time PCR experiments. *Clinical Chemistry*, 55, 611-622.
- CALAME, K. L. 2001. Plasma cells: finding new light at the end of B cell development. *Nature Immunology*, 2, 1103-1108.

- CALON, A., TAURIELLO, D. & BATLLE, E. 2014. TGF-beta in CAF-mediated tumor growth and metastasis. *Seminars in Cancer Biology*. Elsevier, 15-22.
- CANTON, J., NECULAI, D. & GRINSTEIN, S. 2013. Scavenger receptors in homeostasis and immunity. *Nature Reviews Immunology*, 13, 621-634.
- CARDING, S. R. & EGAN, P. J. 2002. γδ T cells: functional plasticity and heterogeneity. *Nature Reviews Immunology*, 2, 336-345.
- CARGILL, E. J. & WOMACK, J. E. 2007. Detection of polymorphisms in bovine toll-like receptors 3, 7, 8, and 9. *Genomics*, 89, 745-55.
- CARON, G., DULUC, D., FRÉMAUX, I., JEANNIN, P., DAVID, C., GASCAN, H. & DELNESTE, Y. 2005. Direct stimulation of human T cells via TLR5 and TLR7/8: flagellin and R-848 up-regulate proliferation and IFN-γ production by memory CD4+ T cells. *The Journal of Immunology*, 175, 1551-1557.
- CESANA, G. C., DERAFFELE, G., COHEN, S., MOROZIEWICZ, D., MITCHAM, J., STOUTENBURG, J., CHEUNG, K., HESDORFFER, C., KIM-SCHULZE, S. & KAUFMAN, H. L. 2006. Characterization of CD4+ CD25+ regulatory T cells in patients treated with high-dose interleukin-2 for metastatic melanoma or renal cell carcinoma. *Journal of Clinical Oncology*, 24, 1169-1177.
- CHAI, J.-G., XUE, S.-A., COE, D., ADDEY, C., BARTOK, I., SCOTT, D., SIMPSON, E., STAUSS, H. J., HORI, S. & SAKAGUCHI, S. 2005. Regulatory T cells, derived from naive CD4+ CD25– T cells by *in vitro* Foxp3 gene transfer, can induce transplantation tolerance. *Transplantation*, 79, 1310-1316.
- CHAPLIN, D. D. 2010. Overview of the immune response. *The Journal of Allergy and Clinical Immunology*, 125, S3-23.
- CHARLESTON, B., FRAY, M., BAIGENT, S., CARR, B. & MORRISON, W. 2001. Establishment of persistent infection with non-cytopathic bovine viral diarrhoea virus in cattle is associated with a failure to induce type I interferon. *Journal of General Virology*, 82, 1893-1897.
- CHATILA, T. A. 2004. Interleukin-4 receptor signaling pathways in asthma pathogenesis. *Trends in Molecular Medicine*, 10, 493-499.
- CHEN, Q., KIM, Y. C., LAURENCE, A., PUNKOSDY, G. A. & SHEVACH, E. M. 2011. IL-2 Controls the Stability of Foxp3 Expression in TGF-β–Induced Foxp3+ T Cells In Vivo. *The Journal of Immunology*, 186, 6329-6337.
- CHEN, W., JIN, W., HARDEGEN, N., LEI, K.-J., LI, L., MARINOS, N., MCGRADY, G. & WAHL, S. M. 2003. Conversion of peripheral CD4+ CD25- naive T

cells to CD4+ CD25+ regulatory T cells by TGF-β induction of transcription factor Foxp3. *The Journal of Experimental Medicine*, 198, 1875-1886.

- CHENG, X., WANG, S., DAI, X., SHI, C., WEN, Y., ZHU, M., ZHAN, S. & MENG, J. 2012. Rabbit as a novel animal model for hepatitis E virus infection and vaccine evaluation. *PloS One*, **7**, e51616.
- CHIBA, E., VILLENA, J., HOSOYA, S., TAKANASHI, N., SHIMAZU, T., ASO, H., TOHNO, M., SUDA, Y., KAWAI, Y., SAITO, T., MIYAZAWA, K., HE, F. & KITAZAWA, H. 2012. A newly established bovine intestinal epithelial cell line is effective for in vitro screening of potential antiviral immunobiotic microorganisms for cattle. *Research in Veterinary Science*, 93, 688-694.
- CHIEN, Y.-H., MEYER, C. & BONNEVILLE, M. 2014. γδ T cells: first line of defense and beyond. *Annual Review of Immunology*, 32, 121-155.
- CHOW, J. C., YOUNG, D. W., GOLENBOCK, D. T., CHRIST, W. J. & GUSOVSKY, F. 1999. Toll-like receptor-4 mediates lipopolysaccharide-induced signal transduction. *Journal of Biological Chemistry*, 274, 10689-10692.
- CHU, C.-Y., LEE, S.-C., LIU, S.-S., LIN, Y.-M., SHEN, P.-C., YU, C., LEE, K.-H., ZHAO, X. & LEE, J.-W. 2011. Cytosine-phosphate-guanine oligodeoxynucleotides containing GACGTT motifs enhance the immune responses elicited by keyhole limpet hemocyanin antigen in dairy cattle. *Nucleic Acid Therapeutics*, 21, 323-332.
- CLARK, S., HALL, Y. & WILLIAMS, A. 2015. Animal models of tuberculosis: guinea pigs. *Cold Spring Harbor Perspectives in Medicine*, 5, a018572.
- COBAN, C., ISHII, K. J., KAWAI, T., HEMMI, H., SATO, S., UEMATSU, S., YAMAMOTO, M., TAKEUCHI, O., ITAGAKI, S. & KUMAR, N. 2005. Toll-like receptor 9 mediates innate immune activation by the malaria pigment hemozoin. *The Journal of Experimental Medicine*, 201, 19-25.
- COMBER, J. D. & PHILIP, R. 2014. MHC class I antigen presentation and implications for developing a new generation of therapeutic vaccines. *Therapeutic Advances in Vaccines and Immunotherapy*, 2, 77-89.
- COMPTON, T., KURT-JONES, E. A., BOEHME, K. W., BELKO, J., LATZ, E., GOLENBOCK, D. T. & FINBERG, R. W. 2003. Human cytomegalovirus activates inflammatory cytokine responses via CD14 and Toll-like receptor 2. Journal of Virology, 77, 4588-4596.
- CUI, Y., KANG, L., CUI, L. & HE, W. 2009. Human γδ T cell Recognition of lipid A is predominately presented by CD1b or CD1c on dendritic cells. *Biology Direct*, 4, 47.

- CURIEL, T. J. 2007. Tregs and rethinking cancer immunotherapy. *Journal of Clinical Investigation*, 117, 1167.
- CUROTTO DE LAFAILLE, M. A. & LAFAILLE, J. J. 2009. Natural and Adaptive Foxp3<sup>+</sup> Regulatory T Cells: More of the Same or a Division of Labor? *Immunity*, 30, 626-635.
- D'ALESSIO, F. R., TSUSHIMA, K., AGGARWAL, N. R., WEST, E. E., WILLETT, M. H., BRITOS, M. F., PIPELING, M. R., BROWER, R. G., TUDER, R. M. & MCDYER, J. F. 2009. CD4+ CD25+ Foxp3+ Tregs resolve experimental lung injury in mice and are present in humans with acute lung injury. *The Journal of Clinical Investigation*, 119, 2898.
- D'HAENE, B., VANDESOMPELE, J. & HELLEMANS, J. 2010. Accurate and objective copy number profiling using real-time quantitative PCR. *Methods*, 50, 262-270.
- DAR, A. A., PATIL, R. S. & CHIPLUNKAR, S. V. 2014. Insights into the Relationship between Toll Like Receptors and Gamma Delta T Cell Responses. *Frontiers in Immunology*, 5(366), 1-12.
- DAVIES, J. & DAVIES, D. 2010. Origins and evolution of antibiotic resistance. *Microbiology and Molecular Biology Reviews*, 74, 417-433.
- DE LA ROSA, M., RUTZ, S., DORNINGER, H. & SCHEFFOLD, A. 2004. Interleukin-2 is essential for CD4+ CD25+ regulatory T cell function. *European Journal of Immunology*, 34, 2480-2488.
- DELVES, P. J., MARTIN, S. J., BURTON, D. R. & ROITT, I. M. 2011. *Roitt's Essential Immunology*, John Wiley & Sons.
- DENHOLM, S., MCNEILLY, T., BANOS, G., COFFEY, M., RUSSELL, G., BAGNALL, A., MITCHELL, M. & WALL, E. 2014. Genetic Parameters of Immune Traits in Dairy Cattle. Proceedings, 10<sup>th</sup> World Congress of Genetics Applied to Livestock Production. asas.org. Pages 1-4.
- D'HAENE, B. 2013. Four tips for RT-qPCR data normalization using reference genes [Online]. Available: https://www.biogazelle.com/four-tips-rt-qpcr-data-normalization-using-reference-genes [Accessed 24-11-2015].
- DHANASEKARAN, S., BISWAS, M., VIGNESH, A. R., RAMYA, R., RAJ, G. D., TIRUMURUGAAN, K. G., RAJA, A., KATARIA, R. S., PARIDA, S. & SUBBIAH, E. 2014. Toll-Like Receptor Responses to Peste des petits ruminants Virus in Goats and Water Buffalo. *PloS one*, 9, e111609.

- DIELI, F., TROYE-BLOMBERG, M., IVANYI, J., FOURNIÉ, J. J., KRENSKY, A. M., BONNEVILLE, M., PEYRAT, M. A., CACCAMO, N., SIRECI, G. & SALERNO, A. 2001. Granulysin-dependent killing of intracellular and extracellular Mycobacterium tuberculosis by Vγ9/Vδ2 T lymphocytes. *The Journal of Infectious Diseases*, 184, 1082-1085.
- DIELI, F., TROYE-BLOMBERG, M., IVANYI, J., FOURNIÉ, J. J., BONNEVILLE, M., PEYRAT, M. A., SIRECI, G. & SALERNO, A. 2000. Vγ9/Vδ2 T lymphocytes reduce the viability of intracellular Mycobacterium tuberculosis. *European Journal of Immunology*, 30, 1512-1519.
- DINARELLO, C. A. 2009. Immunological and inflammatory functions of the interleukin-1 family. *Annual Review of Immunology*, 27, 519-550.
- DINARELLO, C. A. 2011. Interleukin-1 in the pathogenesis and treatment of inflammatory diseases. *Blood*, 117, 3720-3732.
- DOHERTY, R., O'FARRELLY, C. & MEADE, K. 2013. Epigenetic regulation of the innate immune response to LPS in bovine peripheral blood mononuclear cells (PBMC). *Veterinary Immunology and Immunopathology*, 154, 102-110.
- DOMÍNGUEZ, P. M. & ARDAVÍN, C. 2010. Differentiation and function of mouse monocyte-derived dendritic cells in steady state and inflammation. *Immunological Reviews*, 234, 90-104.
- DONIA, M., MANGANO, K., AMOROSO, A., MAZZARINO, M. C., IMBESI, R., CASTROGIOVANNI, P., COCO, M., MERONI, P. & NICOLETTI, F. 2009. Treatment with rapamycin ameliorates clinical and histological signs of protracted relapsing experimental allergic encephalomyelitis in Dark Agouti rats and induces expansion of peripheral CD4+ CD25+ Foxp3+ regulatory T cells. *Journal of Autoimmunity*, 33, 135-140.
- DORNER, M., BRANDT, S., TINGUELY, M., ZUCOL, F., BOURQUIN, J. P., ZAUNER, L., BERGER, C., BERNASCONI, M., SPECK, R. F. & NADAL, D. 2009. Plasma cell toll-like receptor (TLR) expression differs from that of B cells, and plasma cell TLR triggering enhances immunoglobulin production. *Immunology*, 128, 573-579.
- DUCLOUX, D. 2014. Regulatory T cells and cancer: an undesired tolerance. *Kidney International*, 86, 16-18.
- DUURLAND, C. L., BROWN, C. C., O'SHAUGHNESSY, R. F. & WEDDERBURN, L. R. 2017. CD161+ Tconv and CD161+ Treg share a transcriptional and functional phenotype despite limited overlap in TCRβ repertoire. *Frontiers in Immunology*, 8(103), 1-18.

- DÜVEL, A., MAAB, J., HEPPELMANN, M., HUSSEN, J., KOY, M., PIECHOTTA, M., SANDRA, O., SMITH, D. G., SHELDON, I. M. & DIEUZY-LABAYE, I. 2014. Peripheral blood leukocytes of cows with subclinical endometritis show an altered cellular composition and gene expression. *Theriogenology*, 81, 906-917.
- EARLE, K., TANG, Q., ZHOU, X., LIU, W., ZHU, S., BONYHADI, M. & BLUESTONE, J. 2005. *In vitro* expanded human CD4+ CD25+ regulatory T cells suppress effector T cell proliferation. *Clinical Immunology*, 115, 3-9.
- EL SAYED, E. L., MOSTAFA, H. A. & SHAFIK, N. G. 2016. Utility of laboratory animals for relative potency evaluation of bovine respiratory viral vaccine. *Benha Veterinary Medical Journal*, 31(2), 272 -275.
- EL-SHAFAEY, E. S., ATEYA, A., RAMADAN, H., SALEH, R., ELSEADY, Y., ABO EL FADL, E. & EL-KHODERY, S. 2017. Single Nucleotide Polymorphisms in IL8 and TLR4 Genes as Candidates for Digital Dermatitis Resistance/Susceptibility in Holstein Cattle. *Animal Biotechnology*, 28, 131-137.
- EMAM, M., THOMPSON-CRISPI, K. & MALLARD, B. 2015. The effect of immunological status, in-vitro treatment and culture time on expression of eleven candidate reference genes in bovine blood mononuclear cells. *BMC Immunology*, 16, 33.
- ENGELMANN, B. & MASSBERG, S. 2012. Thrombosis as an intravascular effector of innate immunity. *Nature Reviews Immunology*, 13, 34-45.
- FAIST, E., MEWES, A., STRASSER, T. & *ET AL*. 1988. Alteration of monocyte function following major injury. *Archives of Surgery*, 123, 287-292.
- FAJT, M. L., UVALLE, C., TRUDEAU, J. & WENZEL, S. E. 2014. High Blood Eosinophils Predict Reductions in Sputum Mast Cells and Lung Function In Response To Triamcinolone In Asthma. *Journal of Allergy and Clinical Immunology*, 133, AB61-AB61.
- FARBER, D. L., YUDANIN, N. A. & RESTIFO, N. P. 2014. Human memory T cells: generation, compartmentalization and homeostasis. *Nature Reviews Immunology*, 14, 24-35.
- FARHAT, K., RIEKENBERG, S., JUNG, G., WIESMULLER, K. H., JUNGI, T. W. & ULMER, A. J. 2010. Identification of full length bovine TLR1 and functional characterization of lipopeptide recognition by bovine TLR2/1 heterodimer. *Veterinary Research*, 41, 34.
- FAUSTINO, J. V., WANG, X., JOHNSON, C. E., KLIBANOV, A., DERUGIN, N., WENDLAND, M. F. & VEXLER, Z. S. 2011. Microglial cells contribute to

endogenous brain defenses after acute neonatal focal stroke. *The Journal of Neuroscience*, 31, 12992-13001.

- FELDMAN, K. E., LORIAUX, P. M., SAITO, M., TUERO, I., VILLAVERDE, H., SIVA, T., GOTUZZO, E., GILMAN, R. H., HOFFMANN, A. & VINETZ, J. M. 2013. *Ex Vivo* Innate Immune Cytokine Signature of Enhanced Risk of Relapsing Brucellosis. *PLoS Neglected Tropical Diseases*, 7, e2424.
- FERRAND, J. & FERRERO, R. L. 2013. Recognition of Extracellular Bacteria by NLRs and Its Role in the Development of Adaptive Immunity. *Frontiers in Immunology*, 4, 344.
- FIGDOR, C. G., VAN KOOYK, Y. & ADEMA, G. J. 2002. C-type lectin receptors on dendritic cells and Langerhans cells. *Nature Reviews Immunology*, 2, 77-84.
- FILIPPI, C. M. 2015. Toll-like receptor activation in immunity vs. tolerance. *Frontiers in Immunology*, 6, 146.
- FLAHERTY, S. & REYNOLDS, J. M. 2016. TLR Function in Murine CD4+ T Lymphocytes and Their Role in Inflammation. *Toll-Like Receptors: Practice and Methods*, 1390, 215-227.
- FONSECA, I., ANTUNES, G., PAIVA, D., LANGE, C., GUIMARÃES, S. & MARTINS, M. 2011. Differential expression of genes during mastitis in Holstein-Zebu crossbreed dairy cows. *Genetics and Molecular Research*, 10, 1295-1303.
- FONTENOT, J. D., RASMUSSEN, J. P., GAVIN, M. A. & RUDENSKY, A. Y. 2005a. A function for interleukin 2 in Foxp3-expressing regulatory T cells. *Nature Immunology*, 6, 1142.
- FONTENOT, J. D., RASMUSSEN, J. P., WILLIAMS, L. M., DOOLEY, J. L., FARR, A. G. & RUDENSKY, A. Y. 2005b. Regulatory T cell lineage specification by the forkhead transcription factor foxp3. *Immunity*, 22, 329-341.
- FOOKSMAN, D. R. 2014. Organizing MHC Class II Presentation. *Frontiers in Immunology*, 5, 158.
- FREEMAN, C. M., MARTINEZ, F. J., HAN, M. K., WASHKO, G. R., MCCUBBREY, A. L., CHENSUE, S. W., ARENBERG, D. A., MELDRUM, C. A., MCCLOSKEY, L. & CURTIS, J. L. 2013. Lung CD8+ T cells in COPD have increased expression of bacterial TLRs. *Respiratory Research*, 14, 13.
- FRY, T. J. & MACKALL, C. L. 2002. Interleukin-7: from bench to clinic. *Blood*, 99, 3892-904.

- FUJIKI, M., ESQUIVEL, C. O., MARTINEZ, O. M., STROBER, S., UEMOTO, S. & KRAMS, S. M. 2010. Induced tolerance to rat liver allografts involves the apoptosis of intragraft T cells and the generation of CD4+ CD25+ FoxP3+ T regulatory cells. *Liver Transplantation*, 16, 147-154.
- GAFFEN, S. 2016. IL-17 receptor composition. *Nature Reviews Immunology*, 16, 4-4.
- GAFFEN, S. L. & LIU, K. D. 2004. Overview of interleukin-2 function, production and clinical applications. *Cytokine*, 28, 109-123.
- GALLI, S. J., BORREGAARD, N. & WYNN, T. A. 2011. Phenotypic and functional plasticity of cells of innate immunity: macrophages, mast cells and neutrophils. *Nature Immunology*, 12, 1035-1044.
- GARBERS, C., HERMANNS, H. M., SCHAPER, F., MÜLLER-NEWEN, G., GRÖTZINGER, J., ROSE-JOHN, S. & SCHELLER, J. 2012. Plasticity and cross-talk of interleukin 6-type cytokines. *Cytokine & Growth Factor Reviews*, 23, 85-97.
- GAY, N. J., SYMMONS, M. F., GANGLOFF, M. & BRYANT, C. E. 2014. Assembly and localization of Toll-like receptor signalling complexes. *Nature Reviews Immunology*, 14, 546.
- GAZZINELLI, R. T. & DENKERS, E. Y. 2006. Protozoan encounters with Toll-like receptor signalling pathways: implications for host parasitism. *Nature Reviews Immunology*, 6, 895-906.
- GEISSMANN, F., MANZ, M. G., JUNG, S., SIEWEKE, M. H., MERAD, M. & LEY, K. 2010. Development of monocytes, macrophages, and dendritic cells. *Science*, 327, 656-661.
- GELMAN, A. E., ZHANG, J., CHOI, Y. & TURKA, L. A. 2004. Toll-like receptor ligands directly promote activated CD4+ T cell survival. *The Journal of Immunology*, 172, 6065-6073.
- GIBSON, A. J., WOODMAN, S., PENNELEGION, C., PATTERSON, R., STUART, E., HOSKER, N., SIVITER, P., DOUGLAS, C., WHITEHOUSE, J. & WILKINSON, W. 2016. Differential macrophage function in Brown Swiss and Holstein Friesian cattle. *Veterinary Immunology and Immunopathology*, 181, 15-23.
- GIBSON, A., MIAH, S., GRIEBEL, P., BROWNLIE, J. & WERLING, D. 2012. Identification of a lineage negative cell population in bovine peripheral blood with the ability to mount a strong type I interferon response. Developmental & Comparative Immunology, 36, 332-41.

- GIDON, A., ÅSBERG, S. E., LOUET, C., RYAN, L., HAUG, M. & FLO, T. H. 2017. Persistent mycobacteria evade an antibacterial program mediated by phagolysosomal TLR7/8/MyD88 in human primary macrophages. *PLoS Pathogens*, 13, e1006551.
- GILLESPIE, A., CONNELLEY, T., TELFER, J. C. & BALDWIN, C. L. 2017. Interaction of  $\gamma\delta$  TCR with the WC1 hybrid co-receptor/pathogen recognition receptor in cattle. *The Journal of Immunology*, 198 (1 Supplement) 226, 22
- GODFREY, W. R., SPODEN, D. J., YING, G. G., BAKER, S. R., LIU, B., LEVINE, B. L., JUNE, C. H., BLAZAR, B. R. & PORTER, S. B. 2005. Cord blood CD4+ CD25+-derived T regulatory cell lines express FoxP3 protein and manifest potent suppressor function. *Blood*, 105, 750-758.
- GOŁĄB, K., KRZYSTYNIAK, A., MAREK-TRZONKOWSKA, N., MISAWA, R., WANG, L. J., WANG, X., COCHET, O., TIBUDAN, M., LANGA, P. & MILLIS, J. M. 2013. Impact of culture medium on CD4<sup>+</sup> CD25<sup>high</sup>CD127<sup>lo/neg</sup> Treg expansion for the purpose of clinical application. *International Immunopharmacology*, 16, 358-363.
- GOLDAMMER, T., ZERBE, H., MOLENAAR, A., SCHUBERTH, H. J., BRUNNER, R. M., KATA, S. R. & SEYFERT, H. M. 2004. Mastitis increases mammary mRNA abundance of beta-defensin 5, toll-like-receptor 2 (TLR2), and TLR4 but not TLR9 in cattle. *Clinical and Vaccine Immunology*, 11, 174-85.
- GONI, R., GARCÍA, P. & FOISSAC, S. 2009. The qPCR data statistical analysis. Integromics White Paper, 1-9.
- GONZÁLEZ-NAVAJAS, J. M., FINE, S., LAW, J., DATTA, S. K., NGUYEN, K. P., YU, M., CORR, M., KATAKURA, K., ECKMAN, L. & LEE, J. 2010. TLR4 signaling in effector CD4+ T cells regulates TCR activation and experimental colitis in mice. *The Journal of Clinical Investigation*, 120, 570.
- GORDON, S. & TAYLOR, P. R. 2005. Monocyte and macrophage heterogeneity. *Nature Reviews Immunology*, 5, 953-64.
- GREENBERG, S. & GRINSTEIN, S. 2002. Phagocytosis and innate immunity. *Current Opinion in Immunology*, 14, 136-145.
- GRIEBEL, P. J., BROWNLIE, R., MANUJA, A., NICHANI, A., MOOKHERJEE, N., POPOWYCH, Y., MUTWIRI, G., HECKER, R. & BABIUK, L. A. 2005. Bovine toll-like receptor 9: a comparative analysis of molecular structure, function and expression. *Veterinary Immunology and Immunopathology*, 108, 11-6.

- GROEN, B., LINKS, T. P., LEFRANDT, J. D., VAN DEN BERG, P. P., DE VOS, P. & FAAS, M. M. 2013. Aberrant Pregnancy Adaptations in the Peripheral Immune Response in Type 1 Diabetes: A Rat Model. *PLoS One*, 8, e65490.
- GUO, M., WU, F., HAO, G., QI, Q., LI, R., LI, N., WEI, L. & CHAI, T. 2017. Bacillus subtilis improves immunity and Disease resistance in rabbits. Frontiers in Immunology, volume 8, Article 354, 1-13.
- GUZMAN, E., HOPE, J., TAYLOR, G., SMITH, A. L., CUBILLOS-ZAPATA, C. & CHARLESTON, B. 2014. Bovine γδ T cells are a major regulatory T cell subset. *The Journal of Immunology*, 193, 208-222.
- HAMMOND, T., LEE, S., WATSON, M. W., FLEXMAN, J. P., CHENG, W., FERNANDEZ, S. & PRICE, P. 2010. Toll-like receptor (TLR) expression on CD4+ and CD8+ T-cells in patients chronically infected with hepatitis C virus. *Cellular Immunology*, 264, 150-155.
- HANNANI, D., MA, Y., YAMAZAKI, T., DÉCHANET-MERVILLE, J., KROEMER, G. & ZITVOGEL, L. 2012. Harnessing γδ T cells in anticancer immunotherapy. *Trends in Immunology*, 33, 199-206.
- HANOUX, V., PAIRAULT, C., BAKALSKA, M., HABERT, R. & LIVERA, G. 2007. Caspase-2 involvement during ionizing radiation-induced oocyte death in the mouse ovary. *Cell Death & Differentiation*, 14, 671-681.
- HARP, J., KEHRLI JR, M., HURLEY, D., WILSON, R. & BOONE, T. 1991. Numbers and percent of T lymphocytes in bovine peripheral blood during the periparturient period. *Veterinary Immunology and Immunopathology*, 28, 29-35.
- HART, O. M., ATHIE-MORALES, V., O'CONNOR, G. M. & GARDINER, C. M. 2005. TLR7/8-mediated activation of human NK cells results in accessory celldependent IFN-γ production. *The Journal of Immunology*, 175, 1636-1642.
- HARTSHORNE, T., LE, F., LANG, J., LEONG, H., HAYASHIBARA, K., DEWOLF, D. & SHELTON, E. 2014. A High-throughput Real-time PCR Approach to Pharmacogenomics Studies. *Journal of Pharmacogenomics & Pharmacoproteomics*, 5, 1.
- HARTY, J. T., SCHREIBER, R. D. & BEVAN, M. J. 1992. CD8 T cells can protect against an intracellular bacterium in an interferon gamma-independent fashion. *Proceedings of the National Academy of Sciences*, 89, 11612-11616.

- HASHIMOTO, C., HUDSON, K. L. & ANDERSON, K. V. 1988. The Toll gene of drosophila, required for dorsal-ventral embryonic polarity, appears to encode a transmembrane protein. *Cell*, 52, 269-279.
- HAWKINS, E. D., HOMMEL, M., TURNER, M. L., BATTYE, F. L., MARKHAM, J. F. & HODGKIN, P. D. 2007. Measuring lymphocyte proliferation, survival and differentiation using CFSE time-series data. *Nature Protocols*, 2, 2057.
- HAYASHI, F., SMITH, K. D., OZINSKY, A., HAWN, T. R., EUGENE, C. Y.,
  GOODLETT, D. R., ENG, J. K., AKIRA, S., UNDERHILL, D. M. & ADEREM,
  A. 2001. The innate immune response to bacterial flagellin is mediated by
  Toll-like receptor 5. *Nature*, 410, 1099-1103.
- HAYASHI, T., NAKAMURA, T. & TAKAOKA, A. 2010. [Pattern recognition receptors]. Nihon Rinsho Men'eki Gakkai kaishi= Japanese Journal of Clinical Immunology, 34, 329-345.
- HAYDAY, A. & TIGELAAR, R. 2003. Immunoregulation in the tissues by gammadelta T cells. *Nature Reviews Immunology*, 3, 233.
- HAYDAY, A. C. 2000. γδ cells: a right time and a right place for a conserved third way of protection. *Annual Rreview of Immunology*, 18, 975-1026.
- HEDAYAT, M., NETEA, M. G. & REZAEI, N. 2011. Targeting of Toll-like receptors: a decade of progress in combating infectious diseases. *The Lancet Infectious Diseases*, 11, 702-712.
- HEIL, F., HEMMI, H., HOCHREIN, H., AMPENBERGER, F., KIRSCHNING, C., AKIRA, S., LIPFORD, G., WAGNER, H. & BAUER, S. 2004. Species-specific recognition of single-stranded RNA via toll-like receptor 7 and 8. *Science*, 303, 1526-1529.
- HEINZE, E., BALDWIN, S., CHAN, G., HANSEN, J., SONG, J., CLEMENTS, D., ARAGON, R., NISHIMURA, R., REEVES, M. & WEISBART, R. 2009. Antibody-mediated FOXP3 protein therapy induces apoptosis in cancer cells *in vitro* and inhibits metastasis *in vivo*. *International Journal of Oncology*, 35, 167-173.
- HEMMI, H., KAISHO, T., TAKEUCHI, O., SATO, S., SANJO, H., HOSHINO, K., HORIUCHI, T., TOMIZAWA, H., TAKEDA, K. & AKIRA, S. 2002. Small antiviral compounds activate immune cells via the TLR7 MyD88-dependent signaling pathway. *Nature Immunology*, 3, 196.
- HERATH, S., LILLY, S. T., SANTOS, N. R., GILBERT, R. O., GOETZE, L., BRYANT, C. E., WHITE, J. O., CRONIN, J. & SHELDON, I. M. 2009. Expression of genes associated with immunity in the endometrium of cattle with

disparate postpartum uterine disease and fertility. *Reproductive Biology* and Endocrinology, 7, 55.

- HERZIG, C. T. & BALDWIN, C. L. 2009. Genomic organization and classification of the bovine WC1 genes and expression by peripheral blood gamma delta T cells. *BMC Genomics*, 10, 191.
- HIDMARK, A., VON SAINT PAUL, A. & DALPKE, A. H. 2012. Cutting edge: TLR13 is a receptor for bacterial RNA. *The Journal of Immunology*, 189, 2717-2721.
- HILLEBRANDS, J.-L., WHALEN, B., VISSER, J. T., KONING, J., BISHOP, K. D., LEIF, J., ROZING, J., MORDES, J. P., GREINER, D. L. & ROSSINI, A. A. 2006. A regulatory CD4+ T cell subset in the BB rat model of autoimmune diabetes expresses neither CD25 nor Foxp3. *The Journal of Immunology*, 177, 7820-7832.
- HINDSON, B. J., NESS, K. D., MASQUELIER, D. A., BELGRADER, P., HEREDIA, N. J., MAKAREWICZ, A. J., BRIGHT, I. J., LUCERO, M. Y., HIDDESSEN, A. L. & LEGLER, T. C. 2011. High-throughput droplet digital PCR system for absolute quantitation of DNA copy number. *Analytical Chemistry*, 83, 8604-8610.
- HOEBE, K., JANSSEN, E. & BEUTLER, B. 2004. The interface between innate and adaptive immunity. *Nature Immunology*, 5, 971–974.
- HOEK, A., RUTTEN, V. P., KOOL, J., ARKESTEIJN, G. J., BOUWSTRA, R. J., VAN RHIJN, I. & KOETS, A. P. 2009. Subpopulations of bovine WC1 γδ T cells rather than CD4+CD25high Foxp3+ T cells act as immune regulatory cells *ex vivo*. *Veterinary Research*, 40, 1-14.
- HOLTMEIER, W. & KABELITZ, D. 2005. γδ T cells link innate and adaptive immune responses. *Chemical Immunology and Allergy*. Basel, Karger, 86, 151-183.
- HOPE, J., KWONG, L., SOPP, P., COLLINS, R. & HOWARD, C. 2000. Dendritic cells induce CD4+ and CD8+ T-cell responses to *Mycobacterium bovis* and *M. avium* antigens in Bacille Calmette Guerin vaccinated and nonvaccinated cattle. *Scandinavian Journal of Immunology*, 52, 285-291.
- HOPE, J., KWONG, L., THOM, M., SOPP, P., MWANGI, W., BROWN, W., PALMER, G., WATTEGEDERA, S., ENTRICAN, G. & HOWARD, C. 2005. Development of detection methods for ruminant interleukin (IL)-4. *Journal of Immunological Methods*, 301, 114-123.
- HORI, S., NOMURA, T. & SAKAGUCHI, S. 2003. Control of regulatory T cell development by the transcription factor Foxp3. *Science*, 299, 1057-1061.

- HORNUNG, V., ROTHENFUSSER, S., BRITSCH, S., KRUG, A., JAHRSDÖRFER, B., GIESE, T., ENDRES, S. & HARTMANN, G. 2002. Quantitative expression of toll-like receptor 1–10 mRNA in cellular subsets of human peripheral blood mononuclear cells and sensitivity to CpG oligodeoxynucleotides. *The Journal of Immunology*, 168, 4531-4537.
- HOVAV, A.-H. 2017. Human γδ T cells: rapid, stable and clonally reactive. *Cellular and Molecular Immunology*, 14, 646.
- HUANG, H.-Y. & LUTHER, S. A. 2012. Expression and function of interleukin-7 in secondary and tertiary lymphoid organs. *Seminars in Immunology*. Elsevier, 175-189.
- IKE, K., UCHIDA, Y., NAKAMURA, T. & IMAI, S. 2005. Induction of interferongamma (IFN-gamma) and T helper 1 (Th1) immune response by bitter gourd extract. *Journal of Veterinary Medical Science*, 67, 521-4.
- IMANISHI, T., HARA, H., SUZUKI, S., SUZUKI, N., AKIRA, S. & SAITO, T. 2007. Cutting edge: TLR2 directly triggers Th1 effector functions. *The Journal of Immunology*, 178, 6715-6719.
- ISHIDA, W., FUKUDA, K., SUMI, T., EBIHARA, N., KAJISAKO, M., MATSUDA, H., YAGITA, H. & FUKUSHIMA, A. 2011. Adjuvants determine the contribution of basophils to antigen sensitization *in vivo*. *Immunology Letters*, 136, 49-54.
- JI, W., LI, J. & LIU, J. 2012. A Method for Measuring Gene Copy Number in Biological Samples without Using Control Samples of Known Copies. *The Journal of Molecular Diagnostics*, 14, 280-285.
- JIN, H., LUO, Q., ZHENG, Y., NURAHMAT, M., WU, J., LI, B., LV, Y., WANG, G., DUAN, X. & DONG, J. 2013. CD4+ CD25+ Foxp3+ T cells contribute to the antiasthmatic effects of *Astragalus membranaceus* extract in a rat model of asthma. *International Immunopharmacology*, 15, 42-49.
- JUNGI, T. W., FARHAT, K., BURGENER, I. A. & WERLING, D. 2011. Toll-like receptors in domestic animals. *Cell and Tissue Research*, 343, 107-120.
- JUTILA, M. A., BARGATZE, R. F., KURK, S., WARNOCK, R. A., EHSANI, N., WATSON, S. R. & WALCHECK, B. 1994. Cell surface P-and E-selectin support shear-dependent rolling of bovine gamma/delta T cells. *The Journal of Immunology*, 153, 3917-3928.
- KABELITZ, D. 2007. Expression and function of Toll-like receptors in T lymphocytes. *Current Opinion in Immunology*, 19, 39-45.

- KAISHO, T. & AKIRA, S. 2002. Toll-like receptors as adjuvant receptors. Biochimica et Biophysica Acta (BBA)-Molecular Cell Research, 1589, 1-13.
- KALESNIKOFF, J. & GALLI, S. J. 2008. New developments in mast cell biology. *Nature Immunology*, 9, 1215-23.
- KARAN, D., KRIEG, A. M. & LUBAROFF, D. M. 2007. Paradoxical enhancement of CD8 T cell-dependent anti-tumor protection despite reduced CD8 T cell responses with addition of a TLR9 agonist to a tumor vaccine. *International Journal of Cancer*, 121, 1520-1528.
- KARTHIKEYAN, A., RADHIKA, G., ARAVINDHAKSHAN, T. & ANILKUMAR, K. 2016. Expression Profiling of Innate Immune Genes in Milk Somatic Cells During Subclinical Mastitis in Crossbred Dairy Cows. *Animal Biotechnology*, 27, 303-309.
- KASSIOTIS, G. & LISTON, A. 2011. *Regulatory T Cells: Methods and Protocols*, Humana Press.
- KATO-MAEDA, M., SHANLEY, C. A., ACKART, D., JARLSBERG, L. G., SHANG, S., OBREGON-HENAO, A., HARTON, M., BASARABA, R. J., HENAO-TAMAYO, M. & BARROZO, J. C. 2012. Beijing sublineages of *Mycobacterium tuberculosis* differ in pathogenicity in the guinea pig. *Clinical and Vaccine Immunology*, 19, 1227-1237.
- KAWAI, T. & AKIRA, S. 2006. TLR signaling. *Cell Death and Differentiation*, 13, 816.
- KAWAI, T. & AKIRA, S. 2007. TLR signaling. Seminars in Immunology. Elsevier, 24-32.
- KAWAI, T. & AKIRA, S. 2010. The role of pattern-recognition receptors in innate immunity: update on Toll-like receptors. *Nature Immunology*, 11, 373-384.
- KAWASAKI, T. & KAWAI, T. 2014. Toll-like receptor signaling pathways. *Frontiers in Immunology*, 5, 1-8.
- KELLEY, T. & PARKER, C. 2010. CD4 (+) CD25 (+) Foxp3 (+) regulatory T cells and hematologic malignancies. *Frontiers in bioscience (Scholar edition)*, 2, 980-992.
- KELLY, B. & O'NEILL, L. A. 2015. Metabolic reprogramming in macrophages and dendritic cells in innate immunity. *Cell Research*, 25, 771.

- KENNEDY, H. E., WELSH, M. D., BRYSON, D. G., CASSIDY, J. P., FORSTER, F. I., HOWARD, C. J., COLLINS, R. A. & POLLOCK, J. M. 2002. Modulation of immune responses to *Mycobacterium bovis* in cattle depleted of WC1+ γδ T cells. *Infection and Immunity*, 70, 1488-1500.
- KERSSE, K., BERTRAND, M. J., LAMKANFI, M. & VANDENABEELE, P. 2011. NODlike receptors and the innate immune system: coping with danger, damage and death. *Cytokine & Growth Factor Reviews*, 22, 257-276.
- KIM, Y. M., LEE, J. A., JUNG, B. G., KIM, T. H., LEE, B. J. & SUH, G. H. 2016. Reference ranges of hematology and lymphocyte subsets in healthy Korean native cattle (Hanwoo) and Holstein dairy cattle. *Animal Science Journal*. 87, 796-801.
- KIM, Y.-M., BRINKMANN, M. M., PAQUET, M.-E. & PLOEGH, H. L. 2008. UNC93B1 delivers nucleotide-sensing toll-like receptors to endolysosomes. *Nature*, 452, 234-238.
- KIMURA, K., GOFF, J., KEHRLI JR, M. & HARP, J. 1999. Phenotype analysis of peripheral blood mononuclear cells in periparturient dairy cows. *Journal of Dairy Science*, 82, 315-319.
- KOBLANSKY, A. A., JANKOVIC, D., OH, H., HIENY, S., SUNGNAK, W., MATHUR, R., HAYDEN, M. S., AKIRA, S., SHER, A. & GHOSH, S. 2013. Recognition of Profilin by Toll-like Receptor 12 Is Critical for Host Resistance to *Toxoplasma gondii. Immunity*, 38, 119-130.
- KOETS, A., SANTEMA, W., MERTENS, H., OOSTENRIJK, D., KEESTRA, M., OVERDIJK, M., LABOURIAU, R., FRANKEN, P., FRIJTERS, A. & NIELEN, M. 2010. Susceptibility to paratuberculosis infection in cattle is associated with single nucleotide polymorphisms in Toll-like receptor 2 which modulate immune responses against Mycobacterium avium subspecies paratuberculosis. *Preventive Veterinary Medicine*, 93, 305-315.
- KOK, M. G., HALLIANI, A., MOERLAND, P. D., MEIJERS, J. C., CREEMERS, E. E. & PINTO-SIETSMA, S.-J. 2015. Normalization panels for the reliable quantification of circulating microRNAs by RT-qPCR. *The FASEB Journal*, 29, 3853-3862.
- KOMAI-KOMA, M., GILCHRIST, D. S. & XU, D. 2009. Direct recognition of LPS by human but not murine CD8+ T cells via TLR4 complex. *European Journal* of Immunology, 39, 1564-1572.
- KOMAI-KOMA, M., JONES, L., OGG, G. S., XU, D. & LIEW, F. Y. 2004. TLR2 is expressed on activated T cells as a costimulatory receptor. *Proceedings of the National Academy of Sciences of the United States of America*, 101, 3029-3034.

- KORETH, J., MATSUOKA, K.-I., KIM, H. T., MCDONOUGH, S. M., BINDRA, B., ALYEA III, E. P., ARMAND, P., CUTLER, C., HO, V. T. & TREISTER, N. S. 2011. Interleukin-2 and regulatory T cells in graft-versus-host disease. New England Journal of Medicine, 365, 2055-2066.
- KRONSTEINER, B., BASSAGANYA-RIERA, J., PHILIPSON, C., VILADOMIU, M., CARBO, A., PEDRAGOSA, M., VENTO, S. & HONTECILLAS, R. 2013. *Helicobacter pylori* infection in a pig model is dominated by Th1 and cytotoxic CD8+ T cell responses. *Infection and Immunity*, 81, 3803-13.
- KULKARNI, R. R., VILLANUEVA, A. I., ELAWADLI, I., JAYANTH, P., READ, L. R., MANSOUR HAERYFAR, S. & SHARIF, S. 2012. Costimulatory activation of murine invariant natural killer T cells by toll-like receptor agonists. *Cellular Immunology*, 277, 33-43.
- KUMAR, H. & BOT, A. 2013. Innate Immune Recognition Mechanisms and Translational Opportunities. *International Reviews of Immunology*, 32, 113-115.
- KUMAR, H., KAWAI, T. & AKIRA, S. 2009. Toll-like receptors and innate immunity. *Biochemical and Biophysical Research Communications*, 388, 621-625.
- LACETERA, N., BERNABUCCI, U., SCALIA, D., BASIRICÒ, L., MORERA, P. & NARDONE, A. 2006. Heat stress elicits different responses in peripheral blood mononuclear cells from Brown Swiss and Holstein cows. *Journal of Dairy Science*, 89, 4606-4612.
- LANIER, L. L. 2013. Shades of grey the blurring view of innate and adaptive immunity. *Nature Reviews Immunology*, 13, 73-74.
- LANKESTER, F., LUGELO, A., WERLING, D., MNYAMBWA, N., KEYYU, J., KAZWALA, R., GRANT, D., SMITH, S., PARAMESWARAN, N., CLEAVELAND, S., RUSSELL, G. & HAIG, D. 2016. The efficacy of alcelaphine herpesvirus-1 (AIHV-1) immunization with the adjuvants Emulsigen® and the monomeric TLR5 ligand FliC in zebu cattle against AIHV-1 malignant catarrhal fever induced by experimental virus challenge. *Veterinary Microbiology*, 195, 144-153.
- LARSSON, J. 2010. Inflammatory cytokines induced by Bovine Viral Diarrhoea Virus (BVDV) in Peripheral Blood Mononuclear Cell (PBMC) subsets. PhD Thesis, Sveriges Lantbruksuniversitet (SLU) / Uppsala University / Sweden.
- LATZ, E., VERMA, A., VISINTIN, A., GONG, M., SIROIS, C. M., KLEIN, D. C., MONKS, B. G., MCKNIGHT, C. J., LAMPHIER, M. S. & DUPREX, W. P. 2007. Ligand-induced conformational changes allosterically activate Tolllike receptor 9. *Nature Immunology*, *8*, 772.

- LAWAND, M., DÉCHANET-MERVILLE, J. & DIEU-NOSJEAN, M.-C. 2017. Key Features of Gamma-Delta T-Cell Subsets in Human Diseases and Their Immunotherapeutic Implications. *Frontiers in Immunology*, 8, 1-9.
- LEE, C. W., CHUNG, S. W., BAE, M. J., SONG, S., KIM, S.-P. & KIM, K. 2015. Peptidoglycan up-regulates CXCL8 expression via multiple pathways in monocytes/macrophages. *Biomolecules & Therapeutics*, 23, 564.
- LEE, S. M., KOK, K.-H., JAUME, M., CHEUNG, T. K., YIP, T.-F., LAI, J. C., GUAN, Y., WEBSTER, R. G., JIN, D.-Y. & PEIRIS, J. M. 2014. Toll-like receptor 10 is involved in induction of innate immune responses to influenza virus infection. *Proceedings of the National Academy of Sciences*, 111, 3793-3798.
- LEE, S.-R., PHARR, G., BOYD, B. & PINCHUK, L. 2008. Bovine viral diarrhea viruses modulate toll-like receptors, cytokines and co-stimulatory molecules genes expression in bovine peripheral blood monocytes. *Comparative Immunology, Microbiology and Infectious Diseases*, 31, 403-418.
- LEE, Y. K., TURNER, H., MAYNARD, C. L., OLIVER, J. R., CHEN, D., ELSON, C. O. & WEAVER, C. T. 2009. Late developmental plasticity in the T helper 17 lineage. *Immunity*, 30, 92-107.
- LEYSEN, S., VANDERKELEN, L., WEEKS, S., MICHIELS, C. & STRELKOV, S. 2013. Structural basis of bacterial defense against g-type lysozyme-based innate immunity. *Cellular and Molecular Life Sciences*, 70, 1113-1122.
- LI, H., LEBEDEVA, M. I., LLERA, A. S., FIELDS, B. A., BRENNER, M. B. & MARIUZZA, R. A. 1998. Structure of the Vδ domain of a human γδ T-cell antigen receptor. *Nature*, 391, 502-506.
- LIAUDET, L., DEB, A., PACHER, P., MABLEY, J. G., MURTHY, K., SALZMAN, A. L. & SZABÓ, C. 2002. The flagellin-TLR5 axis: therapeutic opportunities. *Drug News & Perspectives*, 15, 397-409.
- LIU, G. E., HOU, Y., ZHU, B., CARDONE, M. F., JIANG, L., CELLAMARE, A., MITRA, A., ALEXANDER, L. J., COUTINHO, L. L. & DELL'AQUILA, M. E. 2010. Analysis of copy number variations among diverse cattle breeds. *Genome Research*, 20, 693-703.
- LIU, L., BOTOS, I., WANG, Y., LEONARD, J. N., SHILOACH, J., SEGAL, D. M. & DAVIES, D. R. 2008. Structural basis of toll-like receptor 3 signaling with double-stranded RNA. *Science*, 320, 379-381.
- LIU, W., PUTNAM, A. L., XU-YU, Z., SZOT, G. L., LEE, M. R., ZHU, S., GOTTLIEB, P. A., KAPRANOV, P., GINGERAS, T. R. & GROTH, B. F. D. S. 2006.

CD127 expression inversely correlates with FoxP3 and suppressive function of human CD4+ Treg cells. *Journal of Experimental Medicine*, 203, 1701-1711.

- LLOP-GUEVARA, A., MARCINKO, J., FATTOUH, R. & JORDANA, M. 2011. Interleukin 3, Interleukin 5, and Granulocyte–Macrophage Colony-Stimulating Factor. *Inflammation and Allergy Drug Design*, 14, 187-196.
- LLOYD, C. M. & SAGLANI, S. 2013. Eosinophils in the Spotlight: Finding the link between obesity and asthma. *Nature medicine*, 19, 976-977.
- LONG, E. O., SIK KIM, H., LIU, D., PETERSON, M. E. & RAJAGOPALAN, S. 2013a. Controlling natural killer cell responses: integration of signals for activation and inhibition. *Annual Review of Immunology*, 31, 227-258.
- LONG, S., PENG, Q., DICARLO, J., WANG, Y. & LADER, E. 2013b. Multicopy reference assay (MRef)—a superior normalizer of sample input in DNA copy number analysis. A report published by QIAGEN Sciences, Inc. 6951 Executive Way, Frederick, Maryland 21703, USA.
- LONZA.COM. 2017. *PBMCs The One Stop Immune Cell Shop.* [online] Available at: http://www.lonza.com/products-services/bio-research/primarycells/hematopoietic-cells/hematopoietic-knowledge-center/pbmcs.aspx [Accessed 22-08-2017].
- LOO, Y.-M. & GALE JR, M. 2011. Immune signaling by RIG-I-like receptors. *Immunity*, 34, 680-692.
- LUND, F. E. & RANDALL, T. D. 2010. Effector and regulatory B cells: modulators of CD4+ T cell immunity. *Nature Reviews Immunology*, 10, 236-247.
- LUO, J., OBMOLOVA, G., MALIA, T. J., WU, S.-J., DUFFY, K. E., MARION, J. D., BELL, J. K., GE, P., ZHOU, Z. H. & TEPLYAKOV, A. 2012. Lateral clustering of TLR3: dsRNA signaling units revealed by TLR3ecd: 3Fabs quaternary structure. *Journal of Molecular Biology*, 421, 112-124.
- LÜTTGENAU, J., HERZOG, K., STRÜVE, K., LATTER, S., BOOS, A., BRUCKMAIER, R. M., BOLLWEIN, H. & KOWALEWSKI, M. P. 2016. LPS-mediated effects and spatio-temporal expression of TLR2 and TLR4 in the bovine corpus luteum. *Reproduction*, 151, 391-399.
- LUZYANINA, T., MRUSEK, S., EDWARDS, J. T., ROOSE, D., EHL, S. & BOCHAROV, G. 2007. Computational analysis of CFSE proliferation assay. *Journal of Mathematical Biology*, 54, 57-89.

- LYONS, A. B., HASBOLD, J. & HODGKIN, P. D. 2001. Flow cytometric analysis of cell division history using dilution of carboxyfluorescein diacetate succinimidyl ester, a stably integrated fluorescent probe. *Methods in Cell Biology*, 63, 375-398.
- MA, L. & CHUNG, W. K. 2014. Quantitative Analysis of Copy Number Variants Based on Real-Time LightCycler PCR. *Current Protocols in Human Genetics*, 7.21. 1-7.21. 8.
- MACHUGH, N., MBURU, J., CAROL, M., WYATT, C., ORDEN, J. & DAVIS, W. 1997. Identification of two distinct subsets of bovine  $\gamma\delta$  T cells with unique cell surface phenotype and tissue distribution. *Immunology*, 92, 340-345.
- MAINOU-FOWLER, T., CRAIG, V., COPPLESTONE, J., HAMON, M. & PRENTICE, A. 1994. Interleukin-5 (IL-5) increases spontaneous apoptosis of B-cell chronic lymphocytic leukemia cells *in vitro* independently of bcl-2 expression and is inhibited by IL-4. Blood, 84, 2297-2304.
- MAN, S. M. & KANNEGANTI, T.-D. 2016. Converging roles of caspases in inflammasome activation, cell death and innate immunity. *Nature Reviews Immunology*, 16, 7.
- MANDRAJU, R., MURRAY, S., FORMAN, J. & PASARE, C. 2014. Differential ability of surface and endosomal TLRs to induce CD8 T cell responses *in vivo*. *The Journal of Immunology*, 192, 4303-4315.
- MANTOVANI, A., CASSATELLA, M. A., COSTANTINI, C. & JAILLON, S. 2011. Neutrophils in the activation and regulation of innate and adaptive immunity. *Nature Reviews Immunology*, 11, 519-531.
- MAPARA, M., THOMAS, B. S. & BHAT, K. 2012. Rabbit as an animal model for experimental research. *Dental Research Journal*, 9, 111.
- MARCONDES, M. C. G., SPINA, C., BUSTAMANTE, E. & FOX, H. 2012. Increased Toll-Like Receptor Signaling Pathways Characterize CD8. *BioMed Research International*, Volume 2013, Article ID 796014, 1-7.
- MAREK-TRZONKOWSKA, N., MYŚLIWIEC, M., DOBYSZUK, A., GRABOWSKA, M., DERKOWSKA, I., JUŚCIŃSKA, J., OWCZUK, R., SZADKOWSKA, A., WITKOWSKI, P. & MŁYNARSKI, W. 2014. Therapy of type 1 diabetes with CD4+ CD25 high CD127-regulatory T cells prolongs survival of pancreatic islets—results of one year follow-up. *Clinical Immunology*, 153, 23-30.
- MARIOTTI, M., WILLIAMS, J., DUNNER, S., VALENTINI, A. & PARISET, L. 2009. Polymorphisms within the Toll-Like Receptor (TLR)-2, -4, and -6 Genes in Cattle. *Diversity*, 1, 7.

- MARTINS, T. M., MUNIZ, C. S., ANDRADE, V. B., PAIXÃO, T. A., SANTOS, R. L. & BORGES, Á. M. 2016. Changes in endometrial transcription of TLR2, TLR4, and CD14 during the first-week postpartum in dairy cows with retained placenta. *Theriogenology*, 85, 1282-1288.
- MASON, P., LOMBARDI, G. & LECHLER, R. 1991. NK cells inhibit T-cell responses: LFA3+ but not LFA3-T-cell responses are suppressed. *Immunology*, 73, 444.
- MASOPUST, D., VEZYS, V., MARZO, A. L. & LEFRANÇOIS, L. 2001. Preferential localization of effector memory cells in nonlymphoid tissue. *Science*, 291, 2413-2417.
- MATSUGUCHI, T., TAKAGI, K., MUSIKACHAROEN, T. & YOSHIKAI, Y. 2000. Gene expressions of lipopolysaccharide receptors, toll-like receptors 2 and 4, are differently regulated in mouse T lymphocytes. *Blood*, 95, 1378-1385.
- MATSUSHIMA, N., TANAKA, T., ENKHBAYAR, P., MIKAMI, T., TAGA, M., YAMADA, K. & KUROKI, Y. 2007. Comparative sequence analysis of leucine-rich repeats (LRRs) within vertebrate toll-like receptors. *BMC Genomics*, 8, 124.
- MAURI, C. & BOSMA, A. 2012. Immune regulatory function of B cells. *Annual Review of Immunology*, 30, 221-241.
- MAYO, P., HARTSHORNE, T., LI, K., MCMUNN-GIBSON, C., SPENCER, K. & SCHNETZ-BOUTAUD, N. 2010. CNV analysis using TaqMan copy number assays. *Current Protocols in Human Genetics*, Chapter 2, Unit 2.13, 67, 2.13.1-2.13.10.
- MCCARRON, M. & REEN, D. J. 2009. Activated human neonatal CD8+ T cells are subject to immunomodulation by direct TLR2 or TLR5 stimulation. *The Journal of Immunology*, 182, 55-62.
- MCGUIRE, K., JONES, M., WERLING, D., WILLIAMS, J., GLASS, E. & JANN, O. 2006. Radiation hybrid mapping of all 10 characterized bovine Toll-like receptors. *Animal Genetics*, 37, 47-50.
- MEADE, K. G., O'GORMAN, G. M., NARCIANDI, F., MACHUGH, D. E. & O'FARRELLY, C. 2012. Functional characterisation of bovine interleukin 8 promoter haplotypes *in vitro*. *Molecular Immunology*, 50, 108-116.
- MEANS, T. K., WANG, S., LIEN, E., YOSHIMURA, A., GOLENBOCK, D. T. & FENTON, M. J. 1999. Human toll-like receptors mediate cellular activation by *Mycobacterium tuberculosis*. *The Journal of Immunology*, 163, 3920-3927.

- MEDZHITOV, R. 2001. Toll-like receptors and innate immunity. *Nature Reviews Immunology*, 1, 135-45.
- MEDZHITOV, R., PRESTON-HURLBURT, P., KOPP, E., STADLEN, A., CHEN, C., GHOSH, S. & JANEWAY JR, C. A. 1998. MyD88 is an adaptor protein in the hToll/IL-1 receptor family signaling pathways. *Molecular Cell*, 2, 253-258.
- MELIEF, C. J. 2013. "License to kill" reflects joint action of CD4 and CD8 T cells. *Clinical Cancer Research*, 19, 4295-4296.
- MENG, X.-M., NIKOLIC-PATERSON, D. J. & LAN, H. Y. 2016. TGF-β: the master regulator of fibrosis. *Nature Reviews Nephrology*, 12, 325-338.
- MENZIES, M. & INGHAM, A. 2006. Identification and expression of Toll-like receptors 1–10 in selected bovine and ovine tissues. *Veterinary Immunology and Immunopathology*, 109, 23-30.
- METCALFE, H. J., LA RAGIONE, R. M., SMITH, D. G. E. & WERLING, D. 2014. Functional characterisation of bovine TLR5 indicates species-specific recognition of flagellin. *Veterinary Immunology and Immunopathology*, 157, 197-205.
- METCALFE, H., BEST, A., KANELLOS, T., LA RAGIONE, R. & WERLING, D. 2010. Flagellin expression enhances Salmonella accumulation in TLR5-positive macrophages. *Developmental & Comparative Immunology*, 34, 797-804.
- MICHAEL, R. 1995. The B-cell antigen receptor complex and co-receptors. *Immunology Today*, 16, 310-313.
- MILLER, L. S., SØRENSEN, O. E., LIU, P. T., JALIAN, H. R., ESHTIAGHPOUR, D., BEHMANESH, B. E., CHUNG, W., STARNER, T. D., KIM, J. & SIELING, P.
  A. 2005. TGF-a regulates TLR expression and function on epidermal keratinocytes. *The Journal of Immunology*, 174, 6137-6143.
- MILLS, K. H. 2011. TLR-dependent T cell activation in autoimmunity. *Nature Reviews Immunology*, 11, 807.
- MINEO, T. W., OLIVEIRA, C. J., GUTIERREZ, F. R. & SILVA, J. S. 2010. Recognition by Toll-like receptor 2 induces antigen-presenting cell activation and Th1 programming during infection by *Neospora caninum*. *Immunology and Cell Biology*, 88, 825.
- MINTON, K. 2014. Microbiota: A'natural'vaccine adjuvant. *Nature Reviews Immunology*, 14, 650-651.

- MISHRA, C., KUMAR, S., YATHISH, H. M., CHOUDHARY, R. & KUMAR, A. 2016. Association of TLR4 gene variants with milk production traits in crossbred cattle. *Agri Gene*, 1, 69-71.
- MITA, Y., DOBASHI, K., ENDOU, K., KAWATA, T., SHIMIZU, Y., NAKAZAWA, T. & MORI, M. 2002. Toll-like receptor 4 surface expression on human monocytes and B cells is modulated by IL-2 and IL-4. *Immunology Letters*, 81, 71-75.
- MOLFINO, N., GOSSAGE, D., KOLBECK, R., PARKER, J. & GEBA, G. 2012. Molecular and clinical rationale for therapeutic targeting of interleukin-5 and its receptor. *Clinical & Experimental Allergy*, 42, 712-737.
- MOOKHERJEE, N., WILSON, H. L., DORIA, S., POPOWYCH, Y., FALSAFI, R., YU, J. J., LI, Y., VEATCH, S., ROCHE, F. M. & BROWN, K. L. 2006. Bovine and human cathelicidin cationic host defense peptides similarly suppress transcriptional responses to bacterial lipopolysaccharide. *Journal of Leukocyte Biology*, 80, 1563-1574.
- MORADI, B., SCHNATZER, P., HAGMANN, S., ROSSHIRT, N., GOTTERBARM, T., KRETZER, J. P., THOMSEN, M., LORENZ, H.-M., ZEIFANG, F. & TRETTER, T. 2014. CD4+ CD25+/high CD127 low/-regulatory T cells are enriched in rheumatoid arthritis and osteoarthritis joints—analysis of frequency and phenotype in synovial membrane, synovial fluid and peripheral blood. Arthritis Research & Therapy, 16, R97.
- MORRISSETTE, N., GOLD, E. & ADEREM, A. 1999. The macrophage--a cell for all seasons. *Trends in Cell Biology*, 9, 199-201.
- MOTTET, C. & GOLSHAYAN, D. 2007. CD4+ CD25+ Foxp3+ regulatory T cells: from basic research to potential therapeutic use. *Swiss Medical Weekly*, 137, 625-634.
- MOVITA, D., KREEFFT, K., BIESTA, P., VAN OUDENAREN, A., LEENEN, P. J., JANSSEN, H. L. & BOONSTRA, A. 2012. Kupffer cells express a unique combination of phenotypic and functional characteristics compared with splenic and peritoneal macrophages. *Journal of Leukocyte Biology*, 92, 723-733.
- MOYNAGH, P. N. 2005. TLR signalling and activation of IRFs: revisiting old friends from the NF-κB pathway. *Trends in Immunology*, 26, 469-476.
- MUNDER, M., MALLO, M., EICHMANN, K. & MODOLELL, M. 1998. Murine macrophages secrete interferon γ upon combined stimulation with interleukin (IL)-12 and IL-18: a novel pathway of autocrine macrophage activation. *Journal of Experimental Medicine*, 187, 2103-2108.

- MURAI, M., TUROVSKAYA, O., KIM, G., MADAN, R., KARP, C. L., CHEROUTRE, H. & KRONENBERG, M. 2009. Interleukin 10 acts on regulatory T cells to maintain expression of the transcription factor Foxp3 and suppressive function in mice with colitis. *Nature Immunology*, 10, 1178-1184.
- NADELLA, V., WANG, V., GRIFFIN, M. & DEVITT, A. 2012. Characterization of tissue transglutaminase 2 in macrophage function. EMBO Conference on Cellular Signaling & Molecular Medicine, 2012.
- NELMS, K., KEEGAN, A. D., ZAMORANO, J., RYAN, J. J. & PAUL, W. E. 1999. The IL-4 receptor: signaling mechanisms and biologic functions. *Annual Review of Immunology*, 17, 701-738.
- NELSON, C. D. 2010. Vitamin D signlaing in the bovine immune system. Graduate Theses and Dissertations. Paper 11517, Iowa State University. USA.
- NISHIOKA, T., SHIMIZU, J., IIDA, R., YAMAZAKI, S. & SAKAGUCHI, S. 2006. CD4+ CD25+ Foxp3+ T cells and CD4+ CD25- Foxp3+ T cells in aged mice. *The Journal of Immunology*, 176, 6586-6593.
- NYLÉN, S., MAURYA, R., EIDSMO, L., MANANDHAR, K. D., SUNDAR, S. & SACKS, D. 2007. Splenic accumulation of IL-10 mRNA in T cells distinct from CD4+ CD25+ (Foxp3) regulatory T cells in human visceral leishmaniasis. Journal of Experimental Medicine, 204, 805-817.
- O'BRIEN, R. L., ROARK, C. L., JIN, N., KEMAL AYDINTUG, M., FRENCH, J. D., CHAIN, J. L., WANDS, J., JOHNSTON, M. & BORN, W. K. 2007. γδ T-cell receptors: functional correlations. *Immunological Reviews*, 215, 77-88.
- OHASHI, K., BURKART, V., FLOHÉ, S. & KOLB, H. 2000. Cutting edge: heat shock protein 60 is a putative endogenous ligand of the toll-like receptor-4 complex. *The Journal of Immunology*, 164, 558-561.
- OHTO, U., SHIBATA, T., TANJI, H., ISHIDA, H., KRAYUKHINA, E., UCHIYAMA, S., MIYAKE, K. & SHIMIZU, T. 2015. Structural basis of CpG and inhibitory DNA recognition by Toll-like receptor 9. *Nature*, 520, 702.
- OLIVEIRA, B. R. S. M., VIEIRA, F. V., DE S. VIEIRA, D., DA SILVA, S. E. L., GAMEIRO, R., FLORES, E. F. & CARDOSO, T. C. 2017. Expression of miR-155 associated with Toll-like receptors 3, 7, and 9 transcription in the olfactory bulbs of cattle naturally infected with BHV5. *Journal of Neurovirology*, 23(5), 772-778.
- OLIVIA, S. H. 2012. Characterisation of toll like receptor 8 as a new innate immune receptor for Mycobacterium tuberculosis. . PhD Thesis.

Department of Microbiology, National University of Singapore (NUS), Singapore.

- O'NEILL, L. A. & BOWIE, A. G. 2007. The family of five: TIR-domain-containing adaptors in Toll-like receptor signalling. *Nature Reviews Immunology*, 7, 353-364.
- OPSAL, M. A., VÅGE, D. I., HAYES, B., BERGET, I. & LIEN, S. 2006. Genomic organization and transcript profiling of the bovine toll-like receptor gene cluster TLR6-TLR1-TLR10. *Gene*, 384, 45-50.
- OUELLETTE, A. J. 2011. Paneth cell a-defensins in enteric innate immunity. *Cellular and Molecular Life Sciences,* 68, 2215-2229.
- OUYANG, W., KOLLS, J. K. & ZHENG, Y. 2008. The biological functions of T helper 17 cell effector cytokines in inflammation. *Immunity*, 28, 454-467.
- PACHECO, G. V., NOH, I. B. N., RUBÍ, M.-H., RAMÍREZ, A. V. A., VILLANUEVA, R. F. L., ORTIZ, I. G. Q., SALOMÓN, L. G. A., RUZ, N. P. & CÁRDENAS, N. A. R. 2016. Expression of TLR-7, MyD88, NF-kB, and INF-a in B Lymphocytes of Mayan Women with Systemic Lupus Erythematosus in Mexico. *Frontiers in Immunology*, 7, 23.
- PALM, N. W. & MEDZHITOV, R. 2009. Pattern recognition receptors and control of adaptive immunity. *Immunological Reviews*, 227, 221-233.
- PALTI, Y. 2011. Toll-like receptors in bony fish: from genomics to function. Developmental & Comparative Immunology, 35, 1263-1272.
- PANDEY, S., KAWAI, T. & AKIRA, S. 2015. Microbial sensing by Toll-like receptors and intracellular nucleic acid sensors. *Cold Spring Harbor Perspectives in Biology*, 7, a016246.
- PARAMESWARAN, N., RUSSELL, G. C., BARTLEY, K., GRANT, D. M., DEANE, D., TODD, H., DAGLEISH, M. P. & HAIG, D. M. 2014. The effect of the TLR9 ligand CpG-oligodeoxynucleotide on the protective immune response to alcelaphine herpesvirus-1-mediated malignant catarrhal fever in cattle. *Veterinary Research*, 45, 59.
- PASARE, C. & MEDZHITOV, R. 2005. Toll-like receptors: linking innate and adaptive immunity. *Mechanisms of Lymphocyte Activation and Immune Regulation X.* Springer.
- PECKHAM, R., BRILL, R., FOSTER, D. S., BOWEN, A., LEIGH, J. A., COFFEY, T. J. & FLYNN, R. J. 2014. Two distinct populations of bovine IL-17+ T-cells

can be induced and WC1+ IL-17+  $\gamma\delta$  T-cells are effective killers of protozoan parasites. *Scientific Reports*, 4, 5431.

- PENG, G., GUO, Z., KINIWA, Y., SHIN VOO, K., PENG, W., FU, T., WANG, D. Y., LI, Y., WANG, H. Y. & WANG, R.-F. 2005. Toll-like receptor 8-mediated reversal of CD4+ regulatory T cell function. *Science*, 309, 1380-1384.
- PENG, G., WANG, H. Y., PENG, W., KINIWA, Y., SEO, K. H. & WANG, R.-F. 2007. Tumor-infiltrating  $\gamma\delta$  T cells suppress T and dendritic cell function via mechanisms controlled by a unique toll-like receptor signaling pathway. *Immunity*, 27, 334-348.
- PENG, X., KNOUSE, J. A. & HERNON, K. M. 2015. Rabbit models for studying human infectious diseases. *Comparative Medicine*, 65, 499-507.
- PESENACKER, A. M., BENDING, D., URSU, S., WU, Q., NISTALA, K. & WEDDERBURN, L. R. 2013. CD161 defines the subset of FoxP3+ T cells capable of producing proinflammatory cytokines. *Blood*, 121, 2647-2658.
- PETERSON, L. W. & ARTIS, D. 2014. Intestinal epithelial cells: regulators of barrier function and immune homeostasis. *Nature Reviews Immunology*, 14, 141.
- PFAFFL, M. W. & HAGELEIT, M. 2001. Validities of mRNA quantification using recombinant RNA and recombinant DNA external calibration curves in real-time RT-PCR. *Biotechnology Letters*, 23, 275-282.
- PFAFFL, M. W. 2001. A new mathematical model for relative quantification in real-time RT–PCR. *Nucleic Acids Research*, 29, e45-e45.
- PFAFFL, M. W., HORGAN, G. W. & DEMPFLE, L. 2002. Relative expression software tool (REST©) for group-wise comparison and statistical analysis of relative expression results in real-time PCR. *Nucleic Acids Research*, 30, e36-e36.
- PHILPOTT, D. J. & GIRARDIN, S. E. 2004. The role of Toll-like receptors and Nod proteins in bacterial infection. *Molecular Immunology*, 41, 1099-1108.
- PICCIRILLO, C. A. & SHEVACH, E. M. 2001. Cutting edge: control of CD8+ T cell activation by CD4+ CD25+ immunoregulatory cells. *The Journal of Immunology*, 167, 1137-1140.
- PIETRETTI, D. & WIEGERTJES, G. F. 2014. Ligand specificities of Toll-like receptors in fish: indications from infection studies. *Developmental & Comparative Immunology*, 43, 205-222.

- PIETSCHMANN, K., BEETZ, S., WELTE, S., MARTENS, I., GRUEN, J., OBERG, H. H., WESCH, D. & KABELITZ, D. 2009. Toll-Like Receptor Expression and Function in Subsets of Human γδ T Lymphocytes. *Scandinavian Journal of Immunology*, 70, 245-255.
- PINHEIRO, L. B., COLEMAN, V. A., HINDSON, C. M., HERRMANN, J., HINDSON, B. J., BHAT, S. & EMSLIE, K. R. 2011. Evaluation of a droplet digital polymerase chain reaction format for DNA copy number quantification. *Analytical Chemistry*, 84, 1003-1011.
- PLATTNER, B., HUFFMAN, E. & HOSTETTER, J. 2013. Gamma-delta T-cell responses during subcutaneous *Mycobacterium avium* subspecies *paratuberculosis* challenge in sensitized or naive calves using matrix biopolymers. *Veterinary Pathology*, 50, 630-637.
- PODACK, E. R. & DENNERT, G. 1983. Assembly of two types of tubules with putative cytolytic function by cloned natural killer cells. *Nature*, 302, 442–445.
- PORCHERIE, A., CUNHA, P., TROTEREAU, A., ROUSSEL, P., GILBERT, F. B., RAINARD, P. & GERMON, P. 2012. Repertoire of Escherichia coli agonists sensed by innate immunity receptors of the bovine udder and mammary epithelial cells. *Veterinary Research*, 43, 14.
- PORTOU, M., BAKER, D., ABRAHAM, D. & TSUI, J. 2015. The innate immune system, toll-like receptors and dermal wound healing: A review. *Vascular Pharmacology*, 71, 31-36.
- PRICE, J. C., BROMFIELD, J. J. & SHELDON, I. M. 2013. Pathogen-associated molecular patterns initiate inflammation and perturb the endocrine function of bovine granulosa cells from ovarian dominant follicles via TLR2 and TLR4 pathways. *Endocrinology*, 154, 3377-3386.
- PRICE, S. J. & HOPE, J. C. 2009. Enhanced secretion of interferon- $\gamma$  by bovine  $\gamma\delta$  T cells induced by coculture with *Mycobacterium bovis*-infected dendritic cells: evidence for reciprocal activating signals. *Immunology*, 126, 201-208.
- QIAGEN-WEBINAR-TEAM. 2015. qPCR technology overview, applications, data analysis and service solutions. An online webinar organised by Qiagen Company. Four parts qPCR Webinar Series: Part 1: "Introduction to Real Time PCR (Q-PCR/qPCR/qRT-PCR)" on 5th of October 2015. Part 2: "Advanced Real-Time PCR Array Technology – Coding & Noncoding RNA Expression Analysis" on 12th of October 2015. Part 3: "PCR Array Data Analysis Tutorial" on 19th of October 2015. Part 4: "Accelerate your discovery with QIAGEN service solutions for biomarker research" on 26th of October 2015. [Online]. Available: https://www.qiagen.com/gb/qdm/qfdk/webinar-series-qpcr/ [Accessed 26-10-2015].

- QUAH, B. J. & PARISH, C. R. 2010. The use of carboxyfluorescein diacetate succinimidyl ester (CFSE) to monitor lymphocyte proliferation. *Journal of Visualized Experiments*, 44, 2259.
- QUAH, B. J. & PARISH, C. R. 2012. New and improved methods for measuring lymphocyte proliferation *in vitro* and *in vivo* using CFSE-like fluorescent dyes. *Journal of Immunological Methods*, 379, 1-14.
- RATHINAM, V. A., JIANG, Z., WAGGONER, S. N., SHARMA, S., COLE, L. E., WAGGONER, L., VANAJA, S. K., MONKS, B. G., GANESAN, S. & LATZ, E. 2010. The AIM2 inflammasome is essential for host defense against cytosolic bacteria and DNA viruses. *Nature Immunology*, 11, 395-402.
- REN, C., ZHANG, Q., DE HAAN, B. J., ZHANG, H., FAAS, M. M. & DE VOS, P. 2016. Identification of TLR2/TLR6 signalling lactic acid bacteria for supporting immune regulation. *Scientific Reports*, 6, 34561.
- RENSETTI, D., MARIN, M., QUINTANA, S., MORÁN, P., VERNA, A., ODEÓN, A. & PÉREZ, S. 2016. Involvement of toll-like receptors 3 and 7/8 in the neuropathogenesis of bovine herpesvirus types 1 and 5. *Research in Veterinary Science*, 107, 1-7.
- REUVEN, E. M., FINK, A. & SHAI, Y. 2014. Regulation of innate immune responses by transmembrane interactions: lessons from the TLR family. *Biochimica et Biophysica Acta*, 1838, 1586-93.
- REYNOLDS, J. M. & DONG, C. 2013. Toll-like receptor regulation of effector T lymphocyte function. *Trends in Immunology*, 34, 511-519.
- REYNOLDS, J. M., MARTINEZ, G. J., CHUNG, Y. & DONG, C. 2012. Toll-like receptor 4 signaling in T cells promotes autoimmune inflammation. *Proceedings of the National Academy of Sciences*, 109, 13064-13069.
- REYNOLDS, J. M., PAPPU, B. P., PENG, J., MARTINEZ, G. J., ZHANG, Y., CHUNG, Y., MA, L., YANG, X. O., NURIEVA, R. I. & TIAN, Q. 2010. Toll-like receptor 2 signaling in CD4+ T lymphocytes promotes T helper 17 responses and regulates the pathogenesis of autoimmune disease. *Immunity*, 32, 692-702.
- RHEE, J. H., LEE, S. E. & KIM, S. Y. 2012. Modified flagellin improved toll-like receptor 5 stimulating activity. Google Patents.
- RITTERSHAUS, C. W., MILLER, D. P., THOMAS, L. J., PICARD, M. D., HONAN, C. M., EMMETT, C. D., PETTEY, C. L., ADARI, H., HAMMOND, R. A. & BEATTIE, D. T. 2000. Vaccine-induced antibodies inhibit CETP activity in vivo and reduce aortic lesions in a rabbit model of atherosclerosis. *Arteriosclerosis, Thrombosis, and Vascular Biology*, 20, 2106-2112.

- ROCK, F. L., HARDIMAN, G., TIMANS, J. C., KASTELEIN, R. A. & BAZAN, J. F. 1998. A family of human receptors structurally related to *Drosophila* Toll. *Proceedings of the National Academy of Sciences*, 95, 588-593.
- ROGERS, A. N., VANBUREN, D. G., HEDBLOM, E. E., TILAHUN, M. E., TELFER, J. C. & BALDWIN, C. L. 2005. γδ T cell function varies with the expressed WC1 coreceptor. *The Journal of Immunology*, 174, 3386-3393.
- ROGERS, A. N., VANBUREN, D. G., ZOU, B., LAHMERS, K. K., HERZIG, C. T., BROWN, W. C., TELFER, J. C. & BALDWIN, C. L. 2006. Characterization of WC1 co-receptors on functionally distinct subpopulations of ruminant γδ T cells. *Cellular Immunology*, 239, 151-161.
- RONCADOR, G., BROWN, P. J., MAESTRE, L., HUE, S., MARTÍNEZ-TORRECUADRADA, J. L., LING, K. L., PRATAP, S., TOMS, C., FOX, B. C. & CERUNDOLO, V. 2005. Analysis of FOXP3 protein expression in human CD4+ CD25+ regulatory T cells at the single-cell level. *European Journal* of Immunology, 35, 1681-1691.
- ROOS, A., XU, W., CASTELLANO, G., NAUTA, A. J., GARRED, P., DAHA, M. R. & VAN KOOTEN, C. 2004. Mini-review: A pivotal role for innate immunity in the clearance of apoptotic cells. *European Journal of Immunology*, 34, 921-929.
- ROUND, J. L. & MAZMANIAN, S. K. 2010. Inducible Foxp3+ regulatory T-cell development by a commensal bacterium of the intestinal microbiota. *Proceedings of the National Academy of Sciences*, 107, 12204-12209.
- RUDRA, D., CHAUDHRY, A., NIEC, R. E., ARVEY, A., SAMSTEIN, R. M., LESLIE, C., SHAFFER, S. A., GOODLETT, D. R. & RUDENSKY, A. Y. 2012. Transcription factor Foxp3 and its protein partners form a complex regulatory network. *Nature Immunology*, 13, 1010-1019.
- RUIZ-LARRANAGA, O., MANZANO, C., IRIONDO, M., GARRIDO, J. M., MOLINA, E., VAZQUEZ, P., JUSTE, R. A. & ESTONBA, A. 2011. Genetic variation of toll-like receptor genes and infection by Mycobacterium avium ssp. paratuberculosis in Holstein-Friesian cattle. J Dairy Sci, 94, 3635-41.
- RUS, H., CUDRICI, C. & NICULESCU, F. 2005. The role of the complement system in innate immunity. *Immunologic Research*, 33, 103-112.
- RUSSELL, C. D. 2012. *Single nucleotide polymorphisms in bovine chemokine and toll-like receptors; impacts on disease susceptibility and productivity in dairy cattle. PhD Thesis, The University of Nottingham.*

- RUSSELL, C. D., WIDDISON, S., LEIGH, J. A. & COFFEY, T. J. 2012. Identification of single nucleotide polymorphisms in the bovine Toll-like receptor 1 gene and association with health traits in cattle. *Veterinary Research*, 43, 17.
- SABAT, R., GRÜTZ, G., WARSZAWSKA, K., KIRSCH, S., WITTE, E., WOLK, K. & GEGINAT, J. 2010. Biology of interleukin-10. *Cytokine & Growth Factor Reviews*, 21, 331-344.
- SABROE, I., DOWER, S. K. & WHYTE, M. K. 2005. The role of Toll-like receptors in the regulation of neutrophil migration, activation, and apoptosis. *Clinical Infectious Diseases*, 41, S421-S426.
- SAIKH, K. U., LEE, J. S., KISSNER, T. L., DYAS, B. & ULRICH, R. G. 2003. Tolllike receptor and cytokine expression patterns of CD56+ T cells are similar to natural killer cells in response to infection with Venezuelan equine encephalitis virus replicons. *Journal of Infectious Diseases*, 188, 1562-1570.
- SAKAGUCHI, S. 2005. Naturally arising Foxp3-expressing CD25+ CD4+ regulatory T cells in immunological tolerance to self and non-self. *Nature Immunology*, 6, 345-352.
- SAKAGUCHI, S., SAKAGUCHI, N., ASANO, M., ITOH, M. & TODA, M. 1995.
   Immunologic self-tolerance maintained by activated T cells expressing IL-2 receptor alpha-chains (CD25). Breakdown of a single mechanism of self-tolerance causes various autoimmune diseases. *The Journal of Immunology*, 155, 1151-1164.
- SAKAGUCHI, S., WING, K., ONISHI, Y., PRIETO-MARTIN, P. & YAMAGUCHI, T. 2009. Regulatory T cells: how do they suppress immune responses? *International Immunology*, 21, 1105-1111.
- SAKAGUCHI, S., YAMAGUCHI, T., NOMURA, T. & ONO, M. 2008. Regulatory T cells and immune tolerance. *Cell*, 133, 775-787.
- SALLUSTO, F., LENIG, D., FÖRSTER, R., LIPP, M. & LANZAVECCHIA, A. 1999. Two subsets of memory T lymphocytes with distinct homing potentials and effector functions. *Nature*, 401, 708-712.
- SALVI, V., VAIRA, X., GIANELLO, V., VERMI, W., BUGATTI, M., SOZZANI, S. & BOSISIO, D. 2016. TLR Signalling Pathways Diverge in Their Ability to Induce PGE2. *Mediators of Inflammation*, Article ID 5678046, 1-10.
- SARAIVA, M. & O'GARRA, A. 2010. The regulation of IL-10 production by immune cells. *Nature Reviews Immunology*, 10, 170.

- SATTLER, S., GHADIALLY, H. & HOFER, E. 2012. Evolution of the C-type lectinlike receptor genes of the DECTIN-1 cluster in the NK gene complex. *The Scientific World Journal,* Article ID 931386, 1-11.
- SCHAUT, R. G., RIDPATH, J. F. & SACCO, R. E. 2016. Bovine Viral Diarrhea Virus Type 2 Impairs Macrophage Responsiveness to Toll-Like Receptor Ligation with the Exception of Toll-Like Receptor 7. *PLoS One*, 11, e0159491.
- SCHELLER, J., CHALARIS, A., SCHMIDT-ARRAS, D. & ROSE-JOHN, S. 2011. The pro-and anti-inflammatory properties of the cytokine interleukin-6. *Biochimica et Biophysica Acta (BBA)-Molecular Cell Research*, 1813, 878-888.
- SCHMITTGEN, T. D. & LIVAK, K. J. 2008. Analyzing real-time PCR data by the comparative CT method. *Nature Protocols*, **3**, 1101-1108.
- SCHNEBERGER, D., LEWIS, D., CALDWELL, S. & SINGH, B. 2011. Expression of toll-like receptor 9 in lungs of pigs, dogs and cattle. *International Journal* of Experimental Pathology, 92, 1-7.
- SCHNUPF, P. & SANSONETTI, P. J. 2012. Quantitative RT-PCR profiling of the rabbit immune response: assessment of acute *Shigella flexneri* infection. *PLoS One*, 7, e36446.
- SCHOENBRUNN, A., FRENTSCH, M., KOHLER, S., KEYE, J., DOOMS, H., MOEWES, B., DONG, J., LODDENKEMPER, C., SIEPER, J., WU, P., ROMAGNANI, C., MATZMOHR, N. & THIEL, A. 2012. A converse 4-1BB and CD40 ligand expression pattern delineates activated regulatory T cells (Treg) and conventional T cells enabling direct isolation of alloantigen-reactive natural Foxp3+ Treg. *Journal of Immunology*, 189, 5985-94.
- SCHÖLCH, S., RAUBER, C., WEITZ, J., KOCH, M. & HUBER, P. E. 2015. TLR activation and ionizing radiation induce strong immune responses against multiple tumor entities. *Onco-Immunology*, 4, e1042201.
- SCHROEDER JR, H. W. & CAVACINI, L. 2010. Structure and function of immunoglobulins. *Journal of Allergy and Clinical Immunology*, 125, S41-S52.
- SEABURY, C. M. & WOMACK, J. E. 2008. Analysis of sequence variability and protein domain architectures for bovine peptidoglycan recognition protein 1 and Toll-like receptors 2 and 6. *Genomics*, 92, 235-245.
- SEABURY, C. M., CARGILL, E. J. & WOMACK, J. E. 2007. Sequence variability and protein domain architectures for bovine Toll-like receptors 1, 5, and 10. *Genomics*, 90, 502-15.
- SEABURY, C. M., SEABURY, P. M., DECKER, J. E., SCHNABEL, R. D., TAYLOR, J. F. & WOMACK, J. E. 2010. Diversity and evolution of 11 innate immune genes in *Bos taurus taurus* and *Bos taurus indicus* cattle. *Proceedings of the National Academy of Sciences*, 107, 151-156.
- SEPULCRE, M. P., LÓPEZ-MUÑOZ, A., ANGOSTO, D., GARCÍA-ALCAZAR, A., MESEGUER, J. & MULERO, V. 2011. TLR agonists extend the functional lifespan of professional phagocytic granulocytes in the bony fish gilthead seabream and direct precursor differentiation towards the production of granulocytes. *Molecular Immunology*, 48, 846-859.
- SHANG, S., HARTON, M., TAMAYO, M. H., SHANLEY, C., PALANISAMY, G. S., CARAWAY, M., CHAN, E. D., BASARABA, R. J., ORME, I. M. & ORDWAY, D.
  J. 2011. Increased Foxp3 expression in guinea pigs infected with W-Beijing strains of *M. tuberculosis*. *Tuberculosis*, 91, 378-385.
- SHEIKH, A. A., AGGARWAL, A. & AARIF, O. 2016. Effect of in vitro zinc supplementation on HSPs expression and Interleukin 10 production in heat treated peripheral blood mononuclear cells of transition Sahiwal and Karan Fries cows. *Journal of Thermal Biology*, 56, 68-76.
- SHEVACH, E. M. 2002. CD4+ CD25+ suppressor T cells: more questions than answers. *Nature Reviews Immunology*, 2, 389-400.
- SHI, F.-D., LJUNGGREN, H.-G., LA CAVA, A. & VAN KAER, L. 2011. Organspecific features of natural killer cells. *Nature Reviews Immunology*, 11, 658-671.
- SHI, Z., CAI, Z., YU, J., ZHANG, T., ZHAO, S., SMEDS, E., ZHANG, Q., WANG, F., ZHAO, C. & FU, S. 2012. Toll-like receptor 11 (TLR11) prevents Salmonella penetration into the murine Peyer patches. *Journal of Biological Chemistry*, 287, 43417-43423.
- SHIMIZU, H., MATSUGUCHI, T., FUKUDA, Y., NAKANO, I., HAYAKAWA, T., TAKEUCHI, O., AKIRA, S., UMEMURA, M., SUDA, T. & YOSHIKAI, Y. 2002. Toll-like receptor 2 contributes to liver injury by *Salmonella* infection through Fas ligand expression on NKT cells in mice. *Gastroenterology*, 123, 1265-1277.
- SHIMIZU, T., KAWASAKI, Y., AOKI, Y., MAGATA, F., KAWASHIMA, C. & MIYAMOTO, A. 2017. Effect of Single Nucleotide Polymorphisms of Toll-Like Receptor 4 (TLR 4) on Reproductive Performance and Immune Function in Dairy Cows. *Biochemical Genetics*, 55, 212-222.
- SHIMIZU, T., KIMURA, Y., KIDA, Y., KUWANO, K., TACHIBANA, M., HASHINO, M.
   & WATARAI, M. 2014. Cytadherence of *Mycoplasma pneumoniae* Induces Inflammatory Responses through Autophagy and Toll-like Receptor 4. *Infection and Immunity*, 82, 3076 – 3086.

- SHIN, S. J., CHANG, C.-F., CHANG, C.-D., MCDONOUGH, S. P., THOMPSON, B., YOO, H. S. & CHANG, Y.-F. 2005. *In vitro* cellular immune responses to recombinant antigens of *Mycobacterium avium* subsp. *paratuberculosis*. *Infection and immunity*, 73, 5074-5085.
- SILVA-SANTOS, B., SERRE, K. & NORELL, H. 2015. γδ T cells in cancer. *Nature reviews Immunology*, 15, 683.
- SILVER, R. & CURLEY, J. P. 2013. Mast cells on the mind: new insights and opportunities. *Trends in Neurosciences*, 36, 513-521.
- SINGH, L. V., SAXENA, S., GUPTA, S., GUPTA, S. K., RAVI KUMAR, G., DESAI, G., SAHOO, A. P. & TIWARI, A. 2014. Evaluation and comparison of the constitutive expression levels of Toll-like receptors 2, 3 and 7 in the peripheral blood mononuclear cells of Tharparkar and crossbred cattle. *Veterinary World*, 7, 209-212.
- SINGH, R. & SINGH, A. P. 2013. Advances in Innate Immunity. *Physiology and Nutri-Genomics*, 204-208.
- SIRACUSA, M. C., COMEAU, M. R. & ARTIS, D. 2011. New insights into basophil biology: initiators, regulators, and effectors of type 2 inflammation. Annals of the New York Academy of Sciences, 1217, 166-77.
- SKEEN, M. J. & ZIEGLER, H. K. 1995. Activation of gamma delta T cells for production of IFN-gamma is mediated by bacteria via macrophagederived cytokines IL-1 and IL-12. *The Journal of Immunology*, 154, 5832-5841.
- SKEVAKI, C., PARARAS, M., KOSTELIDOU, K., TSAKRIS, A. & ROUTSIAS, J. 2015. Single nucleotide polymorphisms of Toll-like receptors and susceptibility to infectious diseases. *Clinical & Experimental Immunology*, 180, 165-177.
- SMILEY, S. T., KING, J. A. & HANCOCK, W. W. 2001. Fibrinogen stimulates macrophage chemokine secretion through toll-like receptor 4. *The Journal* of *Immunology*, 167, 2887-2894.
- SMITH, S. 2012. Genetic Variation in Ruminant Toll-like Receptors 2 and 5. PhD Thesis, The University of Nottingham, UK.
- SMITH, S. A., JANN, O. C., HAIG, D., RUSSELL, G. C., WERLING, D., GLASS, E. J. & EMES, R. D. 2012. Adaptive evolution of Toll-like receptor 5 in domesticated mammals. *BMC Evolutionary Biology*, 12, 122.

- SONG, Y., SUN, L., GUO, A. & YANG, L. 2014. Toll-like receptor 6 gene polymorphisms increase the risk of bovine tuberculosis in Chinese Holstein cattle. *Acta Histochemica*, 116, 1159-62.
- SOPP, P. & HOWARD, C. 2001. IFNy and IL-4 production by CD4, CD8 and WC1  $\gamma\delta$  TCR+ T cells from cattle lymph nodes and blood. *Veterinary Immunology and Immunopathology*, 81, 85-96.
- STABEL, J. R., KIMURA, K. & ROBBE-AUSTERMAN, S. 2007. Augmentation of secreted and intracellular gamma interferon following johnin purified protein derivative sensitization of cows naturally infected with *Mycobacterium avium* subsp. *paratuberculosis*. *Journal of Veterinary Diagnostic Investigation*, 19, 43-51.
- STARY, G., BANGERT, C., TAUBER, M., STROHAL, R., KOPP, T. & STINGL, G. 2007. Tumoricidal activity of TLR7/8-activated inflammatory dendritic cells. *Journal of Experimental Medicine*, 204, 1441-1451.
- STEINBACH, S., VORDERMEIER, H. M. & JONES, G. J. 2016. CD4+ and γδ T Cells are the main Producers of IL-22 and IL-17A in Lymphocytes from *Mycobacterium bovis*-infected Cattle. *Scientific Reports*, 6, 29990.
- SUN, L., SONG, Y., RIAZ, H., YANG, H., HUA, G., GUO, A. & YANG, L. 2012. Polymorphisms in toll-like receptor 1 and 9 genes and their association with tuberculosis susceptibility in Chinese Holstein cattle. *Veterinary Immunology and Immunopathology*, 147, 195-201.
- SUTMULLER, R. P., MORGAN, M. E., NETEA, M. G., GRAUER, O. & ADEMA, G. J. 2006. Toll-like receptors on regulatory T cells: expanding immune regulation. *Trends in Immunology*, 27, 387-393.
- SVINGEN, T., LETTING, H., HADRUP, N., HASS, U. & VINGGAARD, A. M. 2015. Selection of reference genes for quantitative RT-PCR (RT-qPCR) analysis of rat tissues under physiological and toxicological conditions. *PeerJ*, 3, e855.
- SWAIN, D. K., KUMAR, J., YADAV, S., SINGH, S. K., SINGH, Y. & DANG, A. K. 2017. The functional dynamics of neutrophils during different seasons in zebu cattle. *Biological Rhythm Research*, 48, 227-237.
- TABIASCO, J., DEVÊVRE, E., RUFER, N., SALAUN, B., CEROTTINI, J.-C., SPEISER, D. & ROMERO, P. 2006. Human effector CD8+ T lymphocytes express TLR3 as a functional coreceptor. *The Journal of Immunology*, 177, 8708-8713.
- TAHOUN, A., JENSEN, K., CORRIPIO-MIYAR, Y., MCATEER, S. P., CORBISHLEY, A., MAHAJAN, A., BROWN, H., FREW, D., AUMEUNIER, A., SMITH, D. G.,

MCNEILLY, T. N., GLASS, E. J. & GALLY, D. L. 2015. Functional analysis of bovine TLR5 and association with IgA responses of cattle following systemic immunisation with H7 flagella. *Veterinary Research*, 46, 9.

- TAKAHASHI, T., KUNIYASU, Y., TODA, M., SAKAGUCHI, N., ITOH, M., IWATA, M., SHIMIZU, J. & SAKAGUCHI, S. 1998. Immunologic self-tolerance maintained by CD25+ CD4+ naturally anergic and suppressive T cells: induction of autoimmune disease by breaking their anergic/suppressive state. *International Immunology*, 10, 1969-1980.
- TAKAHASHI, T., TAGAMI, T., YAMAZAKI, S., UEDE, T., SHIMIZU, J., SAKAGUCHI, N., MAK, T. W. & SAKAGUCHI, S. 2000. Immunologic self-tolerance maintained by CD25+ CD4+ regulatory T cells constitutively expressing cytotoxic T lymphocyte–associated antigen 4. *The Journal of experimental medicine*, 192, 303-310.
- TAKEBAYASHI, K., HOKARI, R., KURIHARA, C., OKADA, Y., OKUDAIRA, K., MATSUNAGA, H., KOMOTO, S., WATANABE, C., KAWAGUCHI, A. & NAGAO, S. 2009. Oral Tolerance Induced by Enterobacteria Altered the Process of Lymphocyte Recruitment to Intestinal Microvessels: Roles of Endothelial Cell Adhesion Molecules, TGF-beta and Negative Regulators of TLR Signaling. *Microcirculation*, 16, 251-264.
- TAKEDA, K. & AKIRA, S. 2004. TLR signaling pathways. *Seminars in Immunology, Elsevier*, 16, 3-9.
- TAKESHITA, F., LEIFER, C. A., GURSEL, I., ISHII, K. J., TAKESHITA, S., GURSEL, M. & KLINMAN, D. M. 2001. Cutting edge: role of Toll-like receptor 9 in CpG DNA-induced activation of human cells. *The Journal of Immunology*, 167, 3555-3558.
- TAKEUCHI, O. & AKIRA, S. 2010. Pattern recognition receptors and inflammation. *Cell*, 140, 805-820.
- TAKEUCHI, O., KAWAI, T., MUHLRADT, P. F., MORR, M., RADOLF, J. D., ZYCHLINSKY, A., TAKEDA, K. & AKIRA, S. 2001. Discrimination of bacterial lipoproteins by Toll-like receptor 6. *International Immunology*, 13, 933-40.
- TAKEUCHI, O., SATO, S., HORIUCHI, T., HOSHINO, K., TAKEDA, K., DONG, Z., MODLIN, R. L. & AKIRA, S. 2002. Cutting edge: role of Toll-like receptor 1 in mediating immune response to microbial lipoproteins. *The Journal of Immunology*, 169, 10-14.
- TANIGUCHI, T. & TAKAOKA, A. 2001. A weak signal for strong responses: interferon-alpha/beta revisited. *Nature Reviews Molecular Cell Biology*, 2, 378-386.

- TANJI, H., OHTO, U., SHIBATA, T., MIYAKE, K. & SHIMIZU, T. 2013. Structural reorganization of the Toll-like receptor 8 dimer induced by agonistic ligands. *Science*, 339, 1426-1429.
- TARTEY, S. & TAKEUCHI, O. 2014. Toll-Like Receptors: Role in Inflammation and Cancer. *Cancer and Inflammation Mechanisms: Chemical, Biological, and Clinical Aspects*, 83-102.
- TAUBERT, A., BEHRENDT, J. H., SÜHWOLD, A., ZAHNER, H. & HERMOSILLA, C. 2009. Monocyte-and macrophage-mediated immune reactions against *Eimeria bovis*. *Veterinary Parasitology*, 164, 141-153.
- TAYLOR, S., WAKEM, M., DIJKMAN, G., ALSARRAJ, M. & NGUYEN, M. 2010. A practical approach to RT-qPCR—publishing data that conform to the MIQE guidelines. *Methods*, 50, S1-S5.
- TEIJARO, J. R., WALSH, K. B., RICE, S., ROSEN, H. & OLDSTONE, M. B. 2014. Mapping the innate signaling cascade essential for cytokine storm during influenza virus infection. *Proceedings of the National Academy of Sciences*, 111, 3799-3804.
- TELFER, J. C. & BALDWIN, C. L. 2015. Bovine gamma delta T cells and the function of gamma delta T cell specific WC1 co-receptors. *Cellular immunology*, 296, 76-86.
- THORNTON, B. & BASU, C. 2011. Real-time PCR (qPCR) primer design using free online software. *Biochemistry and Molecular Biology Education*, 39, 145-154.
- TOCRIS-BIOSCIENCE. 2017. Tocris Bioscience. (2017). Toll-like Receptors | TLR | Tocris Bioscience. [online] Available at: https://www.tocris.com/pharmacology/toll-like-receptors [Accessed 1 Sep. 2017]. [Online].
- TOKA, F. N., KENNEY, M. A. & GOLDE, W. T. 2011. Rapid and transient activation of  $\gamma\delta$  T cells to IFN- $\gamma$  production, NK cell-like killing, and antigen processing during acute virus infection. *The Journal of Immunology*, 186, 4853-4861.
- TOMBÁCZ, K., MWANGI, D., WERLING, D. & GIBSON, A. J. 2017. Comparison of cellular assays for TLR activation and development of a species-specific reporter cell line for cattle. *Innate Immunity*, 23, 329-335.
- TOMLJENOVIC, L. & SHAW, C. A. 2011. Aluminum vaccine adjuvants: are they safe? *Current Medicinal Chemistry*, 18, 2630-2637.

- TOMLJENOVIC, L. & SHAW, C. A. 2012. Mechanisms of aluminum adjuvant toxicity and autoimmunity in pediatric populations. *Lupus*, 21, 223-230.
- TORNATORE, L., THOTAKURA, A. K., BENNETT, J., MORETTI, M. & FRANZOSO, G. 2012. The nuclear factor kappa B signaling pathway: integrating metabolism with inflammation. *Trends in Cell Biology*, 22, 557-566.
- TRAN, D. Q., RAMSEY, H. & SHEVACH, E. M. 2007. Induction of FOXP3 expression in naive human CD4+ FOXP3- T cells by T-cell receptor stimulation is transforming growth factor-β-dependent but does not confer a regulatory phenotype. *Blood*, 110, 2983-2990.
- TRIANTAFILOU, M., GAMPER, F. G., HASTON, R. M., MOURATIS, M. A., MORATH, S., HARTUNG, T. & TRIANTAFILOU, K. 2006. Membrane sorting of toll-like receptor (TLR)-2/6 and TLR2/1 heterodimers at the cell surface determines heterotypic associations with CD36 and intracellular targeting. *Journal of Biological Chemistry*, 281, 31002-31011.
- TSCHOPP, J. & NABHOLZ, M. 1990. Perforin-mediated target cell lysis by cytolytic T lymphocytes. *Annual Review of Immunology*, 8, 279-302.
- TSENOVA, L., HARBACHEUSKI, R., MOREIRA, A. L., ELLISON, E., DALEMANS, W., ALDERSON, M. R., MATHEMA, B., REED, S. G., SKEIKY, Y. A. & KAPLAN, G. 2006. Evaluation of the Mtb72F polyprotein vaccine in a rabbit model of tuberculous meningitis. *Infection and Immunity*, 74, 2392-2401.
- TURIN, L. & RIVA, F. 2008. Toll-like receptor family in domestic animal species. *Critical Reviews™ in Immunology*, 28, 513-38.
- TURNER, M. L., CRONIN, J. G., HEALEY, G. D. & SHELDON, I. M. 2014. Epithelial and stromal cells of bovine endometrium have roles in innate immunity and initiate inflammatory responses to bacterial lipopeptides *in vitro* via Toll-like receptors TLR2, TLR1, and TLR6. *Endocrinology*, 155, 1453-65.
- VAN DER VLUGT, L. E., HAEBERLEIN, S., DE GRAAF, W., MARTHA, T. E. & SMITS, H. H. 2014. Toll-like receptor ligation for the induction of regulatory B cells. *Regulatory B Cells.* Springer.
- VAN DYKEN, S. J. & LOCKSLEY, R. M. 2013. Interleukin-4-and interleukin-13mediated alternatively activated macrophages: roles in homeostasis and disease. Annual Review of Immunology, 31, 317.
- VAN HAREN, S. D., DOWLING, D. J., FOPPEN, W., CHRISTENSEN, D., ANDERSEN, P., REED, S. G., HERSHBERG, R. M., BADEN, L. R. & LEVY, O. 2016. Age-Specific Adjuvant Synergy: Dual TLR7/8 and Mincle Activation

of Human Newborn Dendritic Cells Enables Th1 Polarization. *The Journal of Immunology*, 197, 4413-4424.

- VAN LEEUWEN, A. M., POELHUIS-LETH, D. J. & BLADH, M. L. 2013. Davis's Comprehensive Handbook of Laboratory & Diagnostic Tests With Nursing Implications, FA Davis.
- VAN PANHUYS, N., TANG, S.-C., PROUT, M., CAMBERIS, M., SCARLETT, D., ROBERTS, J., HU-LI, J., PAUL, W. E. & LE GROS, G. 2008. *In vivo* studies fail to reveal a role for IL-4 or STAT6 signaling in Th2 lymphocyte differentiation. *Proceedings of the National Academy of Sciences*, 105, 12423-12428.
- VANDERS, R. L., MURPHY, V. E., GIBSON, P. G., HANSBRO, P. M. & WARK, P. A. 2015. CD8 T cells and dendritic cells: key players in the attenuated maternal immune response to influenza infection. *Journal of Reproductive Immunology*, 107, 1-9.
- VENUGOPAL, P. G., NUTMAN, T. B. & SEMNANI, R. T. 2009. Activation and regulation of toll-like receptors (TLRs) by helminth parasites. *Immunologic Research*, 43, 252-263.
- VESOSKY, B., TURNER, O., TURNER, J. & ORME, I. 2004. Gamma interferon production by bovine  $\gamma\delta$  T cells following stimulation with mycobacterial mycolylarabinogalactan peptidoglycan. *Infection and Immunity*, 72, 4612-4618.
- VIALLARD, J. F., PELLEGRIN, J. L., RANCHIN, V., SCHAEVERBEKE, T., DEHAIS, J., LONGY-BOURSIER, M., RAGNAUD, J. M., LENG, B. & MOREAU, J. F. 1999. Th1 (IL-2, interferon-gamma (IFN-gamma)) and Th2 (IL-10, IL-4) cytokine production by peripheral blood mononuclear cells (PBMC) from patients with systemic lupus erythematosus (SLE). *Clinical and Experimental Immunology*, 115, 189-95.
- VIGNALI, D. A., COLLISON, L. W. & WORKMAN, C. J. 2008. How regulatory T cells work. *Nature Reviews Immunology*, 8, 523-532.
- VISAN, I. 2014. Unstable Foxp3. Nature Immunology, 15, 14-14.
- VIVIER, E., RAULET, D. H., MORETTA, A., CALIGIURI, M. A., ZITVOGEL, L., LANIER, L. L., YOKOYAMA, W. M. & UGOLINI, S. 2011. Innate or adaptive immunity? The example of natural killer cells. *Science*, 331, 44-49.

VOEHRINGER, D. 2013. Regulation of Type 2 Immunity by Basophils, Springer.

- VON BOEHMER, H. & DANIEL, C. 2013. Therapeutic opportunities for manipulating TReg cells in autoimmunity and cancer. *Nature Reviews Drug Discovery*, 12, 51.
- VORDERMEIER, H. M., CHAMBERS, M. A., COCKLE, P. J., WHELAN, A. O., SIMMONS, J. & HEWINSON, R. G. 2002. Correlation of ESAT-6-specific gamma interferon production with pathology in cattle following *Mycobacterium bovis* BCG vaccination against experimental bovine tuberculosis. *Infection and Immunity*, 70, 3026-3032.
- WAIGHT, J. D., TAKAI, S., MARELLI, B., QIN, G., HANCE, K. W., ZHANG, D., TIGHE, R., LAN, Y., LO, K.-M. & SABZEVARI, H. 2015. Cutting edge: epigenetic regulation of Foxp3 defines a stable population of CD4+ regulatory T cells in tumors from mice and humans. *The Journal of Immunology*, 194, 878-882.
- WALDVOGEL, A., HEDIGER-WEITHALER, B., EICHER, R., ZAKHER, A., ZARLENGA, D., GASBARRE, L. & HEUSSLER, V. 2000. Interferon-γ and interleukin-4 mRNA expression by peripheral blood mononuclear cells from pregnant and non-pregnant cattle seropositive for bovine viral diarrhea virus. *Veterinary Immunology and Immunopathology*, 77, 201-212.
- WALTER, G. J., FLESKENS, V., FREDERIKSEN, K. S., RAJASEKHAR, M., MENON, B., GERWIEN, J. G., EVANS, H. G. & TAAMS, L. S. 2016. Phenotypic, Functional, and Gene Expression Profiling of Peripheral CD45RA+ and CD45RO+ CD4+ CD25+ CD127low Treg Cells in Patients With Chronic Rheumatoid Arthritis. *Arthritis & Rheumatology*, 68, 103-116.
- WANG, C., DENG, L., HONG, M., AKKARAJU, G. R., INOUE, J.-I. & CHEN, Z. J. 2001. TAK1 is a ubiquitin-dependent kinase of MKK and IKK. *Nature*, 412, 346-351.
- WANG, H., DANIEL, V., SADEGHI, M. & OPELZ, G. 2013. Differences in the Induction of Induced Human CD4<sup>+</sup>CD25<sup>+</sup> FoxP3<sup>+</sup> T-Regulatory Cells and CD3<sup>+</sup> CD8<sup>+</sup> CD28<sup>-</sup>T-Suppressor Cells Subset Phenotypes In Vitro: Comparison of Phorbol 12-Myristate 13-Acetate/Ionomycin and Phytohemagglutinin Stimulation. Transplantation Proceedings, 45, 1822-1831.
- WANG, H., LI, L., ZHANG, Y., PAN, S.-C., CHEN, A.-Q. & QIAN, W.-D. 2015. Expression and significance of CD4+ CD25+ CD127-regulatory T cells in peripheral blood of patients with different phenotypes of Guillain-Barré syndrome. *International Journal of Clinical and Experimental Medicine*, 8, 19126-19131.
- WANG, H.-X., LIU, M., WENG, S.-Y., LI, J.-J., XIE, C., HE, H.-L., GUAN, W., YUAN, Y.-S. & GAO, J. 2012. Immune mechanisms of Concanavalin A

model of autoimmune hepatitis. *World Journal of Gastroenterology*, 18, 119-125.

- WANG, W., HU, X., SHEN, P., ZHANG, N. & FU, Y. 2017. Sodium houttuyfonate inhibits LPS-induced inflammatory response via suppressing TLR4/NF-κB signaling pathway in bovine mammary epithelial cells. *Microbial Pathogenesis*, 107, 12-16.
- WANG, Y., LIU, S., LI, Y., WANG, Q., SHAO, J., CHEN, Y. & XIN, J. 2016.
   Mycoplasma bovis-derived lipid-associated membrane proteins activate
   IL-1beta production through the NF-kappaB pathway via toll-like receptor
   2 and MyD88. Developmental & Comparative Immunology, 55, 111-8.
- WEAVER, S., DUBE, S., MIR, A., QIN, J., SUN, G., RAMAKRISHNAN, R., JONES, R. C. & LIVAK, K. J. 2010. Taking qPCR to a higher level: analysis of CNV reveals the power of high throughput qPCR to enhance quantitative resolution. *Methods*, 50, 271-276.
- WEI, T., GONG, J., JAMITZKY, F., HECKL, W. M., STARK, R. W. & RÖSSLE, S. C. 2009. Homology modeling of human Toll-like receptors TLR7, 8, and 9 ligand-binding domains. *Protein Science*, 18, 1684-1691.
- WEILL, J.-C., LE GALLOU, S., HAO, Y. & REYNAUD, C.-A. 2013. Multiple players in mouse B cell memory. *Current Opinion in Immunology*, 25, 334-338.
- WERLING, D. & JUNGI, T. W. 2003. TOLL-like receptors linking innate and adaptive immune response. *Veterinary Immunology and Immunopathology*, 91, 1-12.
- WERLING, D. 2012. The Importance of TLRs [Online]. Bio-Rad Company. Available: https://www.bio-rad-antibodies.com/cow-bovine-toll-likereceptors-review.html. [Accessed 3-4-2018].
- WERLING, D., HOPE, J. C., HOWARD, C. J. & JUNGI, T. W. 2004. Differential production of cytokines, reactive oxygen and nitrogen by bovine macrophages and dendritic cells stimulated with Toll-like receptor agonists. *Immunology*, 111, 41-52.
- WERLING, D., HOPE, J. C., SIDDIQUI, N., WIDDISON, S., RUSSELL, C., SOPP, P. & COFFEY, T. J. 2017. Subset-specific expression of Toll-like receptors by Bovine afferent lymph Dendritic cells. *Frontiers in Veterinary Science*, Volume 4, Article 44, 1-6.
- WERLING, D., JANN, O. C., OFFORD, V., GLASS, E. J. & COFFEY, T. J. 2009. Variation matters: TLR structure and species-specific pathogen recognition. *Trends in Immunology*, 30, 124-130.

- WERLING, D., PIERCY, J. & COFFEY, T. J. 2006. Expression of TOLL-like receptors (TLR) by bovine antigen-presenting cells—Potential role in pathogen discrimination? *Veterinary Immunology and Immunopathology*, 112, 2-11.
- WESCH, D., GLATZEL, A. & KABELITZ, D. 2001. Differentiation of resting human peripheral blood γδ T cells toward Th1-or Th2-phenotype. *Cellular Immunology*, 212, 110-117.
- WHELAN, M. C., CASEY, G., LARKIN, J. O., GUINN, B.-A., O'SULLIVAN, G. C. & TANGNEY, M. 2014. Oral Tolerance to Cancer Can Be Abrogated by T Regulatory Cell Inhibition. *PloS One*, 9, e97602.
- WHERRY, E. J. & AHMED, R. 2004. Memory CD8 T-cell differentiation during viral infection. *Journal of Virology*, 78, 5535-5545.
- WHITE, S. N., KATA, S. R. & WOMACK, J. E. 2003. Comparative fine maps of bovine toll-like receptor 4 and toll-like receptor 2 regions. *Mammalian Genome*, 14, 149-55.
- WHITSETT, J. A. & ALENGHAT, T. 2015. Respiratory epithelial cells orchestrate pulmonary innate immunity. *Nature Immunology*, 16, 27-35.
- WIDDISON, S., WATSON, M., PIERCY, J., HOWARD, C. & COFFEY, T. J. 2008. Granulocyte chemotactic properties of *M. tuberculosis* versus *M. bovis*infected bovine alveolar macrophages. *Molecular Immunology*, 45, 740-749.
- WIJNGAARD, P., METZELAAR, M., MACHUGH, N., MORRISON, W. & CLEVERS, H. 1992. Molecular characterization of the WC1 antigen expressed specifically on bovine CD4-CD8-gamma delta T lymphocytes. *The Journal* of Immunology, 149, 3273-3277.
- WING, K., EKMARK, A., KARLSSON, H., RUDIN, A. & SURI-PAYER, E. 2002. Characterization of human CD25+ CD4+ T cells in thymus, cord and adult blood. *Immunology*, 106, 190-199.
- WRIGHT, H. L., MOOTS, R. J., BUCKNALL, R. C. & EDWARDS, S. W. 2010. Neutrophil function in inflammation and inflammatory diseases. *Rheumatology*, 49, 1618-1631.
- WU, B., ZHAN, Y., LI, F., CHENG, L., ZOU, S. & CHENG, Y. 2015a. Functional Plasticity in CD4+ CD25+ CD127-Treg Population in Primary Immune Thrombocytopenia. *Blood*, 126, 3470.

- WU, W., DIETZE, K. K., GIBBERT, K., LANG, K. S., TRILLING, M., YAN, H., WU, J., YANG, D., LU, M. & ROGGENDORF, M. 2015b. TLR ligand induced IL-6 counter-regulates the anti-viral CD8+ T cell response during an acute retrovirus infection. *Scientific Reports*, 5, 10501.
- WYLLIE, D., KISS-TOTH, E., VISINTIN, A., SMITH, S., BOUSSOUF, S., SEGAL, D., DUFF, G. & DOWER, S. 2000. Evidence for an accessory protein function for Toll-like receptor 1 in anti-bacterial responses. *The Journal of Immunology*, 165, 7125-7132.
- XU, J., LUPU, F. & ESMON, C. 2010. Inflammation, innate immunity and blood coagulation. *Hamostaseologie*, 30, 5.
- YABE, R., IWAKURA, Y. & SAIJO, S. 2014. The Role of C-Type Lectin Receptors C-type lectin receptors in the Host Defense Against Microbial Pathogens. *Glycoscience: Biology and Medicine.* Springer, 1319-1329.
- YANG, M., SUN, L., WANG, S., KO, K.-H., XU, H., ZHENG, B.-J., CAO, X. & LU, L. 2010. Cutting edge: novel function of B cell-activating factor in the induction of IL-10-producing regulatory B cells. *The Journal of Immunology*, 184, 3321-3325.
- YANG, Z., FU, Y., GONG, P., ZHENG, J., LIU, L., YU, Y., LI, J., LI, H., YANG, J. & ZHANG, X. 2015. Bovine TLR2 and TLR4 mediate *Cryptosporidium parvum* recognition in bovine intestinal epithelial cells. *Microbial Pathogenesis*, 85, 29-34.
- YI, H., ZHEN, Y., JIANG, L., ZHENG, J. & ZHAO, Y. 2006. The phenotypic characterization of naturally occurring regulatory CD4+ CD25+ T cells. *Cellular & Molecular Immunology*, 3, 189-195.
- YOUNG, H. A. & HARDY, K. 1995. Role of interferon-gamma in immune cell regulation. *Journal of Leukocyte Biology*, 58, 373-381.
- YUAN, D. & WITTE, P. L. 1988. Transcriptional regulation of mu and delta gene expression in bone marrow pre-B and B lymphocytes. *The Journal of Immunology*, 140, 2808-2814.
- ZANIN-ZHOROV, A., TAL-LAPIDOT, G., CAHALON, L., COHEN-SFADY, M., PEVSNER-FISCHER, M., LIDER, O. & COHEN, I. R. 2007. Cutting edge: T cells respond to lipopolysaccharide innately via TLR4 signaling. *The Journal of Immunology*, 179, 41-44.
- ZASTEPA, E., FITZ-GERALD, L., HALLETT, M., ANTEL, J., BAR-OR, A., BARANZINI, S., LAPIERRE, Y. & HAEGERT, D. G. 2014. Naive CD4 T-cell activation identifies MS patients having rapid transition to progressive MS. *Neurology*, 82, 681-690.

- ZHANG, W., BRAHMAKSHATRIYA, V. & SWAIN, S. L. 2014. CD4 T cell defects in the aged: causes, consequences and strategies to circumvent. *Experimental Gerontology*, 54, 67-70.
- ZHANG, Z., MILLER, W. & LIPMAN, D. J. 1997. Gapped BLAST and PSI-BLAST: a new generation of protein database search programs. *Nucleic Acids Research*, 25, 17.
- ZHAO, Z., XUE, Y., HU, Z., ZHOU, F., MA, B., LONG, T., XUE, Q. & LIU, H. 2017. Toll-like receptor 2 gene polymorphisms in Chinese Holstein cattle and their associations with bovine tuberculosis. *Veterinary Immunology and Immunopathology*, 186, 51-54.
- ZHENG, J., LIU, Y., LAU, Y.-L. & TU, W. 2013. γδ-T cells: an unpolished sword in human anti-infection immunity. *Cellular and Molecular Immunology*, 10, 50.
- ZHENG, S. G., WANG, J., WANG, P., GRAY, J. D. & HORWITZ, D. A. 2007. IL-2 is essential for TGF-β to convert naive CD4+ CD25– cells to CD25+ Foxp3+ regulatory T cells and for expansion of these cells. *The Journal of Immunology*, 178, 2018-2027.
- ZHU, J., VAN DRUNEN LITTEL-VAN DEN HURK, S., BROWNLIE, R., BABIUK, L. A., POTTER, A. & MUTWIRI, G. K. 2009. Multiple molecular regions confer intracellular localization of bovine Toll-like receptor 8. *Molecular Immunology*, 46, 884-92.
- ZIEGLER, S. F. 2006. FOXP3: of mice and men. *Annual Review of Immunology*, 24, 209-226.
- ZIMIN, A. V., DELCHER, A. L., FLOREA, L., KELLEY, D. R., SCHATZ, M. C., PUIU, D., HANRAHAN, F., PERTEA, G., VAN TASSELL, C. P., SONSTEGARD, T. S., MARCAIS, G., ROBERTS, M., SUBRAMANIAN, P., YORKE, J. A. & SALZBERG, S. L. 2009. A whole-genome assembly of the domestic cow, Bos taurus. Genome Biology, 10, R42.